UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

In re

Chapter 15

ARCTIC GLACIER INTERNATIONAL INC., :

Case No. 12-10605 (KG)

et al.,1

(Jointly Administered)

Debtors in a Foreign Proceeding.

Hearing Date: September 16, 2014 at 2:00 p.m. (ET) Obj. Deadline: September 9, 2014 at 4:00 p.m. (ET)

NOTICE OF MOTION FOR AN ORDER RECOGNIZING AND ENFORCING ORDER OF CANADIAN COURT SANCTIONING AND APPROVING CCAA PLAN

PLEASE TAKE NOTICE that Alvarez & Marsal Canada Inc., in its capacity as the court-appointed monitor and authorized foreign representative (the "Monitor") for the above-captioned debtors (collectively, the "Debtors") in a proceeding under Canada's Companies' Creditors Arrangement Act, R.S.C. 1985, c. C-36, as amended, pending before the Court of Queen's Bench Winnipeg Centre, has filed the attached Motion for an Order Recognizing and Enforcing Order of Canadian Court Sanctioning and Approving CCAA Plan (the "Sanction Recognition Motion").

PLEASE TAKE FURTHER NOTICE that a hearing (the "Hearing") to consider the Sanction Recognition Motion will be held on September 16, 2014 at 2:00 p.m. (ET) before the Honorable Kevin Gross at the United States Bankruptcy Court for the District of Delaware, 824 North Market Street, 6th Floor, Courtroom No. 3, Wilmington, Delaware 19801.

PLEASE TAKE FURTHER NOTICE that any objections to the Sanction Recognition Motion must be filed on or before September 9, 2014 at 4:00 p.m. (ET) (the "Objection Deadline") with the United States Bankruptcy Court for the District of Delaware, 824

The last four digits of the United States Tax Identification Number or Canadian Business Number, as applicable, follow in parentheses: (i) Arctic Glacier California Inc. (7645); (ii) Arctic Glacier Grayling Inc. (0976); (iii) Arctic Glacier Inc. (4125); (iv) Arctic Glacier Income Fund (4736); (v) Arctic Glacier International Inc. (9353); (vi) Arctic Glacier Lansing Inc. (1769); (vii) Arctic Glacier Michigan Inc. (0975); (viii) Arctic Glacier Minnesota Inc. (2310); (ix) Arctic Glacier Nebraska Inc. (7790); (x) Arctic Glacier New York Inc. (2468); (xi) Arctic Glacier Newburgh Inc. (7431); (xii) Arctic Glacier Oregon, Inc. (4484); (xiii) Arctic Glacier Party Time Inc. (0977); (xiv) Arctic Glacier Pennsylvania Inc. (9475); (xv) Arctic Glacier Rochester Inc. (6989); (xvi) Arctic Glacier Services Inc. (6657); (xvii) Arctic Glacier Texas Inc. (3251); (xviii) Arctic Glacier Vernon Inc. (3211); (xix) Arctic Glacier Wisconsin Inc. (5835); (xx) Diamond Ice Cube Company Inc. (7146); (xxi) Diamond Newport Corporation (4811); (xxii) Glacier Ice Company, Inc. (4320); (xxiii) Ice Perfection Systems Inc. (7093); (xxiv) ICEsurance Inc. (0849); (xxv) Jack Frost Ice Service, Inc. (7210); (xxvi) Knowlton Enterprises Inc. (8701); (xxvii) Mountain Water Ice Company (2777); (xxviii) R&K Trucking, Inc. (6931); (xxix) Winkler Lucas Ice and Fuel Company (0049); (xxx) Wonderland Ice, Inc. (8662). The Debtors' executive headquarters was located at 625 Henry Avenue, Winnipeg, Manitoba, R3A 0V1, Canada.

North Market Street, 3rd Floor, Wilmington, Delaware 19801. At the same time, you must serve a copy of the objection so as to be actually received by the following parties on or before the Objection Deadline: (i) Willkie Farr & Gallagher LLP, 787 Seventh Avenue, New York, New York 10019 (Attn: Marc Abrams, Mary K. Warren, and Alex W. Cannon); (ii) Osler, Hoskin & Harcourt LLP, 100 King Street West, Suite 6100, Toronto, Ontario, Canada M5X 1B8 (Attn: Marc Wasserman and Jeremy Dacks); (iii) Young Conway Stargatt & Taylor LLP, 1000 North King Street, Wilmington, Delaware 19801 (Attn: Robert S. Brady and Matthew P. Lunn); and (iv) McCarthy Tétrault LLP, 66 Wellington Street West, Suite 5300, Toronto, Ontario, Canada M5K 1E6 (Attn: Kevin P. McElcheran).

PLEASE TAKE FURTHER NOTICE that you need not appear at the Hearing if you do not object to the relief request in the Sanction Recognition Motion.

PLEASE TAKE FURTHER NOTICE that the Hearing may be continued or adjourned from time to time without further notice other than an announcement of the adjourned date or dates at the Hearing or at a later hearing or by notice filed on this Court's docket.

PLEASE TAKE FURTHER NOTICE THAT, IF YOU FAIL TO RESPOND IN ACCORDANCE WITH THIS NOTICE, THE COURT MAY GRANT THE RELIEF REQUESTED IN THE SANCTION RECOGNITION MOTION WITHOUT FURTHER NOTICE OR A HEARING.

PLEASE TAKE FURTHER NOTICE that additional copies of the Sanction Recognition Motion are available: (a) by accessing the Court's internet website at https://ecf.deb.uscourts.gov (a login and a password to the Court's Public Access to Court Electronic Records ("Pacer") are required to access this information and can be obtained through the PACER Service Center at http://www.pacer.psc.uscourts.gov); (b) from the Monitor's website at http://www.amcanadadocs.com/arcticglacier or http://www.kccllc.net/ArcticGlacier (without cost); or (c) upon written request to the Monitor's counsel (by email or facsimile) addressed to: Young Conaway Stargatt & Taylor LLP, Rodney Square, 1000 North King Street, Wilmington, Delaware 19801 (Attn.: Melissa Romano, e-mail, mromano@ycst.com or facsimile, 302-576-3450) (without cost).

Remainder of Page Intentionally Left Blank

Dated: August 26, 2014 Wilmington, Delaware

YOUNG CONAWAY STARGATT & TAYLOR, LLP

/s/ Ian J. Bambrick

Robert S. Brady (No. 2847) Matthew B. Lunn (No. 4119) Ryan M. Bartley (No. 4985) Ian J. Bambrick (No. 5455) Rodney Square 1000 North King Street Wilmington, Delaware 19801 Telephone: (302) 571-6600 Facsimile: (302) 571-1253

- and -

WILLKIE FARR & GALLAGHER LLP

Marc Abrams
Mary K. Warren
Alex W. Cannon
787 Seventh Avenue
New York, New York 10019-6099

Telephone: (212) 728-8000 Facsimile: (212) 728-8111

Co-Counsel to the Monitor and Foreign Representative

IN THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

In re : Chapter 15

ARCTIC GLACIER INTERNATIONAL INC., : Case No. 12-10605 (KG)

et al., (Jointly Administered)

Debtors in a Foreign Proceeding.

Hearing Date: Sept. 16, 2014 at 2:00 p.m.
Obj. Deadline: Sept. 9, 2014 at 4:00 p.m.

MOTION FOR AN ORDER RECOGNIZING AND ENFORCING ORDER OF CANADIAN COURT SANCTIONING AND APPROVING CCAA PLAN

Alvarez & Marsal Canada Inc., in its capacity as the court-appointed monitor and authorized foreign representative (the "Monitor") of the above-captioned debtors (collectively, the "Debtors") in the proceeding (the "Canadian Proceeding") commenced under Canada's Companies' Creditors Arrangement Act, R.S.C. 1985, c. C-36, as amended (the "CCAA"), and pending before the Court of Queen's Bench Winnipeg Centre (the "Canadian Court"), files this motion (the "Motion") for the entry of an order, pursuant to sections 105, 1507, 1525, and 1527 of title 11 of the United States Code (the "Bankruptcy Code"), substantially in the form attached as Exhibit A (the "Sanction Recognition Order"): (a) recognizing and giving full force and effect in the United States to the Canadian Court's Sanction Order (the "Sanction Order") with

The last four digits of the United States Tax Identification Number or Canadian Business Number, as applicable, follow in parentheses: (i) Arctic Glacier California Inc. (7645); (ii) Arctic Glacier Grayling Inc. (0976); (iii) Arctic Glacier Inc. (4125); (iv) Arctic Glacier Income Fund (4736); (v) Arctic Glacier International Inc. (9353); (vi) Arctic Glacier Lansing Inc. (1769); (vii) Arctic Glacier Michigan Inc. (0975); (viii) Arctic Glacier Minnesota Inc. (2310); (ix) Arctic Glacier Nebraska Inc. (7790); (x) Arctic Glacier New York Inc. (2468); (xi) Arctic Glacier Newburgh Inc. (7431); (xii) Arctic Glacier Oregon, Inc. (4484); (xiii) Arctic Glacier Party Time Inc. (0977); (xiv) Arctic Glacier Pennsylvania Inc. (9475); (xv) Arctic Glacier Rochester Inc. (6989); (xvi) Arctic Glacier Services Inc. (6657); (xvii) Arctic Glacier Texas Inc. (3251); (xviii) Arctic Glacier Vernon Inc. (3211); (xix) Arctic Glacier Wisconsin Inc. (5835); (xx) Diamond Ice Cube Company Inc. (7146); (xxi) Diamond Newport Corporation (4811); (xxii) Glacier Ice Company, Inc. (4320); (xxiii) Ice Perfection Systems Inc. (7093); (xxiv) ICEsurance Inc. (0849); (xxv) Jack Frost Ice Service, Inc. (7210); (xxvi) Knowlton Enterprises Inc. (8701); (xxvii) Mountain Water Ice Company (2777); (xxviii) R&K Trucking, Inc. (6931); (xxix) Winkler Lucas Ice and Fuel Company (0049); and (xxx) Wonderland Ice, Inc. (8662). The Debtors' executive headquarters was located at 625 Henry Avenue, Winnipeg, Manitoba, R3A 0V1, Canada.

respect to the *Consolidated Plan of Compromise and Arrangement* of the Debtors, dated as of May 21, 2014 (together with all appendices and exhibits thereto, the "<u>CCAA Plan</u>"),² subject to the consideration and issuance of such Sanction Order following a hearing before the Canadian Court:³ and (b) granting related relief. The Monitor respectfully represents as follows:

PRELIMINARY STATEMENT

States have been extremely successful. The Debtors, with the assistance and participation of the Monitor and the Chief Process Supervisor (as defined herein), conducted a robust sales process that led to the realization of cash well in excess of the value of all known Claims against the Debtors. The CCAA Plan is the mechanism by which that value will be distributed to holders of Proven Claims and the Debtors' Unitholders. Upon implementation of the CCAA Plan, all holders of Proven Claims will receive, in full and final satisfaction of their Claims, the full amount of their Proven Claims. Moreover, implementation of the Plan will result in the distribution of value to the ultimate Unitholders of the Debtors. By this Motion, the Monitor seeks entry of the Sanction Recognition Order to assist the Canadian Court, the Monitor, and the Chief Process Supervisor in implementing the terms of the CCAA Plan and the wind-downs and dissolutions of a significant number of the Debtors in the manner provided for in the CCAA Plan.

Capitalized terms used but not otherwise defined in the Motion have the meanings ascribed to such terms in the CCAA Plan.

A copy of the proposed Sanction Order is annexed hereto as <u>Exhibit A</u>. If approved by the Canadian Court, an as-entered copy of the Sanction Order (together with a blacklined version thereof, if necessary) will be filed with this Court and served upon all persons and entities receiving notice of this Motion.

JURISDICTION AND VENUE

2. This Court has jurisdiction to consider the Motion pursuant to 28 U.S.C. §§ 157 and 1334. This matter is a core proceeding pursuant to 28 U.S.C. §§ 157(b)(2)(P) and 1334 and the *Amended Standing Order of Reference* from the United States District Court for the District of Delaware, dated February 29, 2012. Venue is proper in this Court and in this District pursuant to 28 U.S.C. § 1410. The statutory predicates for the relief requested herein are sections 105(a), 1507, 1525, and 1527 of the Bankruptcy Code.

BACKGROUND

3. On February 22, 2012, the Debtors⁴ commenced the Canadian Proceeding, and the Canadian Court entered an initial order (including any extensions, amendments, or modifications thereto, the "<u>Initial Order</u>"), pursuant to the CCAA, providing various forms of relief thereunder, including, but not limited to, authorizing and directing the Debtors to commence, and the Monitor, the Financial Advisor (as defined in the Sale and Investor Solicitation Process (the "<u>SISP</u>"))⁵ and the Chief Process Supervisor (as defined in the SISP) to perform their obligations under, a process offering potential investors an opportunity to purchase or invest in the Debtors' business and operations in accordance with the SISP.

As described more fully in the *Declaration of Philip J. Reynolds in Support of Verified Petition of Alvarez & Marsal Canada Inc., as Foreign Representative of Arctic Glacier Inc. and Certain of its Affiliates, for:*(I) Recognition of Foreign Main Proceeding and (II) Certain Related Relief [Docket No. 2] (the "Reynolds Declaration"), Glacier Valley Ice Company, L.P ("Glacier L.P."), an affiliate of the Debtors, is not an applicant in the Canadian Proceeding because partnerships are ineligible to be applicants under the CCAA. However, pursuant to this Court's Provisional Relief Order and Recognition Order (as each term is defined below), the stay provided for in section 362 applies to Glacier L.P. Additionally the assets of Glacier L.P. were sold as provided in the CCAA Vesting Order and the U.S. Sale Order (both of which are defined herein). For convenience sake, all references to "Debtors" herein and in the Sanction Recognition Order shall include Glacier L.P., even though such entity is not a Debtor in the Chapter 15 Cases (as defined herein).

A copy of the SISP is annexed as <u>Schedule C</u> to the Initial Order, which is annexed as <u>Exhibit A</u> to the Reynolds Declaration.

- 4. On February 22, 2012 (the "<u>Petition Date</u>"), the Monitor commenced these proceedings (these "<u>Chapter 15 Cases</u>") by filing verified petitions on behalf of the Debtors, pursuant to sections 1504 and 1515, seeking recognition by this Court of the Canadian Proceeding as a foreign main proceeding under chapter 15 of the Bankruptcy Code.
- 5. On February 23, 2012, this Court entered the *Order Granting Provisional Relief* [Docket No. 28] (the "<u>Provisional Relief Order</u>"), providing for, among other things, a stay of all proceedings against or concerning property of the Debtors located within the territorial jurisdiction of the United States.
- 6. On March 16, 2012, this Court entered the *Order Granting Recognition of Foreign Main Proceeding and Certain Related Relief* [Docket No. 70] (the "Recognition Order"). Pursuant to the Recognition Order, this Court (a) granted recognition of the Canadian Proceeding as a foreign main proceeding under section 1517, (b) authorized the Debtors to obtain postpetition secured financing, and (c) enforced the Initial Order on a permanent basis in the United States.
- 7. On June 21, 2012, the Canadian Court entered the *Sale Approval and Vesting Order* (as amended and restated, the "<u>CCAA Vesting Order</u>"), pursuant to which the Canadian Court authorized and approved the sale of the Debtors' right, title, and interest in and to the Purchased Assets (as defined in the CCAA Vesting Order) free and clear of all Interests (as defined in the CCAA Vesting Order), except as provided in the Purchase Agreement (the "Sale").
- 8. On July 17, 2012, this Court entered an order [Docket No. 126] (the "<u>U.S. Sale Order</u>"), which among other things: (a) recognized and enforced the CCAA Vesting Order; (b) authorized and approved the Sale pursuant to section 363(f); (c) authorized

and approved, to the extent provided for in the CCAA Vesting Order, the assignment of the Assigned Contracts (as defined in the U.S. Sale Order); and (d) granted certain related relief.

- 9. As contemplated by the CCAA Vesting Order and described in the *Notice of Filing of Monitor's Certificate* [Docket No. 139] (the "Certificate Filing Notice"), on July 27, 2012, the Monitor delivered the *Monitor's Certificate*, which, among other things, notified the Canadian Court and other parties in interest that the Sale had closed. A copy of the *Monitor's Certificate* was filed with the Certificate Filing Notice on August 2, 2012 [Docket No. 139].
- 10. On July 12, 2012, the Canadian Court entered the *Transition Order*, a copy of which is annexed hereto as Exhibit C (the "Transition Order"), authorizing, but not requiring, the Monitor to prepare various tax returns on behalf of the Debtors and (a) ordering that the Monitor shall incur no liability with respect to such returns, (b) ordering that "Monitor shall not be deemed to be in possession and/or control" of any property of the Debtors, and (c) declaring that "nothing in [the Transition Order] shall constitute or be deemed to constitute the Monitor as a trustee, receiver, assignee, liquidator, administrator, receiver-manager, agent of the creditors or legal representative of any of the [Debtors] within the meaning of any relevant legislation." (Transition Order ¶ 4, 7-8.)
- Order (the "Claims Procedure Order") establishing procedures for the submission and determination of Claims against the Debtors and their directors, officers, and trustees (the "Claims Process"). In accordance with paragraphs 14 and 15 of the Claims Procedure Order, the Monitor mailed the Notice to Claimants (as defined in the Claims Procedure Order) to all known creditors of the Debtors and caused the Notice to Claimants to be published in (a) The Globe and

Mail newspaper (National Edition), (b) the Wall Street Journal (National Edition), and (c) the Winnipeg Free Press.

- 12. On September 14, 2012, this Court entered the *Order Recognizing and Enforcing Claims Procedure Order of the Canadian Court* [Docket No. 166] (the "Claims Procedure Recognition Order") recognizing and giving full force and effect in the United States to the Claims Procedure Order and the Claims Process contemplated thereby.
- Officer Order"), which, among other things, supplemented the Claims Procedure Order by empowering the Claims Officers (as defined in the Claims Officer Order and including the Special Claims Officer (as defined in the Claims Procedure Recognition Order)) with the authority to adjudicate and determine questions of fact and law concerning the validity and value of disputed Claims that could not be resolved consensually.
- 14. On May 7, 2013, this Court entered the Order: (A) Recognizing and Enforcing Certain Provisions of March 7, 2013 Order of the Canadian Court; and (B) Granting Certain Related Relief [Docket No. 227] recognizing and giving full force and effect in the United States to the Claims Officer Order.
- 15. On May 21, 2014, the Canadian Court entered the *Meeting Order* (the "<u>Meeting Order</u>"), which, among other things, authorized the process for (a) the determination of a resolution to approve the CCAA Plan by the Affected Creditors, and (b) the solicitation of votes to accept or reject the resolution to, among other things, approve the CCAA Plan.
- 16. On June 6, 2014, this Court entered the *Order Recognizing the Meeting Order of the Canadian Court* [Docket No. 323] recognizing and giving full force and effect in the United States to the Meeting Order.

- 17. As will be detailed more fully in the forthcoming Seventeenth Report of the Monitor (the "Seventeenth Report"), a copy of which will be filed as soon as practicable, on August 11, 2014, in accordance with the Meeting Order (a) the Creditors' Meeting was deemed to have been duly called and held and, thereat, every Affected Creditor was deemed to have voted in favor of a resolution to approve the CCAA Plan, and (b) a special meeting of the Unitholders was conducted in Toronto, Ontario, whereat the Unitholders considered and voted on a resolution to, among other things, approve the CCAA Plan. The Plan was approved by the Required Unitholder Majority, in conformity with the Meeting Order.
- 18. A hearing before the Canadian Court to consider entry of the Sanction Order is scheduled for September 5, 2014.
- 19. Additional information about the Debtors' former businesses and operations, the Canadian Proceeding (including an update regarding the Claims Process), the Sale, and these Chapter 15 Cases are set forth in (a) the Reynolds Declaration, (b) the *Pre-Filing Report of the Proposed Monitor*, filed on the Petition Date as Exhibit C to the Reynolds Declaration, (c) the reports of the Monitor previously filed with the Canadian Court and this Court, and (d) the forthcoming Seventeenth Report.

RELIEF REQUESTED

20. The Monitor seeks entry of the Sanction Recognition Order, pursuant to sections 105(a), 1507, 1525, and 1527 and under well-established principles of international comity, (a) recognizing and enforcing in full the Sanction Order of the Canadian Court and the CCAA Plan in the United States, and (b) granting related relief.

THE CCAA PLAN

- 21. The CCAA Plan, a copy of which is attached hereto as Exhibit B, is the culmination of over two years of diligent efforts by the Monitor, the Debtors, and the Chief Process Supervisor, and each of their respective professional advisors, to realize and distribute the value obtained from the Sale. These efforts involved administering a robust sales and marketing process resulting in the consummation of the Sale that generated cash proceeds sufficient to satisfy all of the Debtors' prepetition secured liabilities, leaving the Monitor holding significant funds for the benefit of the Debtors' various stakeholders. Additionally, the Monitor and its advisors, with the assistance of the Debtors and their advisors, conducted the Claims Process to identify, reconcile, and resolve all Claims against the Debtors, which included this Court's approval of a settlement concerning a proof of claim based on significant prepetition antitrust litigation.
- 22. If sanctioned by the Canadian Court, the Proposed Consolidated CCAA Plan⁶ and, as applicable, entry of the Sanction Recognition Order is expected to result in:
 - a. full recovery by Affected Creditors with Proven Claims, including applicable interest, if any;
 - b. sufficient reserves to satisfy Affected Creditors with Unresolved Claims once those Claims are resolved, including applicable interest, if any;
 - the wind-downs and dissolutions of certain of the Debtors pursuant to and in accordance with the timing and manner set forth in the Proposed Consolidated CCAA Plan;
 - d. cost-efficient distribution of any surplus of Available Funds to Unitholders on a pro rata basis; and
 - e. termination of the Declaration of Trust and cancellation of the Trust Units on the Final Distribution Date.

This summary of the CCAA Plan is being provided for reference only. In the event there are any discrepancies between this summary and the CCAA Plan, the terms of the CCAA Plan shall control.

23. As set forth more specifically in Section 9.1 of the CCAA Plan and the Sanction Recognition Order, certain parties, including the Debtors and their affiliates, the Monitor, the CPS, the Trustees, the Directors and Officers, and such parties' respective advisors are being released and discharged from, among other things, any and all claims and liabilities, including Claims for any unpaid tax, regardless of whether such tax remains subject to audit, demands and causes of action in any way related to, or arising out of or in connection with the Claims, the Debtors' business and affairs, the CCAA Plan, or the Canadian Proceeding.

BASIS FOR RELIEF

- 24. The Monitor submits that the relief requested herein is authorized pursuant to section 1507(b) of the Bankruptcy Code because enforcement of the CCAA Plan in the United States is consistent with the principles of comity and will reasonably assure:
 - a. just treatment of all holders of Claims against or interests in the Debtors' property, because the laws of Canada and the United States share both the same common law traditions and fundamental insolvency principles, and the Monitor believes that the CCAA Plan justly and fairly distributes the property of the Debtors for the benefit of all of the Debtors' stakeholders, both in the United States abroad;
 - b. protection of holders of Affected Claims and Unitholders in the United States against prejudice and inconvenience in the processing of such Claims in the Canadian Proceedings because Canadian law does not prefer the Claims of Canadian citizens over others, and because the inconvenience faced by United States creditors in the Canadian Proceedings is no greater than the inconvenience faced by foreign creditors bringing Claims in the United States;
 - c. prevention of preferential or fraudulent dispositions of the Debtors' property, because in the absence of enforcement, United States creditors might attempt to proceed against the Debtors' assets without regard for the Canadian Proceeding, leading to unequal treatment of certain creditors in a manner contrary to the fundamental purposes of the Bankruptcy Code; and
 - d. distribution of proceeds of the Debtors' property in a manner that is fundamentally similar to the order prescribed by the Bankruptcy Code.

- 25. In addition, section 1525(a) of the Bankruptcy Code provides that, "consistent with section 1501, the court shall cooperate to the maximum extent possible with a foreign court or a foreign representative," and section 1527(3) of the Bankruptcy Code explicitly provides that one such form of cooperation may include "coordination of the administration and supervision of the debtor's assets and affairs." 11 U.S.C. §§ 1525(a); 1527(3).⁷
- Creditors and Unitholders that it prescribes provide a fair and effective framework by which the Debtors' property will be distributed to Affected Creditors and Unitholders. Entry of the Sanction Recognition Order will further the administration of the Canadian Proceeding and these Chapter 15 Cases by assisting the Canadian Court and the Monitor in its efforts to wind-down and dissolve the corporate existence of a substantial number of the Debtors without the need for the administrative costs associated with obtaining board approvals and/or shareholder consents. As such, enforcement of the Sanction Order in the United States will ensure the efficient administration of the Debtors' estates on a uniform basis while simultaneously promoting the concepts of international cooperation in accordance with chapter 15 of the Bankruptcy Code. Accordingly, the Monitor respectfully requests that this Court recognize and enforce the Sanction Order, subject to its consideration and issuance by the Canadian Court. Additionally, Section 10.3 of the CCAA Plan provides that entry of the Sanction Recognition Order is a condition precedent to the implementation of the CCAA Plan.
- 27. The Monitor submits further that the enforcement and recognition of the Sanction Order and the CCAA Plan in the United States is not manifestly contrary to the public

Section 1521 of the Bankruptcy Code also proscribes various other forms of relief that may be granted in a chapter 15 proceeding following recognition. While the Monitor is of the view that section 1507 of the Bankruptcy Code establishes the basis for granting recognition and enforcement of the Sanction Order, section 1521 also provides a basis for, and does not include a prohibition on, the relief requested herein.

policy of the United States and, therefore, section 1506 of the Bankruptcy Code does not present any bar to the enforcement of such order and the CCAA Plan. The manner in which the Debtors' property is to be distributed under the CCAA Plan is consistent with restructuring methods contemplated by the Bankruptcy Code and United States insolvency principles. As such, the requested relief is consistent with the public policy of the United States.

- 28. While the CCAA Plan contains releases in favors of third parties, similar releases granted in foreign insolvency proceedings have been enforced in the United States under section 1507 of the Bankruptcy Code. In re Sino-Forest Corp., 501 B.R. 655 (Bankr. S.D.N.Y. 2013) (enforcing a third-party release that may not have been permissible under United States law); In re Metcalfe & Mansfield Alternative Invs., 421 B.R. 685 (Bankr. S.D.N.Y. 2010) (same).⁸
- 29. Although some have declined to enforce non-consensual third-party releases in the context of chapter 15 proceedings, see Ad Hoc Group of Vitro Noteholders v. Vitro, S.A.B. de C.V. (In re Vitro, S.A.B. de C.V.), 701 F.3d 1031 (5th Cir. 2012), they have done so (a) under factual circumstances vastly different than those presented in these cases, and (b) pursuant to a framework of statutory interpretation that is not binding in this judicial district. In Vitro, the plan in question, a Concurso plan proposed under the Mexican Business Reorganization Act, provided for the discharge of certain bond debt owed by the debtor, Vitro S.A.B. de C.V. ("Vitro"), and also discharged the obligations of certain non-debtor subsidiaries of Vitro that were guarantors of the debt. Vitro's Plan was objected to by certain of the bond-holders but was confirmed over their objection because the plan was accepted by 74.67% of Vitro's creditors however, over 50% of the votes cast on Vitro's plan were held by insiders of

The Third Circuit has previously held that non-consensual third party releases may be implemented under a chapter 11 plan in certain circumstances. <u>In re Continental Airlines</u>, 203 F.3d 203 (3d Cir. 2000).

Vitro, which are treated the same as votes of non-insider creditors under the Mexican Business Reorganization Act. The bankruptcy court denied recognition and enforcement of the releases in favor of the non-debtor subsidiaries. On appeal, the Fifth Circuit found that such releases were not categorically prohibited, but sustained the bankruptcy court's ruling on the facts presented in Vitro, which included:

- a. the absence of evidence of truly unusual circumstances necessitating the release;
- b. equity retaining substantial value while creditors did not receive a distribution close to what they were originally owed;
- c. the acceptance of the plan was dominated by votes of insider creditors, whose claims existed as a result of pre-petition restructuring of Vitro's financial obligations;
- d. the lack of consent by the affected non-insider creditors to the plan, and
- e. the lack of alternatives for non-consenting creditors to recover what they were owed in full.

The CCAA Plan and circumstances of these cases are vastly different from what was presented in <u>Vitro</u>. Most importantly, all holders of Proven Claims are being paid in full, with interest, under the CCAA Plan. As a result, the most significant concerns that resulted in denying enforcement of the third-party releases in <u>Vitro</u> — insiders and equity holders enriching themselves at the expense, and to the detriment, of non-insider creditors — are simply not present here.

30. The facts and circumstances surrounding the sanction of the CCAA Plan more closely resembles the facts and circumstances of the Metcalfe and Sino-Forest cases in which the plans at issue were overwhelmingly supported by those debtors' creditors. As observed by the Metcalfe court, "the correct inquiry in a chapter 15 case was not whether the Canadian orders could be enforced under U.S. law in a plenary chapter 11 case, but whether

recognition of the Canadian courts' decision was proper in the exercise of comity in a case under chapter 15." In re Metcalfe, 421 BR at 687. Here, because the CCAA Plan proposes to pay all holders of Proven Claims in full, with interest, and because the CCAA Plan is anticipated to be sanctioned by the Canadian Court and enjoys the overwhelming support of all of the Debtors' constituencies, the Monitor submits that the releases contained in the CCAA Plan are appropriate and should be approved by this Court as consistent with the policies animating chapter 15 of the Bankruptcy Code.

31. Based on the foregoing, the Monitor respectfully requests that this Court recognize and give effect in the United States to the Sanction Order and the CCAA Plan pursuant to sections 105(a), 1507, 1525, and 1527 of the Bankruptcy Code, and under well-established principles of international comity and cooperation.

NOTICE

32. Notice of the Motion will be provided to (a) all persons to whom notice is required pursuant to this Court's *Order Scheduling Hearing and Specifying Form and Manner of Service of Notice* [Docket No. 30], and (b) all other known parties in interest in these Chapter 15 Cases. In light of the nature of the relief requested herein, the Monitor submits that no other or further notice of this Motion is necessary or required.

Remainder of Page Intentionally Left Blank

CONCLUSION

WHEREFORE, the Monitor respectfully requests that this Court enter the Sanction Recognition Order, in substantially the form attached hereto as Exhibit A, granting the relief requested herein and such other and further relief as this Court deems just and proper.

Dated: Wilmington, Delaware August 26, 2014 YOUNG CONAWAY STARGATT & TAYLOR, LLP

/s/ Ian J. Bambrick

Robert S. Brady (No. 2847) Matthew B. Lunn (No. 4119) Rodney Square 1000 North King Street Wilmington, Delaware 19801 Telephone: (302) 571-6600

Facsimile: (302) 571-1253

- and -

WILLKIE FARR & GALLAGHER LLP Marc Abrams Alex W. Cannon 787 Seventh Avenue New York, New York 10019-6099 Telephone: (212) 728-8000

Facsimile: (212) 728-8111

Co-Counsel to the Monitor and Foreign Representative

EXHIBIT A

Sanction Recognition Order

IN THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

	:	Docket Ref. No.
Debtors in a Foreign Proceeding.	: :	(Jointly Administered)
ARCTIC GLACIER INTERNATIONAL INC., et al.,1	: :	Case No. 12-10605 (KG)
In re	:	Chapter 15

ORDER RECOGNIZING AND ENFORCING ORDER OF CANADIAN COURT SANCTIONING AND APPROVING CCAA PLAN

Upon consideration of the motion (the "Motion")² of Alvarez & Marsal Canada Inc., in its capacity as the court-appointed monitor and authorized foreign representative (the "Monitor") of the above-captioned debtors (collectively, the "Debtors") in the proceeding (the "Canadian Proceeding") commenced under Canada's *Companies' Creditors Arrangement Act*, R.S.C. 1985, c. C-36, as amended (the "CCAA") and pending before the Court of Queen's Bench Winnipeg Centre (the "Canadian Court"), for the entry of an order, pursuant to sections 105(a), 1507, 1525, and 1527 of title 11 of the United States Code (the "Bankruptcy Code"):

The last four digits of the United States Tax Identification Number or Canadian Business Number, as applicable, follow in parentheses: (i) Arctic Glacier California Inc. (7645); (ii) Arctic Glacier Grayling Inc. (0976); (iii) Arctic Glacier Inc. (4125); (iv) Arctic Glacier Income Fund (4736); (v) Arctic Glacier International Inc. (9353); (vi) Arctic Glacier Lansing Inc. (1769); (vii) Arctic Glacier Michigan Inc. (0975); (viii) Arctic Glacier Minnesota Inc. (2310); (ix) Arctic Glacier Nebraska Inc. (7790); (x) Arctic Glacier New York Inc. (2468); (xi) Arctic Glacier Newburgh Inc. (7431); (xii) Arctic Glacier Oregon, Inc. (4484); (xiii) Arctic Glacier Party Time Inc. (0977); (xiv) Arctic Glacier Pennsylvania Inc. (9475); (xv) Arctic Glacier Rochester Inc. (6989); (xvi) Arctic Glacier Services Inc. (6657); (xvii) Arctic Glacier Texas Inc. (3251); (xviii) Arctic Glacier Vernon Inc. (3211); (xix) Arctic Glacier Wisconsin Inc. (5835); (xx) Diamond Ice Cube Company Inc. (7146); (xxi) Diamond Newport Corporation (4811); (xxii) Glacier Ice Company, Inc. (4320); (xxiii) Ice Perfection Systems Inc. (7093); (xxiv) ICEsurance Inc. (0849); (xxv) Jack Frost Ice Service, Inc. (7210); (xxvi) Knowlton Enterprises Inc. (8701); (xxvii) Mountain Water Ice Company (2777); (xxviii) R&K Trucking, Inc. (6931); (xxix) Winkler Lucas Ice and Fuel Company (0049); and (xxx) Wonderland Ice, Inc. (8662). The Debtors' executive headquarters was located at 625 Henry Avenue, Winnipeg, Manitoba, R3A 0V1, Canada.

² Capitalized terms used but not otherwise defined herein shall have the meaning ascribed to such terms in the Motion or the CCAA Plan, as applicable.

(a) recognizing and giving full force and effect in the United States to the order entered by the Canadian Court sanctioning, approving, and enforcing the CCAA Plan (the "Sanction Order"); and (b) granting related relief; and due and sufficient notice of the Motion and the Sanction Order having been given; and it appearing that no other or further notice need be provided; and the Court having held a hearing to consider the Monitor's request for the relief set forth in the Motion; and no objections to the Motion having been filed or all such objections having been resolved or overruled; and the Court having found and determined that the relief sought in the Motion is consistent with the purposes of chapter 15 of the Bankruptcy Code; and the Court having reviewed and considered the Seventeenth Report; and it appearing that the relief requested in the Motion is in the best interests of the Debtors and other parties in interest in the Chapter 15 Cases; and after due deliberation and sufficient cause appearing therefore,

THE COURT FINDS AND CONCLUDES AS FOLLOWS:

- A. The Court has jurisdiction over this matter pursuant to 28 U.S.C. §§ 157 and 1334 and section 1501 of the Bankruptcy Code.
 - B. This is a core proceeding pursuant to 28 U.S.C. § 157(b)(2).
 - C. Venue is proper in this District pursuant to 28 U.S.C. § 1401(1).
- D. On September [__], 2014, the Canadian Court approved and entered the Sanction Order.
- E. The relief granted herein is necessary and appropriate, in the interests of the public and international comity, consistent with the public policy of the United States, warranted pursuant to section 1507 of the Bankruptcy Code, and will not cause hardship to any party in interest that is not outweighed by the benefits of the relief granted herein.

- F. The relief granted herein will, in accordance with section 1507(b) of the Bankruptcy Code, reasonably assure: (i) the just treatment of all holders of claims against or interests in the Debtors' property; (ii) the protection of claim holders in the United States against prejudice and inconvenience in the processing of claims in the Canadian Proceeding; (iii) the prevention of preferential or fraudulent dispositions of property of the Debtors; (iv) the distribution of proceeds of the Debtors' property substantially in accordance with the order prescribed in the Bankruptcy Code.
 - G. The public interest will be served by the relief granted herein.

IT IS HEREBY ORDERED, ADJUDGED, AND DECREED THAT:

- 1. The Motion is granted.
- 2. The Sanction Order, a copy of which is annexed hereto as <u>Exhibit 1</u>, is fully recognized, given full force and effect in the United States, is warranted pursuant to section 1507 of the Bankruptcy Code, and will not cause hardship to any party in interest that is not outweighed by the benefits of the relief granted herein.
- 3. The Debtors, the Monitor and the CPS, as the case may be, are authorized and directed to take all steps and actions necessary or appropriate to implement the CCAA Plan in accordance with and subject to its terms and conditions, and enter into, adopt, execute, deliver, complete, implement and consummate all of the steps, compromises, settlements, transactions, assignments, arrangements, reorganizations, distributions, payments, deliveries, allocations, instruments, agreements and releases contemplated by, and subject to the terms and conditions of, the CCAA Plan, and all such steps and actions are approved.
- 4. The steps, assumptions, distributions, transfers, payments, contributions, liquidations, dissolutions, wind-downs, reduction of capital, settlements, and releases set forth on

<u>Schedule B</u> to the CCAA Plan shall be deemed to have occurred in the order and manner set forth on such <u>Schedule B</u> without the need for board approval or shareholder consent to the fullest extent permitted by relevant state law.

On the Plan Implementation Date and in accordance with the sequential steps and transactions set out in Section 8.3 of the CCAA Plan, the Debtors, the Monitor, Alvarez and Marsal Canada Inc. and its affiliates, the CPS, the Trustees, the Directors and the Officers, each and every present and former employee who filed or could have filed an indemnity claim or a DO&T Indemnity Claim against the Debtors, each and every affiliate, subsidiary, member (including members of any committee or governance council), auditor, financial advisor, legal counsel and agent thereof and any Person claiming to be liable derivatively through any or all of the foregoing Persons (the "Releasees") shall be released and discharged from any and all demands, claims, including claims for any unpaid tax, regardless of whether such tax remains subject to audit, actions, causes of action, counterclaims, suits, debts, sums of money, accounts, covenants, damages, judgments, orders, including for injunctive relief or specific performance and compliance orders, expenses, executions and other recoveries on account of any liability, obligation, demand or cause of action of whatever nature which any Person may be entitled to assert, whether statutory or otherwise arising, including any and all claims in respect of the payment and receipt of proceeds and statutory liabilities of any of the Trustees, Directors, Officers and employees of the Debtors and any alleged fiduciary or other duty (whether acting as a Trustee, Director, Officer, member or employee or acting in any other capacity in connection with the Debtors' business or an individual Debtor), whether known or unknown, matured or unmatured, direct, indirect or derivative, foreseen or unforeseen, existing or hereafter arising, based in whole or in part on any omission, transaction, duty, responsibility,

indebtedness, liability, obligation, dealing or other occurrence existing or taking place on or prior to the later of the Plan Implementation Date and the date on which actions are taken to implement the CCAA Plan that are in any way related to, arising out of or in connection with the Claims, the Debtors' business and affairs whenever or however conducted, the Plan, the Canadian Proceedings and the Chapter 15 Cases, any Claim that has been barred or extinguished pursuant to the Claims Procedure Order or the Claims Officer Order (excepting only Releasees in respect of Unresolved Claims, unless and until such Unresolved Claims become Proven Claims in accordance with the Claims Procedure Order and the Claims Officer Order), and all claims arising out of such actions or omissions shall be forever waived, discharged and released (other than the right to enforce the Debtors' obligations under the Plan or any related document), all to the full extent permitted by applicable law, provided that nothing in the Plan shall release or discharge a Releasee from any obligation created by or existing under the Plan or any related document.

6. All Persons shall be permanently and forever barred, estopped, stayed and enjoined, from and after the Effective Time, in respect of any and all Releasees, from:

(a) commencing, conducting or continuing in any manner, directly or indirectly, any action, suits, demands or other proceedings of any nature or kind whatsoever (including, without limitation, any proceeding in a judicial, arbitral, administrative or other forum) against the Releasees;

(b) enforcing, levying, attaching, collecting or otherwise recovering or enforcing by any manner or means, directly or indirectly, any judgment, award, decree or order against the Releasees or their respective property; (c) commencing, conducting or continuing in any manner, directly or indirectly, any action, suit or demand, including without limitation by way of contribution or indemnity or other relief, in common law or in equity, for breach of trust or breach of fiduciary

duty, under the provisions of any statute or regulation, or other proceedings of any nature or kind whatsoever (including, without limitation, any proceeding in a judicial, arbitral, administrative or other forum) against any Person who makes such a claim or might reasonably be expected to make such a claim, in any manner or forum, against one or more of the Releasees; (d) creating, perfecting, asserting or otherwise enforcing, directly or indirectly, any lien or encumbrance of any kind against the Releasees or their respective property; or (e) taking any actions to interfere with the implementation or consummation of the CCAA Plan; provided, however, that the foregoing shall not apply to the enforcement of any obligations under the CCAA Plan.

- 7. No action or other proceeding shall be commenced against the Monitor or the CPS in any way arising from or related to their respective capacities or conduct as Monitor or CPS, except with prior leave pursuant to an Order of the Canadian Court made on prior written notice to the Monitor and the CPS and provided any such Order granting leave includes a term granting the Monitor or the CPS, as applicable, security for its costs and the costs of its counsel in connection with any proposed action or proceeding, such security to be on terms the Canadian Court deems just and appropriate; provided, that, that the Canadian Court shall have exclusive jurisdiction over any such action or other proceeding against the Monitor or the CPS.
- 8. The DOJ Claim shall include interest at the United States federal post-judgment interest rate of 0.34%, compounding annually until the date of payment of such DOJ Claim, as provided for in the *Stipulation and Order Among the Monitor, Debtors, and the United States Attorney's Office for the Southern District of Ohio Regarding March 2010 Criminal Judgment of Arctic Glacier International Inc.*, dated July 17, 2012, as entered by this Court.
- 9. Neither the Debtors nor the Monitor shall incur any liability as a result of acting in accordance with the terms of the Plan and this Sanction Recognition Order.

10. The Monitor and the Debtors are authorized and empowered to, and may in their discretion and without any delay, take any such steps or perform such actions as may be necessary to effectuate the terms of this Order and the CCAA Plan, including, but not limited to, seeking any relief from this Court that may be necessary or appropriate to ensure that all conditions to the Plan Implementation Date will be satisfied.

11. Notwithstanding any provisions in the Federal Rules of Bankruptcy

Procedure to the contrary: (a) this Order shall be effective immediately and enforceable upon its
entry; and (b) neither the Monitor nor the Debtors are subject to any stay in the implementation,
enforcement, or realization of the relief granted in this Order.

12. Subject to paragraph 7 hereof, this Court shall retain jurisdiction with respect to all matters relating to the interpretation or implementation of this Order, and nothing contained herein shall be deemed to limit, modify, or impair this Court's jurisdiction over any motion of the Monitor requesting relief in aid of the Canadian Proceeding pursuant to applicable provisions of the Bankruptcy Code and such jurisdiction is hereby retained.

Dated: Wilmington, Delaware		
, 2014		
	The Honorable Kevin Gross	
	United States Bankruptcy Judge	

EXHIBIT 1

Sanction Order

THE QUEEN'S BENCH Winnipeg Centre

IN THE MATTER OF THE *COMPANIES' CREDITORS* ARRANGEMENT ACT, R.S.C. 1985, c. C-36, AS AMENDED

AND IN THE MATTER OF A PROPOSED PLAN OF COMPROMISE OR ARRANGEMENT WITH RESPECT TO ARCTIC GLACIER INCOME FUND, ARCTIC GLACIER INC., ARCTIC GLACIER INTERNATIONAL INC. and the ADDITIONAL APPLICANTS LISTED IN SCHEDULE "A" HERETO

(collectively, the "APPLICANTS")

APPLICATION UNDER THE *COMPANIES' CREDITORS ARRANGEMENT ACT*, R.S.C. 1985, c C-36, AS AMENDED

SANCTION ORDER

KEVIN P. MCELCHERAN PROFESSIONAL CORPORATION

120 Adelaide Street West Suite 420 Toronto ON M5H 1T1

Kevin McElcheran

Tel: 416.855.0444 Email: <u>Kevin@mcelcheranAdr.com</u>

MCCARTHY TÉTRAULT LLP

Suite 5300, Box 48 Toronto Dominion Bank Tower Toronto ON M5K 1E6

Heather Meredith

Tel: 416.601.8342 Email: hmeredith@mccarthy.ca

AIKINS, MACAULAY THORVALDSON LLP

30th Floor —360 Main Street Winnipeg, MB R3C 4G1

G. Bruce Taylor

Tel: 204.957.4669 Fax: 204.957.4218

J.J. Burnell

Tel: 204.957.4663 Fax: 204.957.4285

THE QUEEN'S BENCH

Winnipeg Centre

THE HONOURABLE) FRIDAY, THE 5TH
)
MADAM JUSTICE SPIVAK) DAY OF SEPTEMBER, 2014

IN THE MATTER OF THE *COMPANIES' CREDITORS*ARRANGEMENT ACT, R.S.C. 1985, c. C-36, AS AMENDED

AND IN THE MATTER OF A PROPOSED PLAN OF COMPROMISE OR ARRANGEMENT WITH RESPECT TO ARCTIC GLACIER INCOME FUND, ARCTIC GLACIER INC., ARCTIC GLACIER INTERNATIONAL INC. and the ADDITIONAL APPLICANTS LISTED IN SCHEDULE "A" HERETO

(collectively, the "APPLICANTS")

ORDER (Plan Sanction)

THIS MOTION made by the Applicants for an Order (the "Sanction Order"):

(i) approving and sanctioning the amended and restated consolidated plan of compromise or arrangement dated August 26, 2014 (and as it may be further amended, restated, modified or supplemented from time to time in accordance with its terms) (the "Plan"), attached as Schedule "B" to this Sanction Order; and (ii) extending the stay of proceedings (the "Stay Period") under the Order of the Honourable Madam Justice Spivak made February 22, 2012 (the "Initial Order") was heard this day at the Law

Courts Building at 408 York Avenue, in the City of Winnipeg, in the Province of Manitoba.

ON READING the Notice of Motion, the Sixteenth Report of the Monitor dated August 7, 2014 (the "Sixteenth Report") and the Seventeenth Report of the Monitor dated August 26, 2014 (the "Seventeenth Report"), and on hearing the submissions of counsel for the Applicants and Glacier Valley Ice Company, L.P. (together, the "Arctic Glacier Parties"), counsel for the Monitor and ●, no one appearing for any other party although duly served as appears from the Affidavit of Service, filed:

DEFINITIONS

1. **THIS COURT ORDERS** that any capitalized terms not otherwise defined in this Sanction Order shall have the meanings ascribed to them in the Plan.

SERVICE

2. **THIS COURT ORDERS** that the time for service of this Motion and the Sixteenth and Seventeenth Report be and is hereby abridged and validated, such that this Motion is properly returnable today and hereby dispenses with further service thereof.

THE CREDITORS' MEETING AND THE UNITHOLDERS' MEETING

3. **THIS COURT ORDERS AND DECLARES** that there has been good and sufficient service and delivery of the Meeting Order and the documents referred to in the Meeting Order, including the Notice to Affected Creditors and Notice to Unitholders.

- 4. **THIS COURT DECLARES** that the Creditors' Meeting was deemed to have been duly called and held in accordance with the Orders of this Court in the CCAA Proceedings, including the Meeting Order.
- 5. **THIS COURT ORDERS AND DECLARES** that the Unitholders' Meeting was duly called and held in accordance with the Orders of this Court in the CCAA Proceedings, including the Meeting Order.

MONITOR'S ACTIVITIES AND REPORTS

- 6. **THIS COURT ORDERS** that the Sixteenth Report and the Seventeenth Report, and the activities described therein, are hereby approved.
- 7. THIS COURT ORDERS that the activities and conduct of the Monitor and the CPS in relation to the Arctic Glacier Parties and the CCAA Proceedings, and of the Monitor in conducting and administering the Unitholders' Meeting on August 11, 2014 (as more particularly described in the Seventeenth Report) be and are hereby ratified and approved.

SANCTION OF THE PLAN

- 8. THIS COURT ORDERS AND DECLARES that:
 - (a) pursuant to the Meeting Order, the Plan has been approved unanimously by the Affected Creditors;

- (b) the Plan has been approved by the Required Unitholder Majority, in conformity with the Meeting Order;
- (c) the activities of the Arctic Glacier Parties have complied with the provisions of the CCAA and the Orders of this Court made in the CCAA Proceedings in all respects;
- (d) the CCAA Court is satisfied that the Arctic Glacier Parties have acted, and are acting, in good faith and with due diligence, and have complied with the provisions of the CCAA and the Orders of this Court made in the CCAA Proceedings in all respects;
- (e) the CCAA Court is satisfied that the Arctic Glacier Parties have not done or purported to do anything that is not authorized by the CCAA; and
- (f) the Plan, all terms and conditions thereof, and the matters and the transactions contemplated thereby, are fair and reasonable.
- 9. THIS COURT ORDERS AND DECLARES that the Plan (including, without limitation, the transactions, arrangements, reorganizations, assignments, cancellations, compromises, settlements, extinguishments, discharges, injunctions and releases set out therein) is hereby sanctioned and approved pursuant to the CCAA.

PLAN IMPLEMENTATION

10. **THIS COURT ORDERS** that at the Effective Time, the Plan and all associated steps, compromises, settlements, extinguishments, cancellations, transactions,

assignments, injunctions, arrangements, releases, reorganizations and discharges effected thereby shall be binding and effective upon, and shall enure to the benefit of, the Arctic Glacier Parties; all Affected Creditors; the Directors, Officers, Unitholders and Trustees; the Releasees; and all other Persons named or referred to in, or subject to, the Plan and their respective heirs, executors, administrators and other legal representatives, successors and assigns.

- 11. **THIS COURT ORDERS** that at the Effective Time, the Plan and all associated steps, compromises, settlements, extinguishments, cancellations, transactions, assignments, arrangements, injunctions, releases, reorganizations and discharges effected thereby shall be, and are hereby deemed to be:
 - (a) implemented, in accordance with the provisions of the Plan; and
 - (b) effected in the sequential order and at the times contemplated by Section 8.3 of the Plan, without any further act or formality, on the Plan Implementation Date beginning at the Effective Time.
- THIS COURT ORDERS that the Arctic Glacier Parties, the Monitor and the CPS, as the case may be, are hereby authorized and directed to take all steps and actions necessary or appropriate to implement the Plan in accordance with and subject to its terms and conditions, and enter into, adopt, execute, deliver, complete, implement and consummate all of the steps, compromises, settlements, transactions, assignments, arrangements, reorganizations, distributions, payments, deliveries, allocations, instruments, agreements and releases contemplated by, and subject to the terms and conditions of, the Plan, and all such steps and actions are hereby approved. Further, to the

extent not previously given, all necessary approvals to take such actions shall be and are hereby deemed to have been obtained from the Directors, Officers, or Trustees, as applicable, including the deemed passing by any class of shareholders or unitholders of any resolution or special resolution, and no shareholders' agreement or agreement between a shareholder and another Person limiting in any way the right to vote shares held by such shareholder or shareholders with respect to any of the steps contemplated in the Plan shall be effective or have an force or effect.

- THIS COURT ORDERS that on and after the Plan Implementation Date, the Monitor shall be at liberty to engage such Persons as the Monitor deems necessary or advisable respecting the exercise of its powers and performance of its obligations under the Plan, the Sanction Order or any other Order of this Court and to facilitate the completion of the CCAA Proceedings, and that the fees and costs incurred in respect of such engagement shall constitute an Administrative Reserve Cost within the meaning of the Plan.
- 14. **THIS COURT ORDERS** that none of the Arctic Glacier Parties, the Monitor and/or the CPS shall incur any liability as a result of acting in accordance with the terms of the Plan or this Sanction Order.
- 15. **THIS COURT ORDERS AND DECLARES** that the Applicants, the CPS and the Monitor are hereby authorized and empowered to exercise all consent and approval rights provided for in the Plan in the manner set forth in the Plan, whether before, after or on the Plan Implementation Date.

- 16. THIS COURT ORDERS that the Monitor, the Transfer Agent and any other Person required to make any distributions, payments, deliveries or allocations or take any steps or actions related thereto pursuant to the Plan are hereby authorized and directed to complete such distributions, payments, deliveries or allocations and to take any such related steps or actions, as the case may be, in accordance with the terms of the Plan, and such distributions, payments, deliveries and allocations, and the steps and actions related thereto, are hereby approved.
- 17. **THIS COURT ORDERS** that subject to payment of any amounts secured by the Charges, each of the Charges shall be terminated, discharged and released upon the filing by the Monitor with this Court of the certificate contemplated by Section 8.2 of the Plan.

EFFECT OF PLAN IMPLEMENTATION

18. THIS COURT ORDERS that, from and after the Plan Implementation Date, all Persons shall be deemed to have waived any and all defaults of the Arctic Glacier Parties then existing or previously committed by the Arctic Glacier Parties or caused by the Arctic Glacier Parties or any of the provisions of the Plan or this Sanction Order or non-compliance with any covenant, warranty, representation, term, provision, condition or obligation, express or implied, in any contract, agreement, mortgage, security agreement, indenture, trust indenture, loan agreement, commitment letter, agreement for sale, real property lease, personal property lease or other agreement, written or oral, and any amendments or supplements thereto, existing between such Person and the Arctic Glacier Parties. Any and all notices of default, acceleration of

payments and demands for payments under any instrument, or other notices, including without limitation, any notices of intention to proceed to enforce security, arising from any of such aforesaid defaults shall be deemed to have been rescinded and withdrawn.

- 19. **THIS COURT ORDERS** that, as of the Plan Implementation Date, each Affected Creditor and Unitholder shall be deemed to have consented and agreed to all of the provisions of the Plan in their entirety and, in particular, each Affected Creditor and Unitholder shall be deemed:
 - (a) to have granted, executed and delivered to the Monitor and the Arctic Glacier Parties all documents, consents, releases, assignments, waivers or agreements, statutory or otherwise, required to implement and carry out the Plan in its entirety; and
 - (b) to have agreed that if there is any conflict between the provisions of the Plan and the provisions, express or implied, of any agreement or other arrangement, written or oral, existing between such Affected Creditor or Unitholder and the Arctic Glacier Parties as of the Plan Implementation Date, the provisions of the Plan take precedence and priority, and the provisions of such agreement or other arrangement shall be deemed to be amended accordingly.
- 20. **THIS COURT ORDERS** that, subject to the Claims Procedure Order including the powers of the Monitor set out in paragraph 5 therein, any Affected Claim for which a Proof of Claim has not been filed by the Claims Bar Date or the DO&T Indemnity Claims Bar Date, as applicable, shall be forever barred and extinguished.

- THIS COURT ORDERS that, on the Plan Implementation Date, following completion of the steps in the sequence set forth in Section 8.3 of the Plan, all debentures, notes, bills of exchange, certificates, agreements, invoices and other instruments evidencing Affected Claims shall not entitle the holder thereof to any compensation or participation and shall be and are hereby deemed to be cancelled and shall be and are hereby deemed to be null and void, and the obligations of the Arctic Glacier Parties thereunder or in any way related thereto shall be satisfied and discharged.
- 22. THIS COURT ORDERS that, pursuant to and in accordance with the Plan, any and all Affected Claims shall be forever compromised, discharged, settled and released, and the ability of any Person to proceed against the Arctic Glacier Parties in respect of or relating to any Affected Claims shall be forever barred, discharged, enjoined and restrained, and all proceedings in respect of such Affected Claims are hereby permanently stayed, subject only to (i) the right of Affected Creditors with Unresolved Claims to continue pursuing such Unresolved Claims in accordance with the Claims Procedure Order, the Claims Officer Order and the Plan; and (ii) the right of Affected Creditors to receive payments and distributions pursuant to the Plan.

INTEREST

23. **THIS COURT ORDERS** that the rate of interest to be paid on each Proven Claim (other than the Deemed Proven Claims, the Canadian Direct Purchaser Proven Claim and the Indirect Purchaser Proven Claim) from and after the Claims Bar Date is 1.5%.

- THIS COURT ORDERS that the DOJ Claim shall include interest at the United States federal post-judgment interest rate of 0.34%, compounding annually until the date of payment of such DOJ Claim, as provided for in the *Stipulation and Order Among the Monitor, Debtors, and the United States Attorney's Office for the Southern District of Ohio Regarding March 2010 Criminal Judgment of Arctic Glacier International Inc.*, dated July 17, 2012, as entered by the U.S. Bankruptcy Court in the Chapter 15 Proceedings.
- 25. **THIS COURT ORDERS** that the Direct Purchaser Claim shall include interest at the rate of 0.3% calculated commencing on April 2, 2012.

STAY EXTENSION

- 26. **THIS COURT ORDERS** that the Stay Period is hereby extended until November 28, 2014.
- THIS COURT ORDERS that any and all Persons shall be and are hereby stayed from commencing, taking, applying for or issuing or continuing any and all steps or proceedings, including, without limitation, administrative hearings and orders, declarations or assessments, commenced taken or proceeded with or that may be commenced, taken or proceeded with against any Releasee in respect of all Claims and any matter which is released pursuant to the Plan.

RELEASES AND INJUNCTIONS

- 28. THIS COURT ORDERS AND DECLARES that the releases contemplated by the Plan are approved.
- 29. THIS COURT ORDERS that all Persons shall be permanently and forever barred, estopped, stayed and enjoined, from and after the Effective Time, in respect of any and all Releasees, from: (i) commencing, conducting or continuing in any manner, directly or indirectly, any action, suits, demands or other proceedings of any nature or kind whatsoever (including, without limitation, any proceeding in a judicial, arbitral, administrative or other forum) against the Releasees; (ii) enforcing, levying, attaching, collecting or otherwise recovering or enforcing by any manner or means, directly or indirectly, any judgment, award, decree or order against the Releasees or their respective property; (iii) commencing, conducting or continuing in any manner, directly or indirectly, any action, suit or demand, including without limitation by way of contribution or indemnity or other relief, in common law or in equity, for breach of trust or breach of fiduciary duty, under the provisions of any statute or regulation, or other proceedings of any nature or kind whatsoever (including, without limitation, any proceeding in a judicial, arbitral, administrative or other forum) against any Person who makes such a claim or might reasonably be expected to make such a claim, in any manner or forum, against one or more of the Releasees; (iv) creating, perfecting, asserting or otherwise enforcing, directly or indirectly, any lien or encumbrance of any kind against the Releasees or their respective property; or (v) taking any actions to interfere with the implementation or consummation of the Plan; provided, however, that the foregoing shall not apply to the enforcement of any obligations under the Plan.

ORDERS IN THE CCAA PROCEEDINGS

30. **THIS COURT ORDERS** that the Orders made in the CCAA Proceedings shall continue in full force and effect in accordance with their respective terms, except to the extent that such Orders are varied by or are inconsistent with this Sanction Order or any further Order of this Court.

THE MONITOR AND CHIEF PROCESS SUPERVISOR

31. THIS COURT ORDERS AND DECLARES that the Monitor and the CPS have satisfied all of their obligations up to and including the date of this Sanction Order, and that: (i) in carrying out the terms of this Sanction Order and the Plan and in performing their respective duties as Monitor and CPS, as applicable, in the CCAA Proceedings, the Monitor and the CPS shall have all the protections given to each of them by the CCAA, the Initial Order, the Meeting Order, the Claims Procedure Order and any other Order of this Court and as officers of the Court, including the stay of proceedings in their favour; (ii) the Monitor and the CPS shall incur no liability or obligation for any act or omission as a result of carrying out the provisions of this Sanction Order and the Plan and in performing their duties as Monitor and CPS, respectively, in the CCAA Proceedings, save and except for any gross negligence or wilful misconduct on their parts; and (iii) the Monitor and the CPS shall not be liable for any claims or damages resulting from any errors or omissions in the books and records of the Arctic Glacier Parties and any information provided by the Arctic Glacier Parties, including with respect to reliance thereon by any Person, save and except for any gross negligence or wilful misconduct on the Monitor's or CPS's part, as the case may be. Subject to the foregoing,

and in addition to the protections in favour of the Monitor under the CCAA, and the protections in favour of the Monitor and the CPS as set out in the Orders of this Court, any claims against the Monitor or the CPS in connection with the performance of their respective duties are hereby released, stayed, extinguished and forever barred and the Monitor and the CPS shall have no liability in respect thereof.

- 32. THIS COURT ORDERS that that the Monitor and the CPS shall not incur any liability under the Tax Statutes as a result of the completion of the steps or transactions contemplated by the Plan, including in respect of its making any payments or distributions ordered or permitted under the Plan or the Sanction Order and including any steps or transactions contemplated by Sections 8.3 or 8.4 of the Plan, and that the Monitor and the CPS are released, remised and discharged from any claims against them under or pursuant to the Tax Statutes or otherwise at law, arising in respect of the completion of the steps or transactions contemplated by the Plan, including in respect of making any payments or distributions ordered or permitted under the Plan or the Sanction Order and including any steps or transactions contemplated by Sections 8.3 or 8.4 of the Plan, and that any claims of such a nature are forever barred and extinguished
- 33. THIS COURT ORDERS that no action or other proceeding shall be commenced against the Monitor or the CPS in any way arising from or related to their respective capacities or conduct as Monitor or CPS, except with prior leave pursuant to an Order of this Court made on prior written notice to the Monitor and the CPS and provided any such Order granting leave includes a term granting the Monitor or the CPS, as applicable, security for its costs and the costs of its counsel in connection with any proposed action or proceeding, such security to be on terms this Court deems just and

appropriate. In addition, this Court orders that it has exclusive jurisdiction over any action or other proceeding commenced against the Monitor or the CPS in any way arising from or related to their respective capacities or conduct as Monitor or CPS.

34. THIS COURT ORDERS AND DECLARES that, in addition to the Monitor's prescribed rights under the CCAA, and the powers granted by this Court to the Monitor and the CPS, as the case may be, the powers granted to the Monitor and the CPS are expanded as may be required, and the Monitor and the CPS are empowered and authorized on and after the Plan Implementation Date, to take such additional actions and execute such documents, in the name of and on behalf of the Arctic Glacier Parties, as the Monitor and the CPS consider necessary or desirable in order to perform their respective functions and fulfill their respective obligations under the Plan, the Sanction Order and any Order of this Court in the CCAA Proceedings and to facilitate the implementation of the Plan and the completion of the CCAA Proceedings, including to: (i) take measures to attempt to satisfy or waive the conditions precedent under the Plan; (ii) administer and distribute the Available Funds; (iii) establish, hold, administer and distribute the Administrative Costs Reserve, the Insurance Deductible Reserve, the Unresolved Claims Reserve, the Affected Creditors' Distribution Cash Pool and the Unitholders' Distribution Cash Pool; (iv) resolve any Unresolved Claims; (v) effect payments in respect of Proven Claims to Affected Creditors and effect distributions to the Transfer Agent in respect of distributions to be made to Unitholders; (vi) take such steps, if and as may be necessary, to address Excluded Claims in accordance with the Plan, the Claims Procedure Order and the Claims Officer Order; and (vii) take such steps as are necessary to effect the post-Plan Implementation Date steps and transactions set out in Section 8.4 of the Plan; and, in

each case where the Monitor or the CPS, as the case may be, takes such actions or steps, they shall be exclusively authorized and empowered to do so, to the exclusion of all other Persons including the Arctic Glacier Parties, and without interference from any other Person.

- 35. THIS COURT ORDERS that on or following the Plan Implementation Date, the Monitor shall be and is hereby authorized and directed to make payments out of the Administrative Costs Reserve, on behalf of the Arctic Glacier Parties, in respect of the payment of Administrative Reserve Costs by way of cheque (sent by prepaid ordinary mail to the Monitor's last known address for such recipient Persons) or wire transfer (in accordance with wire transfer instructions, if provided by such recipient Persons to the Monitor at least three (3) Business Days prior to the payment date set by the Monitor).
- 36. THIS COURT ORDERS that on or following the Plan Implementation Date, the Monitor shall be and is hereby authorized and directed to administer and make payments out of Insurance Deductible Reserve and Unresolved Claims Reserve pursuant to and in accordance with the Plan.
- THIS COURT ORDERS that all payments and distributions by or at the direction of the Monitor, the Transfer Agent, and any other Persons required to make payments or distributions, in each case on behalf of the Arctic Glacier Parties or Arctic Glacier Income Fund ("AGIF"), as applicable, under the Plan are for the account of the Arctic Glacier Parties or AGIF, as applicable, and the fulfillment of their obligations under Plan.

- 38. THIS COURT ORDERS that on the Plan Implementation Date or on any Distribution Date, as the case may be, the Monitor shall be and is hereby authorized and directed to make distributions out of the Affected Creditors' Distribution Cash Pool, on behalf of the Arctic Glacier Parties, to each Affected Creditor in the amount of such Affected Creditor's Distribution Claim by way of cheque (sent by prepaid ordinary mail to the address for such Affected Creditor specified in the Proof of Claim filed by such Affected Creditor).
- THIS COURT ORDERS that on the Plan Implementation Date or on any Distribution Date, as the case may be, the Monitor shall be and is hereby authorized and directed to make distributions out of the Unitholders' Distribution Cash Pool, on behalf of the Fund, to the Transfer Agent pursuant to and in accordance with the Plan.
- 40. **THIS COURT ORDERS** that none of the Monitor, the CPS, the Trustees, the Arctic Glacier Parties, or any individuals related thereto shall incur any liability as a result of payments and distributions to the Unitholders, in each case on behalf of AGIF, once such distribution or payment has been made by the Monitor to, and confirmation of receipt has been received by the Monitor from, the Transfer Agent.
- 41. **THIS COURT ORDERS** that the Monitor and CPS are hereby authorized to, in the name of and on behalf of the Arctic Glacier Parties, prepare and file the Arctic Glacier Parties' tax returns based solely upon information provided by the Arctic Glacier Parties and on the basis that the Monitor and the CPS shall incur no liability or obligation to any Person with respect to such returns or related documentation.

- 42. **THIS COURT ORDERS** that the Monitor is hereby authorized and directed to, on and after the Plan Implementation Date, (i) complete the claims process established in the Claims Procedure Order and Claims Officer Order; and (ii) take such further steps and seek such amendments to the Claims Procedure Order, Claims Officer Order or other Orders of this Court as the Monitor considers necessary or appropriate in order to fully determine, resolve or deal with any Claims.
- 43. THIS COURT ORDERS that as of the Effective Time, the Monitor and the CPS shall be discharged and released from their respective duties, other than those obligations, duties and responsibilities (i) necessary or required to give effect to the terms of the Plan and this Sanction Order, (ii) in relation to the claims process and all matters relating thereto as set out in the Claims Procedure Order and the Claims Officer Order, and (iii) in connection with the completion by the Monitor and the CPS of all other matters for which they are respectively responsible in connection with the Plan or pursuant to the Orders of this Court made in the CCAA Proceedings.
- THIS COURT ORDERS that upon completion by the Monitor and the CPS of their duties in respect of the Arctic Glacier Parties pursuant to the CCAA and any Orders in the CCAA Proceedings, including, without limitation, the Monitor's and the CPS' duties in respect of the claims process and distributions made by the Monitor in accordance with the Plan, the Monitor may file with this Court a certificate of Plan termination stating that all of its duties and the duties of the CPS in respect of the Arctic Glacier Parties pursuant to the CCAA, the Plan and the Orders in the CCAA Proceedings have been completed and thereupon (i) Alvarez & Marsal Canada Inc. shall be deemed to be discharged from its duties as Monitor of the Arctic Glacier Parties and released from

all claims relating to its activities as Monitor; and (ii) 7088418 Canada Inc., operating as Grandview Advisors shall be deemed to be discharged from its duties as the CPS of the Arctic Glacier Parties and released from all claims relating to its activities as CPS.

45. THIS COURT ORDERS that to the extent that and at the time that the Monitor and the CPS are discharged pursuant to paragraph 39 or 40, as the case may be, any claims against the Monitor or the CPS in respect of their respective capacities or conduct in these CCAA Proceedings or the performance of their duties as Monitor or CPS, as applicable, are released, stayed, extinguished and forever barred and the Monitor and the CPS shall have no liability in respect thereof, save and except for any gross negligence or wilful misconduct on the Monitor's or the CPS' part.

GENERAL PROVISIONS

46. **THIS COURT ORDERS** that the Arctic Glacier Parties, the CPS and the Monitor may apply to this Court for advice and direction in respect of any matters arising from or under the Plan.

EFFECT, RECOGNITION AND ASSISTANCE

- 47. **THIS COURT ORDERS** that this Sanction Order shall have full force and effect in all provinces and territories in Canada and abroad and as against all Persons to whom it may apply.
- 48. THIS COURT HEREBY REQUESTS the aid and recognition of any court, tribunal, or regulatory or administrative body having jurisdiction in Canada, the

United States or elsewhere to give effect to this Sanction Order and to assist the Arctic Glacier Parties, the Monitor and their respective agents in carrying out the terms of this Sanction Order. All courts, tribunals, regulatory and administrative bodies are hereby respectfully requested to make such Orders and to provide such assistance to the Arctic Glacier Parties and to the Monitor, as an officer of this Court, as may be necessary or desirable to give effect to this Sanction Order, to grant representative status to the Monitor in any foreign proceeding, or to assist the Arctic Glacier Parties and the Monitor and their respective agents in carrying out the terms of this Sanction Order.

49. THIS COURT ORDERS that each of the Arctic Glacier Parties and the Monitor be at liberty and is hereby authorized and empowered to apply to any court, tribunal, or regulatory or administrative body, wherever located, for the recognition of this Sanction Order and for assistance in carrying out the terms of this Sanction Order, and that the Monitor is authorized and empowered to act as a representative in respect of the CCAA Proceedings for the purpose of having the CCAA Proceedings recognized in a jurisdiction outside Canada.

EXHIBIT B

CCAA Plan

THE QUEEN'S BENCH

Winnipeg Centre

IN THE MATTER OF THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c. C-36, AS AMENDED

AND IN THE MATTER OF
A PROPOSED PLAN OF COMPROMISE OR
ARRANGEMENT WITH RESPECT TO ARCTIC GLACIER INCOME FUND, ARCTIC
GLACIER INC., ARCTIC GLACIER INTERNATIONAL INC. and the ADDITIONAL
APPLICANTS LISTED ON SCHEDULE "A" HERETO

(collectively, the "APPLICANTS")

AMENDED AND RESTATED CONSOLIDATED PLAN OF COMPROMISE OR ARRANGEMENT

concerning, affecting and involving

ARCTIC GLACIER INCOME FUND, ARCTIC GLACIER INC., ARCTIC GLACIER INTERNATIONAL INC., GLACIER VALLEY ICE COMPANY, L.P. and the ADDITIONAL APPLICANTS LISTED ON SCHEDULE "A" HERETO

August 26, 2014

TABLE OF CONTENTS

		Page
ARTICLE 1	INTERPRETATION	2
1.1	Definitions	
1.2	Certain Rules of Interpretation	
1.3	Successors and Assigns.	
1.4	Governing Law	
1.5	Schedules	
ARTICLE 2	PURPOSE AND EFFECT OF THE CONSOLIDATED CCAA PLAN	15
2.1	Purpose	15
2.2	Persons Affected	15
2.3	Persons Not Affected	15
ARTICLE 3	CLASSIFICATION OF CREDITORS, VOTING AND RELATED MATTERS	1.6
3.1	Claims Procedure	
3.1	Classification of Creditors	
3.3	Claims of Affected Creditors	
3.4	Creditors' Meeting	
3.5	Voting	
3.6	Guarantees and Similar Covenants	
3.7	Set-Off	
ARTICLE 4	CLASSIFICATION OF UNITHOLDERS, VOTING AND RELATED	
	MATTERS	
4.1	Unitholder Procedure	
4.2	Classification of Unitholders	
4.3	Unitholders' Meeting	
4.4	Voting	
4.5	Approval by Unitholders	
4.6	Guarantees and Similar Covenants	18
ARTICLE 5	AVAILABLE FUNDS, RESERVES AND CASH POOLS	18
5.1	Available Funds	
5.2	Administrative Costs Reserve	
5.3	Insurance Deductible Reserve	
5.4	Unresolved Claims Reserve	
5.5	Composition of the Affected Creditors' Distribution Cash Pool	19
5.6	Composition of the Unitholders' Distribution Cash Pool	20
5.7	Remaining Funds	20
ARTICLE 6	PROVISIONS REGARDING DISTRIBUTIONS AND PAYMENTS	
6.1	Distributions from the Affected Creditors' Distribution Cash Pool	
6.2	Distributions from the Unitholders' Distribution Cash Pool	
6.3	Payment of Administrative Reserve Costs	21

TABLE OF CONTENTS

(continued)

		Page
6.4	Payment of Insurance Deductible Reserve Costs	21
6.5	Cancellation of Instruments Evidencing Affected Claims.	
6.6	Crown Priority Claims	
6.7	Currency	
6.8	Interest	
6.9	Treatment of Undeliverable Distributions	
6.10	Assignment of Claims for Voting and Distribution Purposes	
6.11	Assignment of Trust Units for Voting Purposes	
6.12	Allocation of Distributions	
6.13	Withholding and Reporting Requirements	
ARTICLE 7	PROCEDURE FOR DISTRIBUTIONS REGARDING UNRESOLVED CLAIMS	25
7.1	No Distribution Pending Allowance	
7.2	Unresolved Claims Reserve	
7.3	Distributions After Unresolved Claims Resolved	
ARTICLE 8	COMPANY REORGANIZATION	26
8.1	Corporate Authorizations	
8.2	Charges	
8.3	Plan Implementation Date Steps and Transactions	26
8.4	Post-Plan Implementation Date Transactions	27
ARTICLE 9	RELEASES	28
9.1	Consolidated CCAA Plan Releases	28
ARTICLE 10	COURT SANCTION, CONDITIONS PRECEDENT AND IMPLEMENTATION	20
10.1	Application for Sanction Order	
10.3	Conditions Precedent to Implementation of the Consolidated CCAA Plan	
ARTICLE 11	GENERAL	33
11.1	Binding Effect	
11.2	Waiver of Defaults	
11.3	Claims Bar Date	34
11.4	Deeming Provisions	
11.5	Non-Consummation	34
11.6	Modification of the Consolidated CCAA Plan	35
11.7	Paramountcy	
11.8	Severability of Plan Provisions	36
11.9	Reviewable Transactions	36
11.10	Responsibilities of the Monitor	
11.11	Different Capacities	
11.12	Notices	36

TABLE OF CONTENTS

(continued)

	r:	ıge
11 12	Earth and American	2.
11.13	Further Assurances	. 38

THE QUEEN'S BENCH

Winnipeg Centre

IN THE MATTER OF THE *COMPANIES' CREDITORS*ARRANGEMENT ACT, R.S.C. 1985, c. C-36, AS AMENDED

AND IN THE MATTER OF A PROPOSED PLAN OF COMPROMISE OR ARRANGEMENT WITH RESPECT TO ARCTIC GLACIER INCOME FUND, ARCTIC GLACIER INC., ARCTIC GLACIER INTERNATIONAL INC. and the ADDITIONAL APPLICANTS LISTED IN SCHEDULE "A" HERETO

(collectively, the "APPLICANTS")

CONSOLIDATED PLAN OF COMPROMISE OR ARRANGEMENT

WHEREAS the Applicants and Glacier Valley Ice Company, L.P. (collectively, the "Arctic Glacier Parties") are insolvent;

AND WHEREAS the Applicants obtained an Order made by the Honourable Madam Justice Spivak of the Court of the Queen's Bench of Manitoba (the "CCAA Court") under the *Companies' Creditors Arrangement Act*, R.S.C. 1985, c. C-36, as amended (the "CCAA") dated February 22, 2012 (the "**Initial Order**") that, among other things, appointed Alvarez & Marsal Canada Inc. as Monitor (the "**Monitor**") of the Applicants and permitted the Applicants to file with the CCAA Court one or more plans of compromise or arrangement;

AND WHEREAS the Initial Order was recognized by the U.S. Bankruptcy Court pursuant to Chapter 15 of the U.S. Bankruptcy Code;

AND WHEREAS pursuant to and in accordance with the Initial Order, the Applicants conducted a Sale and Investor Solicitation Process (the "SISP") for the purpose of offering the opportunity for potential investors to purchase or invest in the business and operations of the Applicants;

AND WHEREAS on June 7, 2012, the Applicants entered into an agreement in accordance with the SISP (the "Asset Purchase Agreement") with Arctic Glacier, LLC (formerly H.I.G. Zamboni, LLC, the "Purchaser") providing for the purchase and sale of substantially all of the assets, undertaking and property of the Applicants (other than the assets of Arctic Glacier Income Fund (the "Fund")) used in the conduct of the Applicants' business (the "Assets");

AND WHEREAS the Asset Purchase Agreement was approved by the CCAA Court by an Order dated June 21, 2012, which was amended by an Order dated July 12, 2012, (the "Canadian Vesting and Approval Order");

AND WHEREAS the Canadian Vesting and Approval Order was recognized by an Order of the U.S. Bankruptcy Court in the Chapter 15 Proceedings on July 17, 2012;

AND WHEREAS the transactions contemplated by the Asset Purchase Agreement were completed on July 27, 2012 and, on closing, the Purchaser assumed the Assumed Liabilities (as defined in the Asset Purchase Agreement) and the Purchaser paid the cash portion of the Purchase Price (as defined in the Asset Purchase Agreement) by payment of certain obligations of the Applicants and by payment of the balance of approximately \$130.2 million which is being held by the Monitor in trust pending directions from the CCAA Court;

AND WHEREAS the Applicants no longer carry on any active business and the Available Funds (as defined herein) represent the entire estate available for the benefit of the creditors of the Applicants and the Unitholders;

AND WHEREAS the Monitor obtained an order made by the Honourable Madam Justice Spivak of the CCAA Court on September 5, 2012, as amended, extended, restated or varied from time to time, which, among other things, provided for a claims process and set the Claims Bar Date (the "Claims Procedure Order");

AND WHEREAS pursuant to the Claims Procedure Order, the CCAA Court established a procedure which, among other things, required all Persons having an Affected Claim to file a proof of such Affected Claim with the Monitor on or before the Claims Bar Date or the DO&T Indemnity Claims Bar Date, as applicable;

AND WHEREAS the Claims Procedure Order was recognized by the U.S. Bankruptcy Court on September 14, 2012;

AND WHEREAS the CCAA Court provided for the appointment of claims officers and established the claims officers' authority for adjudicating disputed Affected Claims by order of the Honourable Madam Justice Spivak made on March 7, 2013 (the "Claims Officer Order");

AND WHEREAS the Fund is a publicly traded limited purpose income trust established by the Declaration of Trust;

AND WHEREAS the Consolidated CCAA Plan will facilitate distributions to Affected Creditors and, to the extent of a sufficient surplus of Available Funds, the Unitholders;

NOW THEREFORE the Applicants hereby propose this Consolidated CCAA Plan to the Affected Creditors and the Unitholders under and pursuant to the CCAA:

ARTICLE 1 INTERPRETATION

1.1 Definitions

For the purposes of the Consolidated CCAA Plan, the following terms shall have the following meanings ascribed thereto:

"Administration Charge" has the meaning given to that term in paragraph 50 of the Initial Order.

- "Administrative Costs Reserve" has the meaning given to that term in Section 5.2 of the Consolidated CCAA Plan.
- "Administrative Reserve Costs" means administrative claims and costs outstanding on the Plan Implementation Date (or arising thereafter) falling within one or more categories to be specified by the Monitor, including, without limitation: (a) amounts in respect of the fees and costs to be incurred by (i) the Monitor, its counsel and its advisors; (ii) the Arctic Glacier Parties, their counsel and their advisors; (iii) the Trustees and their counsel; and (iv) the CPS, in each case on a solicitor and own client full indemnity basis (as applicable) with respect to the performance of such parties' duties and obligations whether arising before or after the Plan Implementation Date; (b) amounts secured by the Charges that remain owing on the Plan Implementation Date, if any; (c) amounts in respect of existing or future taxes, expenses and other disbursements that are or may become payable; (d) amounts in respect of outstanding Crown Claims, if any; (e) amounts in respect of potential cost awards regarding litigation associated with Claims; and (f) amounts in respect of general contingency costs.
- "Affected Claim" means any Claim or DO&T Claim that is not an Excluded Claim.
- "Affected Creditor" means any Person having an Affected Claim (including a Class Claim, DOJ Claim, DO&T Claim and/or a DO&T Indemnity Claim), but only with respect to and to the extent of such Affected Claim, and includes, without limitation, the transferee or assignee of an Affected Claim transferred and recognized as a Claimant in accordance with the Claims Procedure Order or a trustee, executor, liquidator, receiver, receiver and manager or other Person acting on behalf of or through such Person.
- "Affected Creditors' Class" has the meaning given to that term in Section 3.2 of the Consolidated CCAA Plan.
- "Affected Creditors' Distribution Cash Pool" has the meaning given to that term in Section 5.5 of the Consolidated CCAA Plan.
- "Aggregate Interest Amount" means the aggregate amount of interest to be paid on the Plan Implementation Date with respect to: (a) all Proven Claims (other than the Deemed Proven Claims, the Canadian Direct Purchaser Proven Claim and the Indirect Purchaser Proven Claim); and (b) all Unresolved Claims on the assumption (for calculation purposes only) that such Unresolved Claims will become Proven Claims in the full amount asserted by the holders of the Unresolved Claims in their respective Proofs of Claim; in each case calculated using the Applicable Interest Rate.
- "AGI-AGIF Payables" has the meaning given to that term in Step 26 in Schedule "B" of the Consolidated CCAA Plan.
- "AGI-AGIF Total Distribution Amount" means the amount determined by the formula (A+B) C, where A is the amount of the Unitholders' Distribution Cash Pool as of the Plan Implementation Date immediately prior to the completion of Step 30 of Schedule "B" of the Consolidated CCAA Plan, B is the aggregate of the amounts to be paid in satisfaction of the Proven Claims pursuant to Step 29 of Schedule "B" of the Consolidated CCAA Plan, C is the portion of the Available Funds held by the Monitor on

- behalf of the Fund immediately prior to the completion of Step 27 of Schedule "B" of the Consolidated CCAA Plan.
- "AGI-AGII Payables" has the meaning given to that term in Step 22 in Schedule "B" of the Consolidated CCAA Plan.
- "AGIF-AGI Payables" has the meaning given to that term in Step 26 in Schedule "B" of the Consolidated CCAA Plan.
- "AGII-AGI Payables" has the meaning given to that term in Step 22 in Schedule "B" of the Consolidated CCAA Plan.
- "AGII-AGI Total Distribution Amount" means the amount determined by the formula (A+B+C) D, where A is the amount of the Unitholders' Distribution Cash Pool as of the Plan Implementation Date immediately prior to the completion of Step 30 of Schedule "B" of the Consolidated CCAA Plan, B is the aggregate of the amounts to be paid in satisfaction of the Proven Claims pursuant to Step 29 of Schedule "B" of the Consolidated CCAA Plan, C is the aggregate of the amounts to be paid in satisfaction of the Proven Claims pursuant to Step 25 of Schedule "B" of the Consolidated CCAA Plan, and D is the portion of the Available Funds held by the Monitor on behalf of Arctic Glacier Inc. and the Fund immediately prior to the completion of Step 23 of Schedule "B" of the Consolidated CCAA Plan.
- "Applicable Interest Rate" means the rate of interest to be paid on each Proven Claim (other than the Deemed Proven Claims, the Canadian Direct Purchaser Proven Claim and the Indirect Purchaser Proven Claim), as such rate is set out in the Sanction Order.
- "Applicable Law" means, in respect of any Person, property, transaction, event or other matter, any law, statute, regulation, code, ordinance, principle of common law or equity, municipal by-law, treaty, or order, domestic or foreign, applicable to that Person, property, transaction, event or other matter and all applicable requirements, requests, official directives, rules, consents, approvals, authorizations, guidelines, and policies, in each case, having the force of law, of any Government Authority having or purporting to have authority over that Person, property, transaction, event or other matter and regarded by such Government Authority as requiring compliance.
- "Arctic Glacier Parties" has the meaning given to that term in the recitals hereto.
- "Asset Purchase Agreement" has the meaning given to that term in the recitals hereto.
- "Assets" has the meaning given to that term in the recitals hereto.
- "Assumed Liabilities" means the liabilities the Purchaser assumed, fulfilled, performed and discharged pursuant to Section 2.03 of the Asset Purchase Agreement.
- "Available Funds" means the total of (i) the proceeds of the sale or disposition of the Assets that have been paid by the Purchaser and are being held by the Monitor; (ii) the cash balances transferred by the Arctic Glacier Parties to the Monitor, in the hands of the Monitor at the Effective Time on the Plan Implementation Date; (iii) all other monies

- held by the Monitor, on behalf of the Arctic Glacier Parties, that are in the hands of the Monitor at the Effective Time on the Plan Implementation Date; and (iv) all monies received by the Monitor, on behalf of the Arctic Glacier Parties, following the Plan Implementation Date; less (v) the amount required to effect payment of the Recovered Fees on the Plan Implementation Date.
- "Beneficial Unitholder" means a holder of a beneficial interest in one or more Trust Units that are held by a Registered Unitholder for and on its behalf.
- "BIA" means the Bankruptcy and Insolvency Act, R.S.C. 1985, c. B-3, as amended.
- "Business Day" means a day, other than a Saturday or a Sunday, on which banks are generally open for business in Winnipeg, Manitoba.
- "Canadian Direct Purchaser Proven Claim" means an Affected Claim in favour of the Canadian Retail Litigation Claimants, as provided for in the Canadian Retail Litigation Settlement Agreement.
- "Canadian Retail Litigation Settlement Agreement" means the settlement agreement entered into as of May 4, 2011 between 1008021 Alberta Ltd., Louise Knowles c.o.b. as Special Event Marketing, Grand-Slam Concert, Productions Ltd., Arctic Glacier, Inc. and Reddy Ice Holdings, Inc., as approved by the Ontario Superior Court of Justice on July 11, 2013.
- "Canadian Retail Litigation Claimants" has the meaning ascribed to it in the Claims Procedure Order.
- "Canadian Vesting and Approval Order" has the meaning given to that term in the recitals hereto.
- "CCAA" has the meaning given to that term in the recitals hereto.
- "CCAA Court" has the meaning given to that term in the recitals hereto.
- "CCAA Proceedings" means the proceedings commenced by the Applicants in the CCAA Court at Winnipeg, Manitoba under Court File No. CI 12-01-76323.
- "CEPA Claim" means the Proven Claim of the California Environmental Protection Agency Department of Toxic Substance Control against Mountain Water Ice Company.
- "Chapter 15 Proceedings" means proceedings commenced by the Monitor in the State of Delaware in which the CCAA Proceedings have been recognized pursuant to Chapter 15 of the U.S. Bankruptcy Code.
- "Charges" means the Administration Charge, Directors' Charge, Critical Supplier Charge, Inter-Company Balances Charge and Class Counsel Charge.
- "Claim" means any right or claim of any Person, including an Equity Claim, that may be asserted or made in whole or in part against an Arctic Glacier Party, whether or not asserted or made, in connection with any indebtedness, liability or obligation of any kind

whatsoever, and any interest accrued thereon or costs payable in respect thereof, including by reason of the commission of a tort (intentional or unintentional), by reason of any breach of contract or other agreement (oral or written), by reason of any breach of duty (including any legal, statutory, equitable or fiduciary duty) or by reason of any right of ownership of or title to property or assets or right to a trust or deemed trust (statutory, express, implied, resulting, constructive or otherwise), and whether or not any indebtedness, liability or obligation is reduced to judgment, liquidated, unliquidated, fixed, contingent, matured, unmatured, disputed, undisputed, legal, equitable, secured, unsecured, perfected, unperfected, present or future, known or unknown, by guarantee, surety or otherwise, and whether or not any right or claim is executory or anticipatory in nature, including any right or ability of any Person (including Directors, Officers and Trustees) to advance a claim for contribution or indemnity or otherwise with respect to any matter, action, cause or chose in action, whether existing at present or commenced in the future, which indebtedness, liability or obligation, and any interest accrued thereon or costs payable in respect thereof (A) is based in whole or in part on facts arising prior to the Claims Bar Date (B) relates to a time period prior to the Claims Bar Date, or (C) is a right or claim of any kind that would be a claim provable in bankruptcy within the meaning of the BIA had the Arctic Glacier Party become bankrupt on the Claims Bar Date.

"Claimant" means any Person having an Affected Claim and includes the transferee or assignee of an Affected Claim or a trustee, executor, liquidator, receiver, receiver and manager, or other Person acting on behalf of or through any such Person.

"Claims Bar Date" means October 31, 2012.

"Claims Procedure Order" has the meaning given to that term in the recitals hereto.

"Claims Officer Order" has the meaning given to that term in the recitals hereto.

"Class Claim" has the meaning ascribed to it in the Claims Procedure Order.

"Class Counsel Charge" has the meaning given to that term in paragraph 6 of the Order made by the CCAA Court dated October 16, 2013, and titled the "Indirect Proven Claim Settlement Order".

"Class Representative" has the meaning ascribed to it in the Claims Procedure Order.

"Consolidated CCAA Plan" means this Plan of Compromise or Arrangement as amended, supplemented or restated from time to time in accordance with the terms hereof.

"CPS" means 7088418 Canada Inc. o/a Grandview Advisors and any successor thereto appointed by the CCAA Court.

"Creditors' Meeting" means the meeting of Affected Creditors that will be deemed to occur pursuant to the Meeting Order with a deemed vote of Affected Creditors in favour of a resolution to approve the Consolidated CCAA Plan.

- "Critical Supplier Charge" has the meaning given to that term in paragraph 36 of the Initial Order.
- "Crown Claims" has the meaning given to that term in Section 6.6 of the Consolidated CCAA Plan.
- "Declaration of Trust" means the Second Amended and Restated Declaration of Trust made as of December 6, 2004 among Robert Nagy, James E. Clark, Peter Hyndman, David Swaine and Gary Filmon, as Trustees, Laxus Holdings Inc., as Settlor, and the Registered Unitholders, as amended from time to time.
- "Deemed Proven Claims" means: (i) an Affected Claim in favour of the Direct Purchaser Claimants in the principal amount of US\$10,000,000 plus applicable interest against the Fund, Arctic Glacier Inc. and Arctic Glacier International Inc. at the interest rate set out in the Sanction Order; and (ii) the DOJ Claim.
- "Direct Purchaser Claim" means a Claim in favour of the members of the class(es) described in the statements of claim issued in the Direct Purchaser Litigation against the Fund, Arctic Glacier Inc. and Arctic Glacier International Inc.
- "Direct Purchaser Claimants" has the meaning ascribed to it in the Claims Procedure Order.
- "Direct Purchaser Litigation" means In re Packaged Ice Antitrust Litigation Direct Purchaser Class, as certified by the United States District Court for the Eastern District of Michigan on December 13, 2011 (Dkt. No. 406, 08-md-1952 E.D. Mich.).
- "Direct Purchaser Settlement Agreement" means the settlement agreement dated March 30, 2011 between the Fund, Arctic Glacier Inc., Arctic Glacier International Inc. and the Plaintiffs (as defined therein), as approved by the United States District Court for the Eastern District of Michigan on December 13, 2011.
- "Director" means any Person who is or was or may be deemed to be or have been, whether by statute, operation of law or otherwise, a director or *de facto* director of an Arctic Glacier Party.
- "Director's Charge" has the meaning given to that term in paragraph 40 of the Initial Order.
- "Distribution Claim" means with respect to: (i) each of the Deemed Proven Claims, the amount of each such Proven Claim, which shall include accrued interest calculated at the interest rates set out in the Sanction Order in respect of each such Proven Claim; (ii) the Canadian Direct Purchaser Proven Claim, the amount of such Proven Claim (iii) the Indirect Purchaser Proven Claim, the amount of such Proven Claim; and (iv) each other Affected Creditor's Proven Claim, the aggregate of each such Affected Creditor's Proven Claim and the applicable portion of the Aggregate Interest Amount in respect of such Proven Claim.

"Distribution Date" means any date from time to time set by the Monitor in accordance with the provisions of the Consolidated CCAA Plan, which shall include the Final Distribution Date, to effect distributions from the Available Funds to Affected Creditors in respect of Distribution Claims and/or distributions to Unitholders, other than distributions that occur on the Plan Implementation Date pursuant to Section 8.3 herein.

"DO&T Claim" means (i) any right or claim of any Person that might have been asserted or made in whole or in part against one or more Directors, Officers or Trustees that relates to a Claim for which such Directors, Officers or Trustees are by law liable to pay in their capacity as Directors, Officers or Trustees, or (ii) any right or claim of any Person that might have been asserted or made in whole or in part against one or more Directors, Officers or Trustees, in that capacity, whether or not asserted or made, in connection with any indebtedness, liability or obligation of any kind whatsoever, and any interest accrued thereon or costs payable in respect thereof, including by reason of the commission of a tort (intentional or unintentional), by reason of any breach of contract or other agreement (oral or written), by reason of any breach of duty (including any legal, statutory, equitable or fiduciary duty) or by reason of any right of ownership of or title to property or assets or right to a trust or deemed trust (statutory, express, implied, resulting, constructive or otherwise), and whether or not any indebtedness, liability or obligation, and any interest accrued thereon or costs payable in respect thereof, is reduced to judgment, liquidated, unliquidated, fixed, contingent, matured, unmatured, disputed, undisputed, legal, equitable, secured, unsecured, perfected, unperfected, present or future, known or unknown, by guarantee, surety or otherwise, and whether or not any right or claim is executory or anticipatory in nature, including any right or ability of any Person to advance a claim for contribution or indemnity from any such Directors, Officers or Trustees or otherwise with respect to any matter, action, cause or chose in action, whether existing at present or commenced in the future, which indebtedness, liability or obligation, and any interest accrued thereon or costs payable in respect thereof (A) is based in whole or in part on facts arising prior to the Claims Bar Date; or (B) relates to a time period prior to the Claims Bar Date, but not including an Excluded Claim.

"DO&T Indemnity Claim" means any existing or future right or claim of any Director, Officer or Trustee against an Arctic Glacier Party which arose or arises as a result of any Person filing a DO&T Proof of Claim in respect of such Director, Officer or Trustee for which such Director, Officer or Trustee is entitled to be indemnified by such Arctic Glacier Party.

"DO&T Indemnity Claims Bar Date" has the meaning set out in paragraph 21 of the Claims Procedure Order.

"DO&T Proof of Claim" means any Proof of Claim filed in respect of a DO&T Claim in accordance with the Claims Procedure Order.

"**DOJ Claim**" means an Affected Claim in favour of the United States Department of Justice against Arctic Glacier International Inc. in the amount of US\$7,032,046.96 as of July 9, 2012, plus applicable interest at the interest rate set out in the Sanction Order.

"Effective Time" means 12:01 a.m. on the Plan Implementation Date or such other time on such date as the Arctic Glacier Parties and the Monitor may agree.

"Equity Claim" has the meaning set forth in Section 2(1) of the CCAA.

"Excluded Claim" means:

- (a) Crown Claims;
- (b) any Claim entitled to the benefit of the Charges;
- (c) any Claim of an Arctic Glacier Party against another Arctic Glacier Party;
- (d) any Claim in respect of Assumed Liabilities; and
- (e) any Claim entitled to the benefit of any applicable insurance policy, excluding any such Claim or portion thereof that is recoverable as against an Arctic Glacier Party, Director, Officer or Trustee, as applicable.

"Filing Date" means February 22, 2012.

"Final Distribution Date" means the date determined by the Monitor, acting reasonably, following the payment in full or final reservation of all Administrative Reserve Costs and the resolution of all Unresolved Claims.

"Fund" has the meaning given to that term in the recitals hereto.

"Government Authority" means any governmental, regulatory or administrative authority, department, agency, commission, bureau, official, minister, board, panel, tribunal, Crown corporation, Crown ministry, court or dispute settlement panel or other law, rule or regulation-making or enforcing entity having or purporting to have jurisdiction on behalf of any nation, or province, territory or state or other subdivision thereof or any municipality, district or other subdivision thereof or other geographic or political subdivision of any of them or exercising, or entitled or purporting to exercise any administrative, executive, judicial, legislative, policy, regulatory or taxing authority or power.

"Indirect Purchaser Claim Settlement Agreement" means the settlement agreement entered into as of October 22, 2013, individually and on behalf of the Settlement Class (as defined in the Indirect Purchaser Claim Settlement Agreement), certain Arctic Glacier Parties and the Monitor, as approved by the U.S. Bankruptcy Court on February 27, 2014.

"Indirect Purchaser Claimants" has the meaning ascribed to it in the Claims Procedure Order.

"Indirect Purchaser Proven Claim" means an Affected Claim in favour of the Indirect Purchaser Claimants, as provided for in the Indirect Purchaser Claim Settlement Agreement, less certain noticing costs and the fees and expenses of UpShot Services LLC that have been paid by the Monitor, on behalf of the Applicants, in accordance with the Indirect Purchaser Settlement.

"Initial Order" has the meaning given to that term in the recitals hereto.

- "Insurance Deductible Reserve" has the meaning given to that term in Section 5.3 of the Consolidated CCAA Plan.
- "Inter-Company Balances Charge" has the meaning given to that term in paragraph 16 of the Initial Order.
- "IRC" means the Internal Revenue Code of 1986, as amended.
- "Meeting Order" means the Order of the CCAA Court under the CCAA that, among other things, sets the date for the Creditors' Meeting and the Unitholders' Meeting, as same may be amended, restated or varied from time to time.
- "Monitor" has the meaning given to that term in the recitals hereto.
- "Monitor's Website" means www.alvarezandmarsal.com/arctic-glacier-income-fund-arctic-glacier-inc.-and-subsidiaries.
- "Nominees" has the meaning given to that term in Section 6.2 of the Consolidated CCAA Plan.
- "Officer" means anyone who is or was or may be deemed to be or have been, whether by statute, operation of law or otherwise, an officer or *de facto* officer of an Arctic Glacier Party.
- "Person" is to be broadly interpreted and includes any individual, firm, corporation, limited or unlimited liability company, general or limited partnership, association, trust, unincorporated organization, joint venture, Government Authority or any agency, regulatory body, officer or instrumentality thereof or any other entity, wherever situate or domiciled, and whether or not having legal status, and whether acting on their own or in a representative capacity.
- "PID Charge Amount" has the meaning given to that term in Section 8.2 of the Consolidated CCAA Plan.
- "Plan Implementation Date" means the date on which the Consolidated CCAA Plan becomes effective, which shall be the Business Day on which the Monitor has filed with the CCAA Court a certificate confirming that all conditions to implementation of the Consolidated CCAA Plan pursuant to Section 10.3 have been satisfied or waived.
- "Plan Sanction Date" means the date the Sanction Order is made by the CCAA Court.
- "Pro Rata Share" means, in respect of the Unitholders' Distribution Cash Pool, the percentage that the Trust Units held by a Unitholder at the applicable Unitholder Distribution Record Date bears to the aggregate of all Trust Units, calculated as at the applicable Unitholder Distribution Record Date.
- "Proof of Claim" means any proof of claim in respect of an Affected Claim filed in accordance with the Claims Procedure Order.

- "Proven Claim" means each of the Deemed Proven Claims, the Canadian Direct Purchaser Proven Claim, the Indirect Purchaser Proven Claim and each Affected Claim that has been accepted as a proven Affected Claim by the Monitor or, if it was an Unresolved Claim, has been finally adjudicated in accordance with the Claims Officer Order, settled or accepted by the Monitor, in each case, for the amount settled, accepted or adjudicated as being owing.
- "Proven Claim Amount" has the meaning given to that term in Section 7.3 of the Consolidated CCAA Plan.
- "Purchase Price" has the meaning ascribed thereto in the Asset Purchase Agreement.
- "Purchaser" has the meaning given to that term in the recitals hereto.
- "Recognition Order" means an order of the U.S. Bankruptcy Court recognizing an Order of the CCAA Court in the Chapter 15 Proceedings.
- "Recovered Fees" has the meaning given to that term in Section 8.3 of the Consolidated CCAA Plan.
- "Registered Unitholder" means, as of the Unitholder Record Date, each holder of one or more Trust Units that, at such time, are outstanding and entitled to the benefits of the Declaration of Trust, as shown on the register of such holders maintained by the Transfer Agent or by the Trustees on behalf of the Fund.
- "Releasees" has the meaning given to that term in Section 9.1 of the Consolidated CCAA Plan.
- "Required Unitholder Majority" has the meaning given to that term in Section 4.5 of the Consolidated CCAA Plan.
- "Return of Capital Amount" has the meaning given to that term in Step 28 in Schedule "B" of the Consolidated CCAA Plan.
- "Sanction Order" means an order by the CCAA Court which, among other things, shall sanction and approve the Consolidated CCAA Plan under the CCAA and shall include provisions as may be necessary or appropriate to give effect to the Consolidated CCAA Plan, including provisions in substance similar to those set out in Section 10.2 of the Consolidated CCAA Plan.
- "SISP" has the meaning given that term in the recitals hereto.
- "Step 3 Companies" has the meaning given to that term in Step 3 in Schedule "B" of the Consolidated CCAA Plan.
- "Step 10 Companies" has the meaning given to that term in Step 10 in Schedule "B" of the Consolidated CCAA Plan.
- "Step 13 Companies" has the meaning given to that term in Step 13 in Schedule "B" of the Consolidated CCAA Plan.

"Step 17 Companies" has the meaning given to that term in Step 13 in Schedule "B" of the Consolidated CCAA Plan.

"Tax Statutes" means all legislative or administrative enactments governing federal, state, local, or foreign income, premium, property (real or personal), sales, excise, employment, payroll, withholding, gross receipts, license, severance, stamp, occupation, windfall profits, environmental, customs duties, capital stock, franchise, profits, social security (or similar, including FICA), unemployment, disability, use, transfer, registration, value added, alternative or add-on minimum, estimated or other tax of any kind or any charge of any kind in the nature of (or similar to) taxes whatsoever, including any interest, penalty or addition thereto, including, without limiting the generality of the foregoing, the IRC, section 159 of the *Income Tax Act* (Canada), section 270 of the Excise Tax Act (Canada); section 117 of the Taxation Act, 2007 (Ontario); section 107 of the Corporations Tax Act (Ontario); section 22 of the Retail Sales Tax Act (Ontario); section 34 of the *Income Tax Act* (British Columbia); section 222 of the *Provincial Sales* Tax Act (British Columbia); section 49 of the Alberta Corporate Tax Act; section 85 of the Income Tax Act, 2000 (Saskatchewan); section 48 of the Revenue and Financial Services Act (Saskatchewan); section 22 of the Income Tax Act (Manitoba); section 73 of the Tax Administration and Miscellaneous Taxes Act (Manitoba); section 14 of the Tax Administration Act (Quebec); and section 313 of the Act Respecting the Quebec Sales Tax.

"Transfer Agent" means such company as may from time to time be appointed by the Fund to act as registrar and transfer agent of the Trust Units, together with any subtransfer agent duly appointed by the Transfer Agent.

"Transferred Shares" has the meaning given to that term in Step 6 in Schedule "B" of the Consolidated CCAA Plan.

"Trust Unit" means, as of the Unitholder Record Date or the applicable Unitholder Distribution Record Date, as the case may be, each trust unit of the Fund authorized and issued under the Declaration of Trust that, at such time, is outstanding and entitled to the benefits of the Declaration of Trust.

"Trustee" means any Person who is or was or may be deemed to be or have been, whether by statute, operation of law or otherwise, a trustee or *de facto* trustee of the Fund, in such capacity and includes James E. Clark, David Swaine and Gary Filmon.

"Unitholder Distribution" has the meaning given to that term in Section 6.2 of the Consolidated CCAA Plan.

"Unitholder Distribution Record Date" means the date(s) determined from time to time by the Monitor that are, in each case, at least 21 days prior to a contemplated Unitholder Distribution including, without limitation, the contemplated Unitholder Distribution on the Plan Implementation Date.

"Unitholder Record Date" means June 16, 2014.

- "Unitholders" means, collectively, (a) each Registered Unitholder that holds one or more Trust Units solely for and on behalf of itself; and (b) each Beneficial Unitholder.
- "Unitholders' Distribution Cash Pool" has the meaning given to that term in Section 5.6 of the Consolidated CCAA Plan.
- "Unitholders' Meeting" means a meeting of Unitholders held pursuant to the Meeting Order to consider and vote on a resolution to approve the Consolidated CCAA Plan and any other matters related to the Consolidated CCAA Plan or its implementation.
- "Unresolved Claim" means an Affected Claim, in the amount specified in the corresponding Proof of Claim, that has not been finally determined as a Proven Claim in accordance with the Claims Procedure Order, the Claims Officer Order and the Meeting Order.
- "Unresolved Claims Reserve" has the meaning given to that term in Section 5.4 of the Consolidated CCAA Plan.
- "U.S. Bankruptcy Code" means Title 11 of the United States Code.
- "U.S. Bankruptcy Court" means the U.S. Bankruptcy Court for the District of Delaware.
- "Withholding Obligation" has the meaning given to that term in Section 6.13 of the Consolidated CCAA Plan.

1.2 Certain Rules of Interpretation

For the purposes of the Consolidated CCAA Plan:

- (a) any reference in the Consolidated CCAA Plan to a contract, instrument, release, indenture, or other agreement or document being in a particular form or on particular terms and conditions means that such document shall be substantially in such form or substantially on such terms and conditions;
- (b) any reference in the Consolidated CCAA Plan to an Order or an existing document or exhibit filed or to be filed means such Order, document or exhibit as it may have been or may be amended, modified, or supplemented;
- (c) unless otherwise specified, all references to currency are to Canadian dollars;
- (d) the division of the Consolidated CCAA Plan into "Articles" and "Sections" and the insertion of a table of contents are for convenience of reference only and do not affect the construction or interpretation of the Consolidated CCAA Plan, nor are the descriptive headings of "Articles" and "Sections" intended as complete or accurate descriptions of the content thereof;
- (e) the use of words in the singular or plural, or with a particular gender, including a definition, shall not limit the scope or exclude the application of any provision

- of the Consolidated CCAA Plan or a Schedule hereto to such Person (or Persons) or circumstances as the context otherwise permits;
- (f) the words "includes" and "including" and similar terms of inclusion shall not, unless expressly modified by the words "only" or "solely", be construed as terms of limitation, but rather shall mean "includes but is not limited to" and "including but not limited to", so that references to included matters shall be regarded as illustrative without being either characterizing or exhaustive;
- (g) unless otherwise specified, all references as to time herein and any document issued pursuant hereto shall mean local time in Winnipeg, Manitoba, Canada, and any reference to an event occurring on a Business Day shall mean prior to 5:00 p.m. CST or CDT, as the case may be, on such Business Day;
- (h) unless otherwise specified, time periods within or following which any payment is to be made or act is to be done shall be calculated by excluding the day on which the period commences and including the day on which the period ends and by extending the period to the next succeeding Business Day if the last day of the period is not a Business Day;
- (i) unless otherwise provided, any reference to the U.S. Bankruptcy Code and to a statute or other enactment of parliament or a legislature includes all regulations made thereunder, all amendments to or re-enactments of such statute or regulations in force from time to time, and, if applicable, any statute or regulation that supplements or supersedes such statute or regulation;
- references to a specified "Article" or "Section" shall, unless something in the subject matter or context is inconsistent therewith, be construed as references to that specified Article or Section of the Consolidated CCAA Plan, whereas the terms "the Consolidated CCAA Plan", "hereof", "herein", "hereto", "hereunder" and similar expressions shall be deemed to refer generally to the Consolidated CCAA Plan and not to any particular "article", "section" or other portion of the Consolidated CCAA Plan and include any documents supplemental hereto; and
- (k) the word "or" is not exclusive.

1.3 Successors and Assigns

The Consolidated CCAA Plan shall be binding upon and shall enure to the benefit of the heirs, administrators, executors, legal representatives, successors and assigns of any Person or party named or referred to in the Consolidated CCAA Plan, including the Arctic Glacier Parties, all Affected Creditors, the Directors and Officers, the Unitholders, the Trustees and the Releasees.

1.4 Governing Law

The Consolidated CCAA Plan shall be governed by and construed in accordance with the laws of the Province of Manitoba and the federal laws of Canada applicable therein. All

questions as to the interpretation or application of the Consolidated CCAA Plan and all proceedings taken in connection with the Consolidated CCAA Plan and its provisions shall be subject to the exclusive jurisdiction of the CCAA Court.

1.5 Schedules

The following are the Schedules to the Consolidated CCAA Plan, which are incorporated by reference into the Consolidated CCAA Plan and form a part of it:

Schedule "A"

Additional Applicants

Schedule "B"

Specified Plan Implementation Date Steps

ARTICLE 2 PURPOSE AND EFFECT OF THE CONSOLIDATED CCAA PLAN

2.1 Purpose

The purpose of the Consolidated CCAA Plan is to:

- (a) permit the settlement and/or determination of all Affected Claims in accordance with the Claims Procedure Order and the Claims Officer Order;
- (b) provide for the distribution of a sufficient amount of the Available Funds to holders of Proven Claims to satisfy such Proven Claims in full (plus applicable interest, if any, calculated at the interest rate set out in the Sanction Order);
- (c) provide for the distribution of any surplus of the Available Funds to each Unitholder, in the amount of their Pro Rata Share, free and clear of any Claims of Affected Creditors; and
- (d) effect the wind-up and dissolution of certain of the Arctic Glacier Parties pursuant to and in accordance with the timing and manner set out in the Consolidated CCAA Plan.

2.2 Persons Affected

The Consolidated CCAA Plan provides for the complete satisfaction of all Proven Claims of Affected Creditors, plus payment of applicable interest, if any, calculated at the interest rate set out in the Sanction Order, in respect of such Proven Claims. The Consolidated CCAA Plan also provides for distributions from time to time to Unitholders from the Unitholders' Distribution Cash Pool based on each Unitholder's Pro Rata Share to the extent that there are Available Funds to fund such distribution, following which the Trust Units will be terminated and the Fund shall cease to be listed and traded on the Canadian National Stock Exchange. The Consolidated CCAA Plan will become effective at the Effective Time on the Plan Implementation Date and shall be binding on and enure to the benefit of the Arctic Glacier Parties, the Affected Creditors, the Directors and Officers, the Unitholders, the Trustees and all other Persons named or referred to in, or subject to, the Consolidated CCAA Plan.

2.3 Persons Not Affected

For greater certainty, the Consolidated CCAA Plan does not affect the holders of Excluded Claims with respect to and to the extent of their Excluded Claims. Nothing in the Consolidated CCAA Plan shall affect the Arctic Glacier Parties' rights and defences, both legal and equitable, with respect to any Excluded Claims, including, but not limited to, all rights with respect to legal and equitable defences or entitlements to set-offs or recoupment against such Excluded Claims.

ARTICLE 3 CLASSIFICATION OF CREDITORS, VOTING AND RELATED MATTERS

3.1 Claims Procedure

The procedure for determining the validity and quantum of the Affected Claims for voting and distribution purposes under the Consolidated CCAA Plan shall be governed by the Claims Procedure Order, the Claims Officer Order, the Meeting Order, the CCAA and the Consolidated CCAA Plan.

3.2 Classification of Creditors

For the purposes of voting on the Consolidated CCAA Plan, there will be one consolidated class of creditors, which will be composed of all of the Affected Creditors (the "Affected Creditors' Class").

3.3 Claims of Affected Creditors

Affected Creditors shall:

- (a) prove their Affected Claims in accordance with the Claims Procedure Order and the Claims Officer Order;
- (b) be deemed to vote their Proven Claims or Unresolved Claims, as the case may be, at the Creditors' Meeting in favour of the resolution to approve the Consolidated CCAA Plan; and
- (c) receive the rights and distributions provided for under and pursuant to the Consolidated CCAA Plan and the Sanction Order.

3.4 Creditors' Meeting

The Creditors' Meeting shall be held in accordance with the Consolidated CCAA Plan, the Meeting Order, the Claims Procedure Order and the Claims Officer Order. Pursuant to the Meeting Order, the Creditors' Meeting shall be deemed to have been duly called and held on August 11, 2014 and every Affected Creditor shall be deemed to have voted in favour of a resolution to approve the Consolidated CCAA Plan.

3.5 Voting

Pursuant to the Meeting Order: (a) the Affected Creditors' Class shall be deemed to have voted in favour of a resolution to approve the Consolidated CCAA Plan at the Creditors' Meeting on August 11, 2014; and (b) the vote on the Consolidated CCAA Plan at the Creditors' Meeting shall be deemed to have been decided unanimously in favour of the resolution to approve the Consolidated CCAA Plan.

3.6 Guarantees and Similar Covenants

No Person who has a Claim under any guarantee, surety, indemnity or similar covenant in respect of any Claim which is affected pursuant to the Consolidated CCAA Plan or who has any right to claim over in respect of or to be subrogated to the rights of any Person in respect of a Claim which is affected pursuant to the Consolidated CCAA Plan shall be entitled to any greater rights as against the Arctic Glacier Parties than the Person whose Claim is affected pursuant to the Consolidated CCAA Plan.

3.7 Set-Off

The law of set-off applies to all Affected Claims.

ARTICLE 4 CLASSIFICATION OF UNITHOLDERS, VOTING AND RELATED MATTERS

4.1 Unitholder Procedure

The procedure for determining the amount of Trust Units held by each Unitholder for voting and distribution purposes under the Consolidated CCAA Plan shall be governed by the Meeting Order, the CCAA and the Consolidated CCAA Plan.

4.2 Classification of Unitholders

For the purposes of considering and voting on the Consolidated CCAA Plan, there will be one consolidated class of Unitholders, which shall be comprised of Unitholders as at the Unitholder Record Date.

4.3 Unitholders' Meeting

The Unitholders' Meeting will be called and held on August 11, 2014 pursuant to the Meeting Order for the purpose of considering and voting on the Consolidated CCAA Plan. The resolution to, among other things, approve the Consolidated CCAA Plan will be passed if it receives an affirmative vote of the Required Unitholder Majority. Notice of the Unitholders' Meeting will be provided to all Unitholders as at Unitholder Record Date.

The quorum required at the Unitholders' Meeting shall be one Registered Unitholder or Beneficial Unitholder present at such meeting in person or by proxy and entitled to vote on the resolution to approve, among other things, the Consolidated CCAA Plan.

4.4 Voting

Each Unitholder shall be entitled to one vote for each Trust Unit held by such Unitholder on the Unitholder Record Date which, if voted in person or by proxy at the Unitholders' Meeting, shall be recorded as a vote for or against the Consolidated CCAA Plan, as the case may be.

4.5 Approval by Unitholders

The proposed resolution to approve the Consolidated CCAA Plan must receive the affirmative votes of more than 66 2/3% of the votes attached to Trust Units represented at the Unitholders' Meeting and cast in accordance with the Meeting Order (the "Required Unitholder Majority").

4.6 Guarantees and Similar Covenants

No Person who holds an interest in the Trust Units under any guarantee, surety, indemnity or similar covenant in respect of the Trust Units or who has any right to claim over in respect of or to be subrogated to the rights of any Unitholder in respect of the Trust Units being affected pursuant to the Consolidated CCAA Plan shall be entitled to any greater rights as against the Arctic Glacier Parties than the Unitholders.

ARTICLE 5 AVAILABLE FUNDS, RESERVES AND CASH POOLS

5.1 Available Funds

The Monitor shall hold the Available Funds, on behalf of the Arctic Glacier Parties, in one or more separate interest-bearing accounts for each of the following reserves and pools (each as more particularly described herein): (a) Administrative Costs Reserve; (b) Insurance Deductible Reserve; (c) Unresolved Claims Reserve; (d) Affected Creditors' Distribution Cash Pool; and (e) Unitholders' Distribution Cash Pool.

5.2 Administrative Costs Reserve

On the Plan Implementation Date and in accordance with the Plan Implementation Date steps and transactions set out in Section 8.3 of the Consolidated CCAA Plan, an administrative costs reserve (the "Administrative Costs Reserve") shall be established out of the Available Funds in the amount of US\$10,000,000, which is to be held by the Monitor, on behalf of the Arctic Glacier Parties, for the purpose of paying the Administrative Reserve Costs in accordance with the Consolidated CCAA Plan.

5.3 Insurance Deductible Reserve

On the Plan Implementation Date and in accordance with the Plan Implementation Date steps and transactions set out in Section 8.3 of the Consolidated CCAA Plan, an insurance deductible reserve (the "Insurance Deductible Reserve") shall be established out of the Available Funds in the amount of US\$850,000, which is to be held by the Monitor, on behalf of the Arctic Glacier Parties, for the purpose of covering the payment of the deductible portion of the run-off of any litigation covered by insurance.

The quantum of the Insurance Deductible Reserve has been agreed to with the insurer and is intended to cover: (i) the deductible amounts currently outstanding as determined by the Monitor, in consultation with the Arctic Glacier Parties; (ii) deductible amounts that may become payable in respect of currently open claims as determined by the Monitor, in consultation with the Arctic Glacier Parties; and (iii) based on historical claim rates, deductible amounts for further claims related to the period prior to July 27, 2012 that have not yet been filed with the Monitor.

Any final remaining balance in the Insurance Deductible Reserve, as determined by the Monitor, will be deemed to have been transferred to the Administrative Costs Reserve on such date as is determined by the Monitor.

If an agreement is reached between the Monitor, on behalf of the Arctic Glacier Parties, and the insurer of the Arctic Glacier Parties with respect to the purchase of a "buy-out" policy (as an alternative to holding the Insurance Deductible Reserve), then the required payment by the Arctic Glacier Parties for such "buy-out" policy shall be paid by the Monitor, on behalf of the Arctic Glacier Parties, to the insurer of the Arctic Glacier Parties using funds in the Insurance Deductible Reserve. Following the completion of such purchase, any remaining balance in the Insurance Deductible Reserve will be deemed to have been transferred to the Administrative Costs Reserve on such date as is determined by the Monitor.

The Monitor shall have no obligation to make any payment out of the Insurance Deductible Reserve, and nothing in the Consolidated CCAA Plan, the Meeting Order or the Sanction Order shall be construed as obligating the Monitor to make any such payment if, in the Monitor's sole and unfettered discretion, the cost of making any such payment is prohibitive for so doing in relation to the quantum of the contemplated payment.

5.4 Unresolved Claims Reserve

On the Plan Implementation Date and in accordance with the Plan Implementation Date steps and transactions set out in Section 8.3 of the Consolidated CCAA Plan, an unresolved claims reserve (the "Unresolved Claims Reserve") shall be established out of the Available Funds and be held by the Monitor, on behalf of the Arctic Glacier Parties, in escrow in accordance with the Consolidated CCAA Plan in an amount equal to (a) the aggregate amount that would have been paid to all Affected Creditors holding Unresolved Claims in accordance with the Consolidated CCAA Plan (calculated on the basis of the amounts specified in such Affected Creditors' Proofs of Claim) if such Unresolved Claims had been Proven Claims on the Plan Implementation Date; and (b) the applicable portion of the Aggregate Interest Amount in respect of such Unresolved Claims.

5.5 Composition of the Affected Creditors' Distribution Cash Pool

On the Plan Implementation Date, an Affected Creditors' distribution cash pool (the "Affected Creditors' Distribution Cash Pool") shall be established from the Available Funds in an amount equal to:

(a) all Proven Claims of Affected Creditors with Affected Claims denominated in Canadian dollars on the Plan Implementation Date plus the applicable portion of the Aggregate Interest Amount in respect of such Proven Claims (save and except for the Canadian Direct Purchaser Proven Claim); and

(b) all Proven Claims of Affected Creditors with Affected Claims denominated in United States dollars on the Plan Implementation Date plus the applicable portion of the Aggregate Interest Amount in respect of such Proven Claims (save and except for the Deemed Proven Claims, which shall include accrued interest calculated at the interest rates set out in the Sanction Order in respect of each such Proven Claims, and the Indirect Purchaser Proven Claim).

The Monitor shall hold the monies in the Affected Creditors' Distribution Cash Pool, on behalf of the Arctic Glacier Parties, in escrow for distribution to Affected Creditors with Proven Claims pursuant to and in accordance with the Consolidated CCAA Plan. The Available Funds in the Affected Creditors' Distribution Cash Pool shall be denominated in Canadian dollars or United States dollars depending upon whether the Proven Claim is denominated in Canadian dollars or United States dollars.

5.6 Composition of the Unitholders' Distribution Cash Pool

On the Plan Implementation Date, a Unitholders' distribution cash pool (the "Unitholders' Distribution Cash Pool") shall be established out of the Available Funds in an amount equal to the Available Funds less the amounts used to fund the: (a) Administrative Costs Reserve; (b) Insurance Deductible Reserve; (c) Unresolved Claims Reserve; and (d) Affected Creditors' Distribution Cash Pool. The Monitor shall hold the Unitholders' Distribution Cash Pool in a separate interest-bearing account in escrow for distribution to the Unitholders in accordance with the Consolidated CCAA Plan.

5.7 Remaining Funds

Any final remaining balance in the Administrative Costs Reserve or the Unitholders' Distribution Cash Pool that have not been distributed by the Final Distribution Date on account of the cost of making any such distribution being prohibitive for so doing in relation to the quantum of the distribution contemplated in the Consolidated CCAA Plan will be paid to a charity in Winnipeg, Manitoba that will be determined at a later date.

ARTICLE 6 PROVISIONS REGARDING DISTRIBUTIONS AND PAYMENTS

6.1 Distributions from the Affected Creditors' Distribution Cash Pool

The Affected Creditors' Distribution Cash Pool shall be distributed by the Monitor, on behalf and for the account of the Arctic Glacier Parties, on the Plan Implementation Date or on any Distribution Date, as the case may be, to each Affected Creditor in the amount of such Affected Creditor's Distribution Claim by way of cheque sent by prepaid ordinary mail to the address for such Affected Creditor specified in the Proof of Claim filed by such Affected Creditor.

Following the distribution to be made by the Monitor, on behalf of the Arctic Glacier Parties, to Affected Creditors on the Plan Implementation Date pursuant to, and in accordance with, Section 8.3 of the Consolidated CCAA Plan, the Monitor shall have no further obligation to make any payment out of the Affected Creditors' Distribution Cash Pool, and nothing in the Consolidated CCAA Plan, the Meeting Order or the Sanction Order shall be construed as

obligating the Monitor to make any such payment if, in the Monitor's sole and unfettered discretion, the cost of making any such payment is prohibitive for so doing in relation to the quantum of the contemplated payment.

6.2 Distributions from the Unitholders' Distribution Cash Pool

The Monitor shall declare a Unitholder Distribution Record Date prior to any distribution, deemed or otherwise, from the Unitholders' Distribution Cash Pool. On the Plan Implementation Date or on any Distribution Date, as the case may be, the Monitor shall transfer amounts as determined by the Monitor in accordance with the Consolidated CCAA Plan, on behalf and for the account of the Fund, from the Unitholders' Distribution Cash Pool (each such transfer being a "Unitholder Distribution") to the Transfer Agent. As soon as reasonably practicable, and in no event later than five (5) Business Days following receipt of the Unitholder Distribution, the Transfer Agent shall distribute each Unitholder Distribution, on behalf and for the account of the Fund, by way of cheque sent by prepaid ordinary mail or by way of wire transfer to each Registered Unitholder, as of the applicable Unitholder Distribution Record Date that the Transfer Agent is aware of and has contact information in respect of, based on each Registered Unitholder's Pro Rata Share (a) for such Registered Unitholder, in respect of Trust Units held by such Registered Unitholder solely for and on behalf of itself, as applicable; or (b) for distribution by such Registered Unitholder to (i) Beneficial Unitholders, as applicable, or (ii) participant holders of the Trust Units or the intermediary holders of the Trust Units (collectively, the "Nominees"), or the agents of such Nominees for subsequent distribution to the applicable Beneficial Unitholders.

The Monitor shall have no obligation to make any payment out of the Unitholders' Distribution Cash Pool, and nothing in the Consolidated CCAA Plan, the Meeting Order or the Sanction Order shall be construed as obligating the Monitor to make any such payment if, in the Monitor's sole and unfettered discretion, the cost of making any such payment is prohibitive for so doing in relation to the quantum of the contemplated payment.

6.3 Payment of Administrative Reserve Costs

On the Plan Implementation Date, the Administrative Costs Reserve will be funded in accordance with Section 5.2 of the Consolidated CCAA Plan and shall be administered in accordance with the Consolidated CCAA Plan.

Any final remaining balance in the Administrative Costs Reserve following (a) payment in full or final reservation of all Administrative Reserve Costs, as determined by the Monitor; and (b) declaration by the Monitor of a Unitholder Distribution Record Date; shall be transferred by the Monitor to the Transfer Agent and shall be deemed to have first been transferred to the Unitholders' Distribution Cash Pool and then distributed therefrom by the Monitor, on behalf of the Fund, to the Transfer Agent. As soon as reasonably practicable and in no event later than five (5) Business Days following its receipt, such remaining final balance shall then be distributed by the Transfer Agent, on behalf and for the account of the Fund, to each Registered Unitholder, as of the applicable Unitholder Distribution Record Date that the Transfer Agent is aware of and has contact information in respect of in the manner prescribed in Section 6.2 herein based on each Registered Unitholder's Pro Rata Share (a) for such Registered Unitholder, in respect of Trust Units held by such Registered Unitholder solely for and on behalf of itself, as applicable; or (b) for distribution by such Registered Unitholder to (i) Beneficial Unitholders, as

applicable, or (ii) Nominees, or the agents of such Nominees for subsequent distribution to the applicable Beneficial Unitholders.

The Monitor shall have no obligation to make any payment or transfer out of the Administrative Costs Reserve, and nothing in the Consolidated CCAA Plan, the Meeting Order or the Sanction Order shall be construed as obligating the Monitor to make any such payment if, in the Monitor's sole and unfettered discretion, the cost of making any such payment is prohibitive for so doing in relation to the quantum of the contemplated payment.

6.4 Payment of Insurance Deductible Reserve Costs

On the Plan Implementation Date, the Insurance Deductible Reserve will be funded in accordance with Section 5.3 of the Consolidated CCAA Plan and shall be administered in accordance with the Consolidated CCAA Plan.

6.5 Cancellation of Instruments Evidencing Affected Claims

Following completion of the steps and transactions in the sequence set forth in Section 8.3 of the Consolidated CCAA Plan, all agreements, invoices and other instruments evidencing Affected Claims will not entitle any holder thereof to any compensation or participation other than as expressly provided for in the Consolidated CCAA Plan and will be cancelled and will be null and void.

6.6 Crown Priority Claims

Within six (6) months after the Plan Sanction Date, the Monitor, on behalf of the Arctic Glacier Parties, shall pay in full to Her Majesty in Right of Canada or any province all amounts of a kind that could be subject to a demand under Section 6(3) of the CCAA that were outstanding on the Filing Date and which have not been paid by the Plan Implementation Date ("Crown Claims").

6.7 Currency

Unless specifically provided for in the Consolidated CCAA Plan or the Sanction Order, for the purposes of distribution, an Affected Claim shall be denominated in the currency in which it is owed and all payments and distributions to the Affected Creditors on account of their Affected Claims shall be made in the currency in which they are owed. To the extent that there are insufficient funds to pay an Affected Claim in the currency in which it is owed, the Monitor shall be authorized to convert the currency on a date that is within five (5) Business Days of the Plan Implementation Date or any Distribution Date, as the case may be.

6.8 Interest

The interest rate that will be used to calculate the quantum of the Deemed Proven Claims and the Aggregate Interest Amount in respect of each other Proven Claim (save and except for the Canadian Direct Purchaser Proven Claim and the Indirect Purchaser Proven Claim) will be specified in the Sanction Order.

6.9 Treatment of Undeliverable Distributions

If any Affected Creditor's distribution by way of cheque is returned as undeliverable or is not cashed, no further distributions to such Affected Creditor shall be made unless and until the Arctic Glacier Parties and the Monitor are notified by such Affected Creditor of such Affected Creditor's current address, at which time all such distributions shall be made to such Affected Creditor without interest accruing on account of the cheque being undeliverable or not cashed. All claims for undeliverable or uncashed distributions in respect of Proven Claims will expire six (6) months after the date of such distribution, after which date the Proven Claims of any Affected Creditor or successor of such Affected Creditor with respect to such unclaimed or uncashed distributions shall be forever discharged and forever barred, without any compensation therefor, notwithstanding any federal, state or provincial laws to the contrary, at which time the cash amount held by the Monitor in relation to such Proven Claims will be, or will be deemed to be, transferred to the Administrative Costs Reserve, and will be distributed in accordance with the terms of the Consolidated CCAA Plan. Nothing contained in the Consolidated CCAA Plan shall require the Arctic Glacier Parties or the Monitor to attempt to locate any Affected Creditor.

If any distribution to a Registered Unitholder by way of cheque is returned as undeliverable or is not cashed, no further distributions to such Registered Unitholder shall be effected unless and until the Arctic Glacier Parties, the Monitor and the Transfer Agent are notified by or on behalf of such Registered Unitholder of such Registered Unitholder's current address, at which time all such distributions shall be effected towards such Registered Unitholder without interest. All claims for undeliverable or uncashed distributions to a Registered Unitholder will expire six (6) months after the date of such distribution, after which date the entitlement of any Registered Unitholder, as provided for in this Consolidated CCAA Plan, or of any successor of such Registered Unitholder with respect to such unclaimed or uncashed distribution shall be forever discharged and forever barred, without any compensation therefor, notwithstanding any federal, state or provincial laws to the contrary, at which time the cash amount held by the Transfer Agent in relation to such distribution will be transferred by the Transfer Agent to the Monitor and shall be held by the Monitor, on behalf of the Arctic Glacier Parties, in the Administrative Costs Reserve, and will be distributed in accordance with the terms of the Consolidated CCAA Plan. Nothing contained in the Consolidated CCAA Plan shall require the Arctic Glacier Parties, the Trustees, the Transfer Agent or the Monitor to attempt to locate any Registered Unitholder.

6.10 Assignment of Claims for Voting and Distribution Purposes

(a) Assignment of Claims Prior to the Creditors' Meeting

Subject to any restrictions contained in Applicable Laws, Affected Creditors may transfer or assign the whole of their Claims (or where a Claim includes an indemnity claim, the whole of their Claims other than that part of the Claim relative to the indemnity) prior to the Creditors' Meeting provided that the Arctic Glacier Parties and the Monitor shall not be obliged to deal with any transferee or assignment, together with satisfactory evidence of such transfer or assignment has been given to the Arctic Glacier Parties and the Monitor by 5:00 p.m. (Toronto time) on the day that is at least five (5) Business Days immediately prior to the Creditors' Meeting, or such other date as the Monitor may agree. In the event of such notice of transfer or assignment prior to the Creditors' Meeting, the transferee or assignee shall, for all purposes, be

treated as the Affected Creditor of the assigned or transferred Claim, will be bound by any and all notices previously given to the transferor or assignor in respect of such Claim and shall be bound, in all respects, by any and all notices given and by the Orders of the CCAA Court in the CCAA Proceedings. For greater certainty, other than as described above, the Arctic Glacier Parties shall not recognize partial transfers or assignments of Claims.

(b) Assignment of Claims Subsequent to the Creditors' Meeting

Subject to any restrictions contained in Applicable Laws, Affected Creditors may transfer or assign the whole of their Claims (or where a Claim includes an indemnity claim, the whole of their Claims other than that part of the Claim relative to the indemnity) after the Creditors' Meeting provided that the Arctic Glacier Parties and the Monitor shall not be obliged to deal with any transferee or assignee as an Affected Creditor and the Monitor shall not be obliged to make any distributions to the transferee or assignee in respect thereof unless and until actual notice of the transfer or assignment, together with evidence of the transfer or assignment and a letter of direction executed by the transferor or assignor, all satisfactory to the Arctic Glacier Parties and the Monitor, has been given to the Arctic Glacier Parties and the Monitor by 5:00 p.m. on the day that is at least five (5) Business Days immediately prior to the Plan Implementation Date or any Distribution Date(s), as the case may be, or such other date as the Monitor may agree. Thereafter, the transferee or assignee shall, for all purposes, be treated as the Affected Creditor of the assigned or transferred Claim, will be bound by any notices previously given to the transferor or assignor in respect of such Claim and shall be bound, in all respects, by notices given and steps taken, and by the orders of the CCAA Court in the CCAA Proceedings. For greater certainty, other than as described above, the Arctic Glacier Parties shall not recognize partial transfers or assignments of Claims.

6.11 Assignment of Trust Units for Voting Purposes

Subject to any restrictions contained in Applicable Laws, Unitholders may transfer or assign their Trust Units provided that the Arctic Glacier Parties, the Transfer Agent and the Monitor shall not be obliged to deal with any transferee or assignee of a Unitholder in respect thereof for purposes of their eligibility to consider and vote on the Consolidated CCAA Plan unless and until actual notice of the transfer or assignment, together with satisfactory evidence of such transfer or assignment has been given to and received by the Arctic Glacier Parties, the Transfer Agent and the Monitor by 5:00 p.m. (Toronto time) on the day immediately prior to the Unitholder Record Date. In the event of receipt of such notice of transfer or assignment prior to the Unitholder Record Date (as provided for in the immediately preceding sentence), the transferee or assignee shall, for all purposes be treated as the Unitholder of the assigned or transferred Trust Units, will be bound by any and all notices previously given to the transferor or assignor in respect of such Trust Units and shall be bound, in all respects, by any and all notices given and steps taken, and by the Orders of the CCAA Court in the CCAA Proceedings. For greater certainty, the Arctic Glacier Parties and the Transfer Agent shall not recognize partial transfers or assignments of Trust Units. In addition, under no circumstances shall the Arctic Glacier Parties, the Transfer Agent and the Monitor be obliged to deal with any transferee or assignee of a Unitholder for purposes of their eligibility to consider and vote on the Consolidated CCAA Plan who are not reflected as a Unitholder on the Unitholder Record Date.

6.12 Allocation of Distributions

All distributions made by the Monitor, on behalf of the Arctic Glacier Parties, pursuant to the Consolidated CCAA Plan shall be first in consideration for the outstanding principal amount of the Claims and secondly in consideration for accrued and unpaid interest and penalties, if any, which forms part of such Claims.

6.13 Withholding and Reporting Requirements

The Arctic Glacier Parties and the Monitor shall be entitled to deduct and withhold, or direct the Transfer Agent to deduct and withhold, from any distribution, payment or consideration otherwise payable to an Affected Creditor or Unitholder such amounts (a "Withholding Obligation") as the Arctic Glacier Parties, the Monitor or the Transfer Agent, as the case may be, is required or entitled to deduct and withhold with respect to such payment under the *Income Tax Act* (Canada), the IRC, or any other provision of any Applicable Law. To the extent that amounts are so deducted or withheld and remitted to the applicable Government Authority or as required by Applicable Law, such amounts deducted or withheld shall be treated for all purposes of the Consolidated CCAA Plan as having been paid to such Person as the remainder of the payment in respect of which such withholding and deduction were made. For greater certainty, no distribution, payment or other consideration shall be made to or on behalf of a holder of a Proven Claim or a Unitholder pursuant to the Consolidated CCAA Plan unless and until such Person has made arrangements satisfactory to the Arctic Glacier Parties, the Monitor, or the Transfer Agent, as the case may be, for the payment and satisfaction of any Withholding Obligations imposed on the Arctic Glacier Parties, the Monitor or the Transfer Agent by any Government Authority.

ARTICLE 7 PROCEDURE FOR DISTRIBUTIONS REGARDING UNRESOLVED CLAIMS

7.1 No Distribution Pending Allowance

Notwithstanding any other provision of the Consolidated CCAA Plan, no payments or distributions shall be made with respect to all or any portion of an Unresolved Claim unless and to the extent it has become a Proven Claim, in whole or in part.

7.2 Unresolved Claims Reserve

On the Plan Implementation Date, the Monitor shall establish and maintain the Unresolved Claims Reserve from the Available Funds, in accordance with Section 5.4 of the Consolidated CCAA Plan.

7.3 Distributions After Unresolved Claims Resolved

The Unresolved Claims shall be finally determined in accordance with the Claims Procedure Order and the Claims Officer Order. If an Affected Creditor's Unresolved Claim is finally determined to be a Proven Claim pursuant to and in accordance with the Claims Procedure Order and the Claims Officer Order or if an Affected Creditor's Unresolved Claim is accepted, in each case, in whole or in part, (a) the Monitor, on behalf of the Arctic Glacier Parties, shall distribute the amount from the Unresolved Claims Reserve equal to such Affected

Creditor's Distribution Claim, if any, that would have been distributed on the Plan Implementation Date or on a Distribution Date, as the case may be, had such Affected Claim been a Proven Claim (the "Proven Claim Amount") to such Affected Creditor in full satisfaction, payment, settlement, release and discharge of such Affected Creditor's Distribution Claim; and (b) that Proven Claim Amount shall be deemed to have first been transferred to the Affected Creditors' Distribution Cash Pool and then paid therefrom by the Monitor, on behalf of the Arctic Glacier Parties. When all Unresolved Claims have been finally determined in accordance with the Claims Procedure Order and the Claims Officer Order and when all Proven Claim Amounts have been paid, any balance that remains in the Unresolved Claims Reserve will be deemed to be transferred to the Administrative Costs Reserve.

The Monitor shall have no obligation to make any payment out of the Unresolved Claims Reserve, and nothing in the Consolidated CCAA Plan, the Meeting Order or the Sanction Order shall be construed as obligating the Monitor to make any such payment if, in the Monitor's sole and unfettered discretion, the cost of making any such payment is prohibitive for so doing in relation to the quantum of the contemplated payment.

ARTICLE 8 COMPANY REORGANIZATION

8.1 Corporate Authorizations

The adoption, execution, delivery, implementation and consummation of all matters contemplated under the Consolidated CCAA Plan involving corporate action of the Arctic Glacier Parties will occur and be effective as of the Plan Implementation Date, and will be authorized and approved under the Consolidated CCAA Plan and by the CCAA Court, where appropriate, as part of the Sanction Order, in all respects and for all purposes without any requirement of further action by any shareholders, Unitholders, Directors, Officers or Trustees. All necessary approvals to take actions shall be deemed to have been obtained from the Directors, Trustees, Unitholders or shareholders of the Arctic Glacier Parties, as applicable, including the deemed passing by the Unitholders or shareholders of any resolution or special resolution and no shareholders' agreement or Unitholders' agreement or agreement between a shareholder or Unitholder (as applicable) and another Person limiting in any way the right to vote shares or Trust Units (as applicable) held by such shareholder(s) or Unitholder(s) (as applicable) with respect to any of the steps contemplated by the Consolidated CCAA Plan shall be deemed to be effective and shall have no force and effect.

8.2 Charges

The beneficiaries of the Charges shall provide the Monitor with evidence of all outstanding, invoiced obligations, liabilities, fees and disbursements secured by the Charges as of three (3) Business Days prior to the Plan Implementation Date, along with a reasonable estimate of the additional obligations, liabilities, fees and disbursements that are secured by the Charges and will be incurred up to the Plan Implementation Date (collectively, the "PID Charge Amount"). On the Plan Implementation Date, the PID Charge Amount shall be fully paid by the Monitor, on behalf of the Arctic Glacier Parties. Upon receipt by the Monitor of confirmation from each of the beneficiaries of the Charges that it has received the applicable portion of the PID Charge Amount that was paid by the Monitor, on behalf of the Arctic Glacier Parties, on the Plan Implementation Date, the Monitor shall file a certificate with the CCAA Court confirming

that all outstanding, invoiced obligations, liabilities, fees and disbursements secured by the Charges as of the Plan Implementation Date have been paid and thereafter, the Charges shall be and be deemed to be discharged from the assets of the Arctic Glacier Parties without the need for any other formality.

8.3 Plan Implementation Date Steps and Transactions

The steps, transactions, settlements and releases to be effected in the implementation of the Consolidated CCAA Plan shall occur, and be deemed to have occurred, in the following order without any further act of formality, beginning at the Effective Time on the Plan Implementation Date:

- (a) the Monitor, on behalf of the Arctic Glacier Parties, shall use the Available Funds to fund the following reserves and distribution cash pools in the order specified below:
 - (i) Administrative Costs Reserve;
 - (ii) Insurance Deductible Reserve;
 - (iii) Unresolved Claims Reserve;
 - (iv) Affected Creditors' Distribution Cash Pool; and
 - (v) Unitholders' Distribution Cash Pool; and

administer such reserves and distribution cash pools pursuant to and in accordance with the Consolidated CCAA Plan;

- (b) the Monitor, on behalf of the Arctic Glacier Parties, shall pay from the Administrative Costs Reserve the applicable portion of the PID Charge Amount, if any, to each of the beneficiaries of the Charges;
- (c) the Arctic Glacier Parties shall pay to the Monitor an amount of \$426,252.16 (including HST) in respect of the discounted component of fees earned by Alvarez & Marsal Canada Inc. during the period of November 21, 2011 to December 31, 2012 (the "Recovered Fees");
- (d) the steps, assumptions, distributions, transfers, payments, contributions, liquidations, dissolutions, wind-ups, reduction of capital, settlements and releases set out in Schedule "B" of the Consolidated CCAA Plan shall be deemed to be completed in the order specified therein; and
- (e) the releases referred to in Section 9 of the Consolidated CCAA Plan shall become effective in accordance with the Consolidated CCAA Plan.

8.4 Post-Plan Implementation Date Transactions

As specified herein, the Fund, Arctic Glacier Inc. and Arctic Glacier International Inc., or the Monitor on their behalf, as the case may be, shall take the following steps after the Plan Implementation Date:

- (a) the Monitor, on behalf of the Arctic Glacier Parties, shall take all steps necessary to pay any amounts required to be paid to an Affected Creditor or to the Unitholders after the Plan Implementation Date pursuant to, and in accordance with, this Consolidated CCAA Plan;
- (b) (i) the Monitor, on behalf of the Arctic Glacier Parties, shall take all steps necessary to make any distributions, payments, or transfers in order to fund, or otherwise in connection with, the making of the payments referred to in subparagraph (a) above; and (ii) the Fund, Arctic Glacier Inc. and Arctic Glacier International Inc., in consultation with the Monitor, shall take all steps necessary to undertake any other transactions as between the Fund, Arctic Glacier Inc. and Arctic Glacier International Inc. in order to fund, or otherwise take steps in connection with, the making of the payments referred to in subparagraph (a) above; and
- (c) (i) the Fund, Arctic Glacier Inc. and Arctic Glacier International Inc., in consultation with the Monitor, shall take all steps necessary to wind-up, liquidate, terminate and dissolve each of Arctic Glacier International Inc., Arctic Glacier Inc. and the Fund or undertake any other steps in connection therewith, including causing the Fund's units to cease to be listed and traded on the Canadian National Stock Exchange on (and for greater certainty, not prior to) the Final Distribution Date; and (ii) the Monitor, on behalf of the Arctic Glacier Parties, shall make any distributions, payments or transfers in connection therewith;

in each case, as tax efficiently for the Arctic Glacier Parties as is reasonably possible.

ARTICLE 9 RELEASES

9.1 Consolidated CCAA Plan Releases

On the Plan Implementation Date and in accordance with the sequential steps and transactions set out in Section 8.3 of the Consolidated CCAA Plan, the Arctic Glacier Parties, the Monitor, Alvarez and Marsal Canada Inc. and its affiliates, the CPS, the Trustees, the Directors and the Officers, each and every present and former employee who filed or could have filed an indemnity claim or a DO&T Indemnity Claim against the Arctic Glacier Parties, each and every affiliate, subsidiary, member (including members of any committee or governance council), auditor, financial advisor, legal counsel and agent thereof and any Person claiming to be liable derivatively through any or all of the foregoing Persons (the "Releasees") shall be released and discharged from any and all demands, claims, actions, causes of action, counterclaims, suits, debts, sums of money, accounts, covenants, damages, judgments, orders, including for injunctive relief or specific performance and compliance orders, expenses, executions and other recoveries on account of any liability, obligation, demand or cause of action of whatever nature which any

Person may be entitled to assert, including any and all claims in respect of the payment and receipt of proceeds and statutory liabilities of Trustees, Directors, Officers and employees of the Arctic Glacier Parties and any alleged fiduciary or other duty (whether acting as a Trustee, Director, Officer, member or employee or acting in any other capacity in connection with the Arctic Glacier Parties' business or an individual Arctic Glacier Party), whether known or unknown, matured or unmatured, direct, indirect or derivative, foreseen or unforeseen, existing or hereafter arising, based in whole or in part on any omission, transaction, duty, responsibility, indebtedness, liability, obligation, dealing or other occurrence existing or taking place on or prior to the later of the Plan Implementation Date and the date on which actions are taken to implement the Consolidated CCAA Plan that are in any way related to, or arising out of or in connection with the Claims, the Arctic Glacier Parties' business and affairs whenever or however conducted, the Consolidated CCAA Plan, the CCAA Proceedings, any Claim that has been barred or extinguished pursuant to the Claims Procedure Order or the Claims Officer Order (excepting only Releasees in respect of Unresolved Claims, unless and until such Unresolved Claims become Proven Claims in accordance with the Claims Procedure Order and the Claims Officer Order), and all claims arising out of such actions or omissions shall be forever waived and released (other than the right to enforce the Arctic Glacier Parties' obligations under the Consolidated CCAA Plan or any related document), all to the full extent permitted by applicable law, provided that nothing in the Consolidated CCAA Plan shall release or discharge a Releasee from any obligation created by or existing under the Consolidated CCAA Plan or any related document.

ARTICLE 10 COURT SANCTION, CONDITIONS PRECEDENT AND IMPLEMENTATION

10.1 Application for Sanction Order

If the Required Unitholder Majority approves the Consolidated CCAA Plan, the Applicants shall apply for the Sanction Order on or before the date set for the hearing of the Sanction Order or such later date as the CCAA Court may set.

10.2 Sanction Order

The Sanction Order shall, among other things, include provisions in substance similar to the following:

- (a) declare that each of the Creditors' Meeting and the Unitholders' Meeting shall have been duly called and held in accordance with the Meeting Order;
- (b) declare that (i) the Consolidated CCAA Plan has been unanimously approved by the Affected Creditors in conformity with the CCAA; (ii) the Consolidated CCAA Plan has been approved by the required majorities of Unitholders in conformity with the Meeting Order; (iii) the activities of the Arctic Glacier Parties have been in reasonable compliance with the provisions of the CCAA and the Orders of the CCAA Court made in the CCAA Proceeding in all respects; (iv) the CCAA Court is satisfied that the Arctic Glacier Parties have not done or purported to do anything that is not authorized by the CCAA; and (v) the Consolidated CCAA Plan and the transactions contemplated thereby are fair and reasonable;

- (c) declare that as of the Effective Time, the Consolidated CCAA Plan and all associated steps, settlements, transactions, arrangements and releases effected thereby are approved, binding and effective upon the Arctic Glacier Parties, all Affected Creditors, the Directors and Officers, the Unitholders, the Trustees, the Releasees and all other Persons named or referred to in, or subject to, the Consolidated CCAA Plan and their respective heirs, executors, administrators and other legal representatives, successors and assigns;
- (d) declare that the steps to be taken and the releases to be effective on the Plan Implementation Date are deemed to occur and be effected in the sequential order contemplated by the Consolidated CCAA Plan on the Plan Implementation Date, beginning at the Effective Time;
- (e) settle, discharge and release the Arctic Glacier Parties from any and all Affected Claims of any nature in accordance with the Consolidated CCAA Plan, and declare that the ability of any Person to proceed against the Arctic Glacier Parties in respect of or relating to any Affected Claims shall be forever discharged and restrained, and all proceedings with respect to, in connection with or relating to such Affected Claims are permanently stayed, subject only to (i) the right of Affected Creditors with Unresolved Claims to continue pursuing such Unresolved Claims in accordance with the Claims Procedure Order, the Claims Officer Order and the Consolidated CCAA Plan; and (ii) the right of Affected Creditors and Unitholders to receive payments and distributions pursuant to the Consolidated CCAA Plan;
- (f) stay the commencing, taking, applying for or issuing or continuing of any and all steps or proceedings, including without limitation, administrative hearings and orders, declarations or assessments, commenced, taken or proceeded with or that may be commenced, taken or proceeded with against any Releasee in respect of all Claims and any matter which is released pursuant to the Consolidated CCAA Plan;
- (g) declare the interest rates that will be used to calculate the amount of interest to be paid to Affected Creditors, if applicable;
- (h) extend the stay of proceedings under the Initial Order;
- (i) declare that on or following the Plan Implementation Date, the Monitor shall be and is authorized and directed to make payments out of the Administrative Costs Reserve, on behalf of the Arctic Glacier Parties, in respect of the payment of Administrative Reserve Costs by way of cheque (sent by prepaid ordinary mail to the Monitor's last known address for such recipient Persons) or by wire transfer (in accordance with wire transfer instructions, if provided by such recipient Persons to the Monitor at least three (3) Business Days prior to the payment date set by the Monitor);
- (j) declare that all payments and distributions by or at the direction of the Monitor, in each case on behalf of the Arctic Glacier Parties or the Fund, as applicable, under the Consolidated CCAA Plan are for the account of the Arctic Glacier

- Parties or the Fund, as applicable, and the fulfillment of their obligations under Consolidated CCAA Plan;
- (k) declare that none of the Monitor, the CPS, the Trustees and the Applicants shall incur any liability as a result of payments and distributions to the Unitholders, in each case on behalf of the Fund, once such distribution or payment has been made by the Monitor to, and confirmation of receipt has been received by the Monitor from, the Transfer Agent;
- (l) declare that the Monitor and the CPS shall not incur any liability under the Tax Statutes as a result of the completion of the steps or transactions contemplated by the Consolidated CCAA Plan, including in respect of its making any payments or distributions ordered or permitted under the Consolidated CCAA Plan or the Sanction Order and including any steps or transactions contemplated by Section 8.4 of this Consolidated CCAA Plan, and are released, remised and discharged from any claims against them under or pursuant to the Tax Statutes or otherwise at law, arising in respect of the completion of the steps or transactions contemplated by the Consolidated CCAA Plan, including in respect of its making any payments or distributions ordered or permitted under the Consolidated CCAA Plan or the Sanction Order and including any steps or transactions contemplated by Section 8.4 of this Consolidated CCAA Plan, and that any claims of such a nature are forever barred and extinguished;
- (m) subject to payment thereof, declare that each of the Charges shall be terminated, discharged and released upon the filing by the Monitor with the CCAA Court of the certificate contemplated by Section 8.2 of the Consolidated CCAA Plan;
- (n) declare that any Affected Claims for which a Proof of Claim has not been filed by the Claims Bar Date or the DO&T Indemnity Claims Bar Date, as applicable, shall be forever barred and extinguished;
- (o) authorize and direct the Monitor to, on and after the Plan Implementation Date, (i) complete the claims procedure established in the Claims Procedure Order and Claims Officer Order; and (ii) take such further steps and seek such amendments to the Claims Procedure Order, Claims Officer Order or additional orders of the CCAA Court as the Monitor considers necessary or appropriate in order to fully determine, resolve or deal with any Claims;
- (p) declare that, in addition to its prescribed rights under the CCAA and the powers granted by the CCAA Court, the powers granted to the Monitor are expanded as may be required to, and the Monitor is empowered and authorized on and after the Plan Implementation Date to, take such additional actions and execute such documents, in the name of and on behalf of the Arctic Glacier Parties, as the Monitor considers necessary or desirable in order to perform its functions and fulfill its obligations under the Consolidated CCAA Plan, the Sanction Order and any order of the CCAA Court in the CCAA Proceedings and to facilitate the implementation of the Consolidated CCAA Plan and the completion of the CCAA proceedings, including to: (i) administer and distribute the Available Funds; (ii) establish and hold the Administrative Costs Reserve, the Insurance

Deductible Reserve, the Unresolved Claims Reserve, the Affected Creditors' Distribution Cash Pool and the Unitholders' Distribution Cash Pool; (iii) resolve any Unresolved Claims; (iv) effect payments in respect of Proven Claims to the Affected Creditors and effect distributions to the Transfer Agent in respect of distributions to be made to Unitholders; (v) take such steps, if and as may be necessary, to address Excluded Claims in accordance with the Consolidated CCAA Plan, the Claims Procedure Order and the Claims Officer Order; and (vi) take such steps as are necessary to effect the post-Plan Implementation Date steps and transactions set out in Section 8.4 of the Consolidated CCAA Plan; and, in each case where the Monitor takes such actions or steps, it shall be exclusively authorized and empowered to do so, to the exclusion of all other Persons including the Arctic Glacier Parties, and without interference from any other Person;

- (q) authorize the Monitor, in the name of and on behalf of the Arctic Glacier Parties, to prepare and file the Arctic Glacier Parties' tax returns based solely upon information provided by the Arctic Glacier Parties and on the basis that the Monitor shall incur no liability or obligation to any Person with respect to such returns or related documentation;
- (r) declare that on and after the Plan Implementation Date, the Monitor shall be at liberty to engage such Persons as the Monitor deems necessary or advisable respecting the exercise of its powers and performance of its obligations under the Consolidated CCAA Plan, the Sanction Order or any other order of the CCAA Court and to facilitate the completion of the CCAA proceedings;
- (s) declare that upon completion by the Monitor of its duties in respect of the Arctic Glacier Parties pursuant to the CCAA and any orders in the CCAA Proceedings, including, without limitation, the Monitor's duties in respect of the claims process and distributions made by the Monitor in accordance with the Consolidated CCAA Plan, the Monitor may file with the CCAA Court a certificate of Consolidated CCAA Plan termination stating that all of its duties in respect of the Arctic Glacier Parties pursuant to the CCAA and the orders in the CCAA Proceedings have been completed and thereupon, Alvarez & Marsal Canada Inc. shall be deemed to be discharged from its duties as Monitor of the Arctic Glacier Parties and released of all claims relating to its activities as Monitor;
- (t) declare that the Arctic Glacier Parties, the CPS and the Monitor may apply to the CCAA Court for advice and direction in respect of any matters arising from or under the Consolidated CCAA Plan; and
- (u) such other relief which the Arctic Glacier Parties or the Monitor may request.

10.3 Conditions Precedent to Implementation of the Consolidated CCAA Plan

The implementation of the Consolidated CCAA Plan shall be conditional upon the fulfillment of the following conditions on or prior to the Plan Implementation Date, as the case may be:

(a) Consolidated CCAA Plan Approval

The Affected Creditor Class shall have been deemed to have unanimously voted in favour of the Consolidated CCAA Plan at the Creditors' Meeting and the Consolidated CCAA Plan shall be approved by the Required Unitholder Majority.

(b) Plan Sanction Order

The Sanction Order shall have been made and be in full force and effect, and all applicable appeal periods in respect thereof shall have expired and any appeals therefrom shall have been finally disposed of, leaving the Sanction Order wholly operable.

(c) Recognition Order

A Recognition Order in the Chapter 15 Proceedings shall have been made recognizing the Sanction Order and such order shall be in full force and effect, and all applicable appeal periods in respect thereof shall have expired and any appeals therefrom shall have been finally disposed of, leaving such Recognition Order wholly operable.

(d) Resolution of Certain Liabilities

CPS and the Monitor are satisfied that (a) all tax returns required to be filed by or on behalf of the Arctic Glacier Parties have or will be duly filed in all appropriate jurisdictions; and (b) all taxes required to be paid in respect thereof have or will be paid.

10.4 Monitor's Certificate

Upon CPS and the Monitor determining, based on inquiries and consultation with the Arctic Glacier Parties or otherwise, that the conditions to implementation of the Consolidated CCAA Plan set out in Section 10.3 have been satisfied or waived, the Monitor shall deliver to the Arctic Glacier Parties a certificate which states that all conditions precedent set out in Section 10.3 have been satisfied or waived and that the Plan Implementation Date has occurred. Following the Plan Implementation Date, the Monitor shall file such certificate with the CCAA Court.

ARTICLE 11 GENERAL

11.1 Binding Effect

On the Plan Implementation Date:

- (a) the Consolidated CCAA Plan will become effective at the Effective Time;
- (b) the treatment of Affected Claims under the Consolidated CCAA Plan shall be final and binding for all purposes and enure to the benefit of the Arctic Glacier

Parties, all Affected Creditors, the Directors and Officers, the Unitholders, the Trustees, the Releasees and all other Persons and parties named or referred to in, or subject to, the Consolidated CCAA Plan and their respective heirs, executors, administrators and other legal representatives, successors and assigns;

- (c) all Affected Claims shall be forever discharged and released, excepting only (i) the right of Affected Creditors with Unresolved Claims to continue pursuing such Unresolved Claims in accordance with the Claims Procedure Order, the Claims Officer Order and the Consolidated CCAA Plan; and (ii) the obligation of the Arctic Glacier Parties to make payments and distributions in respect of such Affected Claims in the manner and to the extent provided for in the Consolidated CCAA Plan;
- (d) each Affected Creditor will be deemed to have consented and agreed to all of the provisions of the Consolidated CCAA Plan, in its entirety;
- (e) each Unitholder will be deemed to have consented and agreed to all of the provisions of the Consolidated CCAA Plan, in its entirety; and
- (f) each Affected Creditor and Unitholder shall be deemed to have executed and delivered to the Arctic Glacier Parties all consents, releases, assignments and waivers, statutory or otherwise, required to implement and carry out the Consolidated CCAA Plan in its entirety.

11.2 Waiver of Defaults

From and after the Plan Implementation Date, all Persons shall be deemed to have waived any and all defaults of the Arctic Glacier Parties then existing or previously committed by the Arctic Glacier Parties, or caused by the Arctic Glacier Parties, any of the provisions in the Consolidated CCAA Plan or steps contemplated in the Consolidated CCAA Plan, or non-compliance with any covenant, warranty, representation, term, provision, condition or obligation, expressed or implied, in any contract, instrument, credit document, lease, guarantee, agreement for sale or other agreement, written or oral, and any and all amendments or supplements thereto, existing between such Person and the Arctic Glacier Parties and any and all notices of default and demands for payment or any step or proceeding taken or commenced in connection therewith under any such agreement shall be deemed to have been rescinded and of no further force or effect, provided that nothing shall be deemed to excuse the Arctic Glacier Parties from performing their obligations under the Consolidated CCAA Plan or be a waiver of defaults by the Arctic Glacier Parties under the Consolidated CCAA Plan and the related documents. This Section does not affect the rights of any Person to pursue any recoveries for an Affected Claim that may be obtained from a guarantor and any security granted by such guarantor.

11.3 Claims Bar Date

Nothing in the Consolidated CCAA Plan extends or shall be interpreted as extending or amending the Claims Bar Date or the DO&T Indemnity Claims Bar Date, as applicable, or gives or shall be interpreted as giving any rights to any Person in respect of Affected Claims that have been barred or extinguished pursuant to the Claims Procedure Order or the Claims Officer Order.

11.4 Deeming Provisions

In the Consolidated CCAA Plan, the deeming provisions are not rebuttable and are conclusive and irrevocable.

11.5 Non-Consummation

The Arctic Glacier Parties reserve the right to revoke or withdraw the Consolidated CCAA Plan at any time prior to the Plan Sanction Date. If the Arctic Glacier Parties revoke or withdraw the Consolidated CCAA Plan, if the Sanction Order is not issued, or if the Plan Implementation Date does not occur, (a) the Consolidated CCAA Plan shall be null and void in all respects, (b) any settlement or compromise embodied in the Consolidated CCAA Plan including the fixing or limiting to an amount certain any Claim, or any document or agreement executed pursuant to the Consolidated CCAA Plan shall be deemed null and void, and (c) nothing contained in the Consolidated CCAA Plan, and no acts taken in preparation for consummation of the Consolidated CCAA Plan, shall (i) constitute or be deemed to constitute a waiver or release of any Affected Claims by or against the Arctic Glacier Parties or any other Person; (ii) prejudice in any manner the rights of the Arctic Glacier Parties or any other Person in any further proceedings involving the Arctic Glacier Parties; or (iii) constitute an admission of any sort by the Arctic Glacier Parties or any other Person.

11.6 Modification of the Consolidated CCAA Plan

- (a) The Arctic Glacier Parties reserve the right, at any time and from time to time, to amend, restate, modify and/or supplement the Consolidated CCAA Plan, provided that any such amendment, restatement, modification or supplement must be contained in a written document which is filed with the CCAA Court and (i) if made prior to the Creditors' Meeting and/or the Unitholders' Meeting, communicated to the Affected Creditors and/or the Unitholders, as applicable, in the manner required by the CCAA Court (if so required); and (ii) if made following the Creditors' Meeting and/or the Unitholders' Meeting, approved by the CCAA Court following notice to the Affected Creditors and/or the Unitholders, as applicable.
- (b) Notwithstanding Section 11.6(a), any amendment, restatement, modification or supplement may be made by the Arctic Glacier Parties with the consent of the Monitor or pursuant to an Order following the Plan Sanction Date, provided that it concerns a matter which, in the opinion of the Arctic Glacier Parties, acting reasonably, is of an administrative nature required to better give effect to the implementation of the Consolidated CCAA Plan and the Sanction Order or to cure any errors, omissions or ambiguities and is not materially adverse to the financial or economic interests of the Affected Creditors or the Unitholders.
- (c) Any amended, restated, modified or supplementary plan or plans of compromise filed with the CCAA Court and, if required by this Section, approved by the CCAA Court, shall, for all purposes, be and be deemed to be a part of and incorporated in the Consolidated CCAA Plan.

(d) In the event that this Consolidated CCAA Plan is amended, the Monitor shall post such amended Consolidated CCAA Plan on the Monitor's Website and such posting shall constitute adequate notice of such amendment.

11.7 Paramountcy

From and after the Effective Time on the Plan Implementation Date, any conflict between:

- (a) the Consolidated CCAA Plan; and
- (b) the Meeting Order and the covenants, warranties, representations, terms, conditions, provisions or obligations, expressed or implied, of any contract, mortgage, security agreement, indenture, trust indenture, loan agreement, commitment letter, agreement for sale, articles or bylaws of the Arctic Glacier Parties, lease or other agreement, written or oral and any and all amendments or supplements thereto existing between one or more of the Affected Creditors or Unitholders, as the case may be, and the Arctic Glacier Parties as at the Plan Implementation Date;

will be deemed to be governed by the terms, conditions and provisions of the Consolidated CCAA Plan and the Sanction Order, which shall take precedence and priority.

11.8 Severability of Plan Provisions

If, prior to the Plan Sanction Date, any term or provision of the Consolidated CCAA Plan is held by the CCAA Court to be invalid, void or unenforceable, the CCAA Court, at the request of the Arctic Glacier Parties, shall have the power to either (a) sever such term or provision from the balance of the Consolidated CCAA Plan and provide the Arctic Glacier Parties with the option to proceed with the implementation of the balance of the Consolidated CCAA Plan as of and with effect from the Plan Implementation Date, or (b) alter and interpret such term or provision to make it valid or enforceable to the maximum extent practicable, consistent with the original purpose of the term or provision held to be invalid, void or unenforceable, and such term or provision shall then be applicable as altered or interpreted. Notwithstanding any such holding, alteration or interpretation, and provided that the Arctic Glacier Parties proceed with the implementation of the Consolidated CCAA Plan, the remainder of the terms and provisions of the Consolidated CCAA Plan shall remain in full force and effect and shall in no way be affected, impaired or invalidated by such holding, alteration or interpretation.

11.9 Reviewable Transactions

Section 36.1 of the CCAA, sections 38 and 95 to 101 of the BIA and any other federal or provincial law relating to preferences, fraudulent conveyances or transfers at undervalue, shall not apply to the Consolidated CCAA Plan or to any payments or distributions made in connection with transactions entered into by or on behalf of the Arctic Glacier Parties, whether before or after the Filing Date, including to any and all of the payments, distributions and transactions contemplated by and to be implemented pursuant to the Consolidated CCAA Plan.

11.10 Responsibilities of the Monitor

Alvarez & Marsal Canada Inc. is acting in its capacity as Monitor in the CCAA Proceedings with respect to the Arctic Glacier Parties and the Consolidated CCAA Plan and not in its personal or corporate capacity, and will not be responsible or liable for any obligations of the Arctic Glacier Parties under the Consolidated CCAA Plan or otherwise.

11.11 Different Capacities

Persons who are affected by the Consolidated CCAA Plan may be affected in more than one capacity. Unless expressly provided herein to the contrary, a Person will be entitled to participate hereunder in each such capacity. Any action taken by a Person in one capacity will not affect such Person in any other capacity, unless expressly agreed by the Person in writing or unless its Claims overlap or are otherwise duplicative.

11.12 Notices

Any notice or other communication to be delivered hereunder must be in writing and reference the Consolidated CCAA Plan and may, subject as hereinafter provided, be made or given by personal delivery, ordinary mail or by facsimile or email addressed to the respective parties as follows:

If to the Arctic Glacier Parties:

c/o CPS

39 Wynford Drive

Toronto ON M3C 3K5

Attention:

Bruce Robertson

Fax:

416-446-0050

Email:

bkrobertson@yahoo.com

with copies to:

Aikins, MacAulay & Thorvaldson LLP

30th Floor Commodity Exchange Tower

360 Main Street, Winnipeg, Manitoba R3C 4G1

Attention:

Hugh A. Adams and Dale R. Melanson

Fax:

204-957-4437

Email:

haa@aikins.com / drm@aikins.com

Kevin P. McElcheran Professional Corporation

120 Adelaide St. West

Suite 420, P.O. Box 43

Toronto, Ontario M5H 1T1

Attention:

Kevin P. McElcheran

Email:

kevin@mcelcheranadr.com

If to an Affected Creditor:

to the address or facsimile number or email address for such Creditor specified in the Proof of Claim filed by such Creditor;

If to the Monitor:

Alvarez & Marsal Canada Inc. 200 Bay Street, Suite 2900 Toronto, Ontario M5J 2J1

Attention:

Richard Morawetz/ Melanie MacKenzie

Fax:

416-847-5201

Email:

<u>rmorawetz@alvarezandmarsal.com/</u> mmackenzie@alvarezandmarsal.com

with a copy to:

Osler, Hoskin & Harcourt LLP

100 King Street West

1 First Canadian Place, Suite 6100, P.O. Box 50

Toronto, Ontario M5X 1B8

Attention:

Jeremy Dacks / Marc S. Wasserman / Michael De Lellis

Fax:

(416) 862-6666

Email:

jdacks@osler.com/mwasserman@osler.com/mdelellis@osler.com

or to such other address as any party may from time to time notify the others in accordance with this Section. Any such communication so given or made shall be deemed to have been given or made and to have been received on the day of delivery if delivered, or on the day of faxing or sending by other means of recorded electronic communication, provided that such day in either event is a Business Day and the communication is so delivered, faxed or sent before 5:00 p.m. CST or CDT, as the case may be, on such day. Otherwise, such communication shall be deemed to have been given and made and to have been received on the next following Business Day.

If, during any period during which notices or other communications are being given pursuant to this Consolidated CCAA Plan, a postal strike or postal work stoppage of general application should occur, such notices or other communications sent by ordinary mail and then not received shall not, absent further Order of the CCAA Court, be effective and notices and other communications given hereunder during the course of any such postal strike or work stoppage of general application shall only be effective if given by courier, personal delivery or electronic or digital transmission in accordance with this Order.

11.13 Further Assurances

Each of the Persons named or referred to in, or subject to, the Consolidated CCAA Plan will execute and deliver all such documents and instruments and do all such acts and things as may be necessary or desirable to carry out the full intent and meaning of the Consolidated CCAA Plan and to give effect to the transactions contemplated herein.

DATED as of the 26th day of August, 2014.

SCHEDULE "A" ADDITIONAL APPLICANTS

Arctic Glacier Lansing Inc. Arctic Glacier Michigan Inc. Arctic Glacier Minnesota Inc. Arctic Glacier Nebraska Inc. Arctic Glacier Newburgh Inc. Arctic Glacier New York Inc. Arctic Glacier Oregon Inc. **Arctic Glacier Party Time Inc.** Arctic Glacier Pennsylvania Inc. Arctic Glacier Rochester Inc. Arctic Glacier Services Inc. Arctic Glacier Texas Inc. Arctic Glacier Vernon Inc. Arctic Glacier Wisconsin Inc. Diamond Ice Cube Company Inc. **Diamond Newport Corporation** Glacier Ice Company, Inc. Ice Perfection Systems Inc. ICEsurance Inc. Jack Frost Ice Service, Inc. **Knowlton Enterprises, Inc. Mountain Water Ice Company** R&K Trucking, Inc. Winkler Lucas Ice and Fuel Company Wonderland Ice, Inc.

Arctic Glacier California Inc.

Arctic Glacier Grayling Inc.

SCHEDULE "B" SPECIFIED PLAN IMPLEMENTATION DATE STEPS

In order to effect the wind-up, liquidation and dissolution of certain of the Arctic Glacier Parties to facilitate the satisfaction of Proven Claims and a distribution by the Fund to Unitholders pursuant to and in accordance with the Consolidated CCAA Plan, the following steps, assumptions, distributions, transfers, payments, contributions, liquidations, dissolutions, wind-ups, reduction of capital, settlements and releases shall be deemed to occur (a) immediately after the completion of the step set out in Section 8.3(c) of the Consolidated CCAA Plan; (b) in the order specified in this Schedule "B"; and (c) in the manner specified in this Schedule "B".

Step 1: Assumption of Liabilities of Glacier Valley Ice Company, L.P.

All of the liabilities of Glacier Valley Ice Company, L.P. shall be assumed by, and become liabilities of, its limited partner, Arctic Glacier California Inc. and such assumption shall constitute a contribution of capital by Arctic Glacier California Inc. to Glacier Valley Ice Company, L.P. in an amount equal to the aggregate amount of such liabilities.

Step 2: Liquidation and dissolution of Glacier Valley Ice Company, L.P.

Glacier Valley Ice Company, L.P. is wound-up and dissolved and, upon such dissolution:

- (a) a 99.9% undivided interest in each of the assets of Glacier Valley Ice Company, L.P. shall be distributed to, and become property of, its limited partner, Arctic Glacier California Inc.; and
- (b) a 0.1% undivided interest in each of the assets of Glacier Valley Ice Company, L.P. shall be distributed to, and become property of, its general partner, Mountain Water Ice Company (California).

Step 3: Contribution of Intercompany Debts owing by Jack Frost Ice Service, Inc., Glacier Ice Company, Inc., Mountain Water Ice Company, Diamond Newport Corporation and Arctic Glacier Vernon Inc. (together, the "Step 3 Companies")

- (a) Arctic Glacier Inc. shall transfer any debt owing by a Step 3 Company to Arctic Glacier Inc. immediately prior to the completion of this Step 3(a) to Arctic Glacier International Inc. as a contribution to the capital stock of Arctic Glacier International Inc.
- (b) Arctic Glacier International Inc. shall transfer any debt owing by a Step 3 Company to Arctic Glacier International Inc. immediately prior to the completion of this Step 3(b) (including, for greater certainty, the intercompany debt contributed by Arctic Glacier Inc. to Arctic Glacier International Inc. pursuant to Step 3(a)) to Arctic Glacier California Inc. as a contribution to the capital stock of Arctic Glacier California Inc.
- (c) Arctic Glacier California Inc. shall transfer any debt owing by a Step 3 Company to Arctic Glacier California Inc. immediately prior to the completion of this Step 3(c) (including, for greater certainty, the intercompany debt contributed by Arctic

Glacier International Inc. to Arctic Glacier California Inc. pursuant to Step 3(b)) to the applicable Step 3 Company as a contribution to the capital stock of that Step 3 Company and, upon such contribution, such debt shall be cancelled.

Step 4: Assumption of Remaining Liabilities of the Step 3 Companies

All of the remaining liabilities of each Step 3 Company shall be assumed by, and become liabilities of Arctic Glacier California Inc. and such assumption shall constitute a contribution of capital by Arctic Glacier California Inc. to such Step 3 Company in an amount equal to the aggregate amount of such liabilities.

Step 5: Liquidation and dissolution of the Step 3 Companies

Each of Step 3 Companies shall be liquidated and dissolved into Arctic Glacier California Inc. and, on such liquidation and dissolution:

- (a) all of the assets of each of the Step 3 Companies shall be distributed to, and shall become property of, Arctic Glacier California Inc. and such assets shall be so received by Arctic Glacier California Inc. in respect of the shares of the capital stock of the Step 3 Companies; and
- (b) all of the shares of each of the Step 3 Companies shall be cancelled.

Arctic Glacier California Inc. and each of the Step 3 Companies intend that this Consolidated CCAA Plan shall constitute a plan of liquidation within the meaning of the U.S. treasury regulations promulgated under Section 332 of the IRC.

Step 6: Transfer of Shares of Winkler Lucas Ice and Fuel Company to Knowlton Enterprises Inc.

All of the shares of Winkler Lucas Ice and Fuel Company that are owned by Arctic Glacier Michigan Inc. (the "Transferred Shares") shall be transferred to Knowlton Enterprises Inc. and, in consideration therefore, Knowlton Enterprises Inc. shall be deemed to have issued to Arctic Glacier Michigan Inc., shares of its common stock with a fair market value equal to the fair market value of the Transferred Shares.

Step 7: Contribution of Intercompany Debts owing by Winkler Lucas Ice and Fuel Company

- (a) Arctic Glacier Inc. shall transfer any debt owing by Winkler Lucas Ice and Fuel Company to Arctic Glacier Inc. immediately prior to the completion of this Step 7(a) to Arctic Glacier International Inc. as a contribution to the capital stock of Arctic Glacier International Inc.
- (b) Arctic Glacier International Inc. shall transfer any debt owing by Winkler Lucas Ice and Fuel Company to Arctic Glacier International Inc. immediately prior to the completion of this Step 7(b) (including, for greater certainty, the intercompany debt contributed by Arctic Glacier Inc. to Arctic Glacier International Inc. pursuant to Step 7(a)) to Arctic Glacier Michigan Inc. as a contribution to the capital stock of Arctic Glacier Michigan Inc.

- (c) Arctic Glacier Michigan Inc. shall transfer any debt owing by Winkler Lucas Ice and Fuel Company to Arctic Glacier Michigan Inc. immediately prior to the completion of this Step 7(c) (including, for greater certainty, the intercompany debt contributed by Arctic Glacier International Inc. to Arctic Glacier Michigan Inc. pursuant to Step 7(b)) to Knowlton Enterprises Inc. as a contribution to the capital stock of Knowlton Enterprises Inc.
- (d) Knowlton Enterprises Inc. shall transfer any debt owing by Winkler Lucas Ice and Fuel Company to Knowlton Enterprises Inc. immediately prior to the completion of this Step 7(d) (including, for greater certainty, the intercompany debt contributed by Arctic Glacier Michigan Inc. to Knowlton Enterprises Inc. pursuant to Step 7(c)) to Winkler Lucas Ice and Fuel Company as a contribution to the capital stock of Winkler Lucas Ice and Fuel Company, and, upon such contribution, such debt shall be cancelled.

Step 8: Assumption of Remaining Liabilities of Winkler Lucas Ice and Fuel Company

All of the remaining liabilities of Winkler Lucas Ice and Fuel Company shall be assumed by, and become liabilities of Knowlton Enterprises Inc. and such assumption shall constitute a contribution of capital by Knowlton Enterprises Inc. to Winkler Lucas Ice and Fuel Company in an amount equal to the aggregate amount of such liabilities.

Step 9: Liquidation and dissolution of Winkler Lucas Ice and Fuel Company

Winkler Lucas Ice and Fuel Company shall be liquidated and dissolved into Knowlton Enterprises Inc. and, on such liquidation and dissolution:

- (a) all of the assets of Winkler Lucas Ice and Fuel Company shall be distributed to, and shall become property of, Knowlton Enterprises Inc. and such assets shall be so received by Knowlton Enterprises Inc. in respect of the shares of the capital stock of Winkler Lucas Ice and Fuel Company; and
- (b) all of the shares of Winkler Lucas Ice and Fuel Company shall be cancelled.

Knowlton Enterprises Inc. and Winkler Lucas Ice and Fuel Company intend that this Consolidated CCAA Plan shall constitute a plan of liquidation within the meaning of the U.S. treasury regulations promulgated under Section 332 of the IRC.

Step 10: Contribution of Intercompany Debts owing by Arctic Glacier Lansing Inc., Arctic Glacier Grayling Inc, Arctic Glacier Party Time Inc., Wonderland Ice, Inc., R&K Trucking, Inc. and Knowlton Enterprises, Inc. (together, the "Step 10 Companies").

- (a) Arctic Glacier Inc. shall transfer any debt owing by a Step 10 Company to Arctic Glacier Inc. immediately prior to the completion of this Step 10(a) to Arctic Glacier International Inc. as a contribution to the capital stock of Arctic Glacier International Inc.
- (b) Arctic Glacier International Inc. shall transfer any debt owing by a Step 10 Company to Arctic Glacier International Inc. immediately prior to the completion

of this Step 10(b) (including, for greater certainty, the intercompany debt contributed by Arctic Glacier Inc. to Arctic Glacier International Inc. pursuant to Step 10(a)) to Arctic Glacier Michigan Inc. as a contribution to the capital stock of Arctic Glacier Michigan Inc.

(c) Arctic Glacier Michigan Inc. shall transfer any debt owing by a Step 10 Company to Arctic Glacier Michigan Inc. immediately prior to the completion of this Step 10(c) (including, for greater certainty, the intercompany debt contributed by Arctic Glacier International Inc. to Arctic Glacier Michigan Inc. pursuant to Step 10(b)) to the applicable Step 10 Company as a contribution to the capital stock of that Step 10 Company and, upon such contribution, such debt shall be cancelled.

Step 11: Assumption of Remaining Liabilities of the Step 10 Companies

All of the remaining liabilities of each Step 10 Company shall be assumed by, and become liabilities of Arctic Glacier Michigan Inc. and such assumption shall constitute a contribution of capital by Arctic Glacier Michigan Inc. to such Step 10 Company in an amount equal to the aggregate amount of such liabilities.

Step 12: Liquidation and dissolution of the Step 10 Companies.

Each of Step 10 Companies shall be liquidated and dissolved into Arctic Glacier Michigan Inc. and, on such liquidation and dissolution:

- (a) all of the assets of each of the Step 10 Companies shall be distributed to, and shall become property of, Arctic Glacier Michigan Inc. and such assets shall be so received by Arctic Glacier Michigan Inc. in respect of shares of the capital stock of the Step 10 Companies; and
- (b) all of the shares of each of the Step 10 companies shall be cancelled.

Arctic Glacier Michigan Inc. and each of the Step 10 Companies intend that this Consolidated CCAA Plan shall constitute a plan of liquidation within the meaning of the U.S. treasury regulations promulgated under Section 332 of the IRC.

Step 13: Contribution of Intercompany Debts owing by Arctic Glacier Rochester Inc. and Diamond Ice Cube Company Inc. (the "Step 13 Companies").

- (a) Arctic Glacier Inc. shall transfer any debt owing by a Step 13 Company to Arctic Glacier Inc. immediately prior to the completion of this Step 13(a) to Arctic Glacier International Inc. as a contribution to the capital stock of Arctic Glacier International Inc.
- (b) Arctic Glacier International Inc. shall transfer any debt owing by a Step 13 Company to Arctic Glacier International Inc. immediately prior to the completion of this Step 13(b) (including, for greater certainty, the intercompany debt contributed by Arctic Glacier Inc. to Arctic Glacier International Inc. pursuant to Step 13(a)) to Arctic Glacier New York Inc. as a contribution to the capital stock of Arctic Glacier New York Inc.

(c) Arctic Glacier New York Inc. shall transfer any debt owing by a Step 13 Company to Arctic Glacier New York Inc. immediately prior to the completion of this Step 13(c) (including, for greater certainty, the intercompany debt contributed by Arctic Glacier International Inc. to Arctic Glacier New York Inc. pursuant to Step 13(b)) to the applicable Step 13 Company as a contribution to the capital stock of that Step 13 Company and, upon such contribution, such debt shall be cancelled.

Step 14: Assumption of Remaining Liabilities of the Step 13 Companies

All of the remaining liabilities of each Step 13 Company shall be assumed by, and become liabilities of Arctic Glacier New York Inc. and such assumption shall constitute a contribution of capital by Arctic Glacier New York Inc. to such Step 13 Company in an amount equal to the aggregate amount of such liabilities.

Step 15: Liquidation and dissolution of the Step 13 Companies

Each of Step 13 Companies shall be liquidated and dissolved into Arctic Glacier New York Inc. and, on such liquidation and dissolution:

- (a) all of the assets of each of the Step 13 Companies shall be distributed to, and shall become property of, Arctic Glacier New York Inc. and such assets shall be so received by Arctic Glacier New York Inc. in respect of shares of the capital stock of the Step 13 Companies; and
- (b) all of the shares of each of the Step 13 companies shall be cancelled.

Arctic Glacier New York Inc. and each of the Step 13 Companies intend that this Consolidated CCAA Plan shall constitute a plan of liquidation within the meaning of the U.S. treasury regulations promulgated under Section 332 of the IRC.

Step 16: Satisfaction of the CEPA Claim

The CEPA Claim shall be deemed to have been fully paid and satisfied by Arctic Glacier California Inc., released and discharged, and such portion of the Affected Creditors' Distribution Cash Pool as is equal to the Distribution Claim in respect of the CEPA Claim shall be held by the Monitor on behalf of the California Environmental Protection Agency — Department of Toxic Substance Control and distributed by the Monitor in accordance with Section 6.1 of the Consolidated CCAA Plan.

Step 17: Contribution of Intercompany Debts owing by Arctic Glacier Texas Inc., Arctic Glacier California Inc., Arctic Glacier Michigan Inc., Arctic Glacier Nebraska Inc., Arctic Glacier Wisconsin Inc., Arctic Glacier Minnesota Inc., Arctic Glacier New York Inc., Ice Perfection Systems Inc., Arctic Glacier Newburgh Inc., Arctic Glacier Pennsylvania Inc., Arctic Glacier Oregon Inc., Arctic Glacier Services Inc., and ICEsurance Inc. (together, the "Step 17 Companies")

(a) Arctic Glacier Inc. shall transfer any debt owing by a Step 17 Company to Arctic Glacier Inc. immediately prior to the completion of this Step 17(a) to Arctic

Glacier International Inc. as a contribution to the capital stock of Arctic Glacier International Inc.

(b) Arctic Glacier International Inc. shall transfer any debt owing by a Step 17 Company to Arctic Glacier International Inc. immediately prior to the completion of this Step 17(b) (including, for greater certainty, the intercompany debt contributed by Arctic Glacier Inc. to Arctic Glacier International Inc. pursuant to Step 17(a)) to the applicable Step 17 Company as a contribution to the capital stock of that Step 17 Company and, upon such contribution, such debt shall be cancelled.

Step 18: Assumption of Remaining Liabilities of the Step 17 Companies

All of the remaining liabilities of each Step 17 Company shall be assumed by, and become liabilities of Arctic Glacier International Inc. and such assumption shall constitute a contribution of capital by Arctic Glacier International Inc. to such Step 17 Company in an amount equal to the aggregate amount of such liabilities.

Step 19: Liquidation and dissolution of the Step 17 Companies

Each of Step 17 Companies shall be liquidated and dissolved into Arctic Glacier International Inc. and, on such liquidation and dissolution:

- (a) all of the assets of each of the Step 17 Companies shall be distributed to, and shall become property of, Arctic Glacier International Inc. and such assets shall be so received by Arctic Glacier International Inc. in respect of shares of the capital stock of the Step 17 Companies; and
- (b) all of the shares of each of the Step 17 Companies shall be cancelled.

Arctic Glacier International Inc. and each of the Step 17 Companies intend that this Consolidated CCAA Plan shall constitute a plan of liquidation within the meaning of the U.S. treasury regulations promulgated under Section 332 of the IRC.

Step 20: Satisfaction of the Proven Claims against Arctic Glacier International Inc.

- (a) The DOJ Claim shall be deemed to have been fully paid and satisfied by Arctic Glacier International Inc., released and discharged and such portion of the Affected Creditors' Distribution Cash Pool as is equal to the DOJ Claim shall be held by the Monitor on behalf of the US Department of Justice and distributed by the Monitor in accordance with Section 6.1 of the Consolidated CCAA Plan; and
- (b) The portion of the Proven Claim of Macquarie Bank Limited that is denominated in US dollars shall be deemed to have been fully paid and satisfied by Arctic Glacier International Inc., released and discharged and such portion of the Affected Creditors' Distribution Cash Pool as is equal to the Distribution Claim in respect of that portion of the Proven Claim shall be held by the Monitor on behalf of Macquarie Bank Limited and distributed by the Monitor in accordance with Section 6.1 of the Consolidated CCAA Plan.

Step 21: Contribution of Intercompany Debts owing by Arctic Glacier International Inc.

Arctic Glacier Income Fund shall transfer any debt owing by Arctic Glacier International Inc. to Arctic Glacier Income Fund immediately prior to the completion of this Step 21 to Arctic Glacier Inc. as a contribution to the capital of Arctic Glacier Inc.

Step 22: Set Off of intercompany debts between Arctic Glacier International Inc. and Arctic Glacier Inc.

All or such portion of the aggregate of any amounts owing by Arctic Glacier Inc. to Arctic Glacier International Inc. immediately prior to the completion of this Step 22 (the "AGI-AGII Payables") as is equal to the lesser of:

- (i) the amount of the AGI-AGII Payables, and
- (ii) the aggregate of any amounts owing by Arctic Glacier International Inc. to Arctic Glacier Inc. immediately prior to the completion of this Step 22 (including, for greater certainty, the amount of intercompany debt contributed by Arctic Glacier Income Fund to Arctic Glacier Inc. pursuant to Step 21) (the "AGII-AGI Payables")

shall be fully and absolutely paid and satisfied by way of set off against all or such portion of the AGII-AGI Payables as is equal to the lesser of:

- (i) the amount of the AGII-AGI Payables, and
- (ii) the amount of the AGI-AGII Payables,

and, upon such set off, the portion of the AGI-AGII Payables and the portion of the AGII-AGI Payables that has been set off pursuant to the foregoing shall be deemed to have been absolutely paid and satisfied as a result of such set off.

Step 23: Repayment of any remaining AGII-AGI Payables

Arctic Glacier International Inc. shall be deemed to have paid to Arctic Glacier Inc. an amount equal to the least of:

- (i) the aggregate amount of the AGII-AGI Payables, if any, that remains outstanding following the set off described in Step 22,
- (ii) the AGII-AGI Total Distribution Amount, and
- (iii) the Available Funds held by the Monitor on behalf of AGII immediately prior to the completion of this Step 23,

from the Available Funds held by the Monitor on behalf of Arctic Glacier International Inc. immediately prior to the completion of this Step 23 on account of the amount owing by Arctic Glacier International Inc. to Arctic Glacier Inc. under the AGII-AGI Payables and such amount shall be held by the Monitor on behalf of Arctic Glacier Inc.

Step 24: Distribution by Arctic Glacier International Inc.

Arctic Glacier International Inc. shall be deemed to have paid a distribution to Arctic Glacier Inc. on its shares of common stock in an amount equal to difference, if any, between the AGII-AGI Total Distribution Amount and the amount paid by Arctic Glacier International Inc. on Step 23 and such amount shall be held by the Monitor on behalf of Arctic Glacier Inc.

Step 25: Satisfaction of the Proven Claims against Arctic Glacier Inc.

- (a) The Proven Claim of Brisson, Rosemary shall be deemed to have been fully paid and satisfied by Arctic Glacier Inc., released and discharged and such portion of the Affected Creditors' Distribution Cash Pool as is equal to the Distribution Claim in respect of such Proven Claim shall be held by the Monitor on behalf of Brisson, Rosemary and distributed by the Monitor in accordance with Section 6.1 of the Consolidated CCAA Plan.
- (b) The Proven Claim of Fontaine, Mark shall be deemed to have been fully paid and satisfied by Arctic Glacier Inc., released and discharged and such portion of the Affected Creditors' Distribution Cash Pool as is equal to the Distribution Claim in respect of such Proven Claim shall be held by the Monitor on behalf of Fontaine, Mark and distributed by the Monitor in accordance with Section 6.1 of the Consolidated CCAA Plan.
- (c) The Proven Claim of Waddell, Garth shall be deemed to have been fully paid and satisfied by Arctic Glacier Inc., released and discharged and such portion of the Affected Creditors' Distribution Cash Pool as is equal to the Distribution Claim in respect of such Proven Claim shall be held by the Monitor on behalf of Waddell, Garth and distributed by the Monitor in accordance with Section 6.1 of the Consolidated CCAA Plan.
- (d) The Proven Claim of Winther, Neil shall be deemed to have been fully paid satisfied by Arctic Glacier Inc., released and discharged and such portion of the Affected Creditors' Distribution Cash Pool as is equal to the Distribution Claim in respect of such Proven Claim shall be held by the Monitor on behalf of Winther, Neil and distributed by the Monitor in accordance with Section 6.1 of the Consolidated CCAA Plan.
- (e) The Proven Claim of Wohlgemuth, Michael shall be deemed to have been fully paid and satisfied by Arctic Glacier Inc., released and discharged and such portion of the Affected Creditors' Distribution Cash Pool as is equal to the Distribution Claim in respect of such Proven Claim shall be held by the Monitor on behalf of Wohlgemuth, Michael and distributed by the Monitor in accordance with Section 6.1 of the Consolidated CCAA Plan.
- (f) The Proven Claim of Bailey, Doug shall be deemed to have been fully paid and satisfied by Arctic Glacier Inc., released and discharged and such portion of the Affected Creditors' Distribution Cash Pool as is equal to the Distribution Claim in respect of such Proven Claim shall be held by the Monitor on behalf of Bailey,

- Doug and distributed by the Monitor in accordance with Section 6.1 of the Consolidated CCAA Plan.
- (g) The Proven Claim of Burrows, Keith shall be deemed to have been fully paid and satisfied by Arctic Glacier Inc., released and discharged and such portion of the Affected Creditors' Distribution Cash Pool as is equal to the Distribution Claim in respect of such Proven Claim shall be held by the Monitor on behalf of Burrows, Keith and distributed by the Monitor in accordance with Section 6.1 of the Consolidated CCAA Plan.
- (h) The Proven Claim of McMahon, Keith shall be deemed to have been fully paid and satisfied by Arctic Glacier Inc., released and discharged and such portion of the Affected Creditors' Distribution Cash Pool as is equal to the Distribution Claim in respect of such Proven Claim shall be held by the Monitor on behalf of McMahon, Keith and distributed by the Monitor in accordance with Section 6.1 of the Consolidated CCAA Plan.
- (i) The Proven Claim of Knowles, Louise shall be deemed to have been fully paid and satisfied by Arctic Glacier Inc., released and discharged and such portion of the Affected Creditors' Distribution Cash Pool as is equal to the Distribution Claim in respect of such Proven Claim shall be held by the Monitor on behalf of Knowles, Louise and distributed by the Monitor in accordance with Section 6.1 of the Consolidated CCAA Plan.
- (j) The Proven Claim of Corbin, Keith and Shirley shall be deemed to have been fully paid and satisfied by Arctic Glacier Inc., released and discharged and such portion of the Affected Creditors' Distribution Cash Pool as is equal to the Distribution Claim in respect of such Proven Claim shall be held by the Monitor on behalf of Corbin, Keith and Shirley and distributed by the Monitor in accordance with Section 6.1 of the Consolidated CCAA Plan.
- (k) The portion of the Proven Claim of Macquarie Bank Limited that is denominated in Canadian dollars shall be deemed to have been fully paid and satisfied by Arctic Glacier Inc., released and discharged and such portion of the Affected Creditors' Distribution Cash Pool as is equal to the Distribution Claim in respect of such Proven Claim shall be held by the Monitor on behalf of Macquarie Bank Limited and distributed by the Monitor in accordance with Section 6.1 of the draft Consolidated CCAA Plan.

If all, or any portion of, such Proven Claims were liabilities of Arctic Glacier International Inc. (including liabilities assumed by Arctic Glacier International Inc. pursuant to this Consolidated CCAA Plan), the satisfaction of such, or the applicable portion of such, Proven Claims by Arctic Glacier Inc. shall be deemed to be a contribution by Arctic Glacier Inc. to the capital of Arctic Glacier International Inc. in an amount equal to the aggregate amount of such liabilities of Arctic Glacier International Inc.

Step 26: Set Off of intercompany debts between Arctic Glacier Inc. and Arctic Glacier Income Fund.

All or such portion of the aggregate of any amounts owing by Arctic Glacier Income Fund to Arctic Glacier Inc. immediately prior to the completion of this Step 26 (the "AGIF-AGI Payables") as is equal to the lesser of:

- (i) the amount of the AGIF-AGI Payables, and
- (ii) the aggregate of any amounts owing by Arctic Glacier Inc. to Arctic Glacier Income Fund immediately prior to the completion of this Step 26 (the "AGI-AGIF Payables")

shall be fully and absolutely paid and satisfied by way of set off against all or such portion of the AGI-AGIF Payables as is equal to the lesser of:

- (i) the amount of the AGIF-AGI Payables, and
- (ii) the amount of the AGI-AGIF Payables,

and, upon such set off, the portion of the AGIF-AGI Payables and the portion of the AGI-AGIF Payables that has been set off pursuant to the foregoing shall be deemed to have been absolutely paid and satisfied as a result of such set off.

Step 27: Repayment of any remaining AGI-AGIF Payables

Arctic Glacier Inc. shall be deemed to have paid to Arctic Glacier Income Fund an amount equal to the least of:

- (i) the aggregate amount of the AGI-AGIF Payables, if any, that remains outstanding following the set off described in Step 26,
- (ii) the AGI-AGIF Total Distribution Amount, and
- (iii) the Available Funds held by the Monitor on behalf of AGI immediately prior to the completion of this Step 27,

from the Available Funds held by the Monitor on behalf of Arctic Glacier Inc. immediately prior to the completion of this Step 27 on account of the amount owing by Arctic Glacier Inc. to Arctic Glacier Income Fund under the AGI-AGIF Payables and such amount shall be held by the Monitor on behalf of Arctic Glacier Income Fund.

Step 28: Return of Capital by Arctic Glacier Inc.

The stated capital of Arctic Glacier Inc. shall be reduced by an amount (the "Return of Capital Amount") equal to the AGI-AGIF Total Distribution Amount less the amount of cash paid by AGI to AGIF on Step 27, by deducting that amount from the stated capital account maintained by Arctic Glacier Inc. for its common shares, and Arctic Glacier Inc. shall be deemed to have made a distribution of the Return of Capital Amount on the reduction of stated capital to Arctic Glacier Income Fund. The amount of cash in the Affected Creditors' Distribution Cash Pool and

the Unitholders' Distribution Cash Pool equal to the Return of Capital Amount shall be held by the Monitor on behalf of Arctic Glacier Income Fund.

Step 29: Satisfaction of the Proven Claims against Arctic Glacier Income Fund and the Arctic Glacier Parties

All the Proven Claims against Arctic Glacier Income Fund and the Arctic Glacier Parties outstanding following the completion of Step 1 through 28, including for greater certainty, the Direct Purchaser Claim, shall be deemed to have been fully paid and satisfied, released and discharged and the remainder of the Affected Creditors' Distribution Cash Pool as is equal to the amount of the Distribution Claims in respect of such Proven Claims shall be held by the Monitor on behalf of the applicable creditors in respect of those Proven Claims and distributed by the Monitor in accordance with Section 6.1 of the Consolidated CCAA Plan.

If all, or any portion of, such Proven Claims were liabilities of Arctic Glacier Inc. and/or Arctic Glacier International Inc. (including, for greater certainty, any liabilities assumed by Arctic Glacier International Inc. on Step 18), the satisfaction of such, or the applicable portion of such, Proven Claims by Arctic Glacier Income Fund shall be deemed to be a contribution by Arctic Glacier Income Fund to the capital of Arctic Glacier Inc. and (where applicable) from Arctic Glacier Inc. to Arctic Glacier International Inc. in amounts equal to the aggregate amount of such liabilities of Arctic Glacier Inc. and Arctic Glacier International Inc. respectively.

Step 30: Distribution by Arctic Glacier Income Fund.

Arctic Glacier Income Fund shall be deemed to have paid a distribution to each Unitholder in the amount of their Pro Rata Share of the Unitholders' Distribution Cash Pool immediately following the completion of Steps 1 through 29 above and such amount shall be transferred by the Monitor to the Transfer Agent and distributed by the Transfer Agent to the Unitholders in accordance with Section 6.2 of the Consolidated CCAA Plan.

EXHIBIT C

Transition Order

THE QUEEN'S BENCH Winnipeg Centre

IN THE MATTER OF THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c. C-36, AS AMENDED

AND IN THE MATTER OF A PROPOSED PLAN OF COMPROMISE OR ARRANGEMENT WITH RESPECT TO ARCTIC GLACIER INCOME FUND, ARCTIC GLACIER INC. AND ARCTIC GLACIER INTERNATIONAL INC. and the ADDITIONAL APPLICANTS LISTED ON SCHEDULE "A" HERETO"

(collectively, the "Applicants")

APPLICATION UNDER THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c. C-36, AS AMENDED

<u>CERTIFIED COPY</u>

TRANSITION ORDER

DATE OF HEARING: THURSDAY JULY 12, 2012 AT 10 A.M. BEFORE THE HONOURABLE MADAM JUSTICE SPIVAK

McCARTHY TÉTRAULT LLP

Suite 5300, Box 48
Toronto Dominion Bank Tower
Toronto-Dominion Centre
Toronto, ON M5K 1E6

Kevin McElcheran

Tel: (416) 601-7730 Fax: (416) 868-0673 Law Society No. 22119H

Heather L. Meredith

Tel: (416) 601-8342 Fax: (416) 868-0673 Law Society No. 48354R

File No. 10671373

AIKINS, MacAULAY & THORVALDSON LLP

30th Floor – 360 Main Street Winnipeg, MB R3C 4G1

G. Bruce Taylor

Tel: (204) 957-4669 Fax: (204) 957-4218

J.J. Burnell

Tel: (204) 957-4663 Fax: (204) 957-4285

File No.: 1103500

THE QUEEN'S BENCH Winnipeg Centre

THE HONOURABLE MADAM)	THURSDAY, THE 12th
)	
JUSTICE SPIVAK)	DAY OF JULY, 2012

IN THE MATTER OF THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c. C-36, AS AMENDED

AND IN THE MATTER OF A PROPOSED PLAN OF COMPROMISE OR ARRANGEMENT WITH RESPECT TO ARCTIC GLACIER INCOME FUND, ARCTIC GLACIER INC. AND ARCTIC GLACIER INTERNATIONAL INC. and the ADDITIONAL APPLICANTS LISTED ON SCHEDULE "A" HERETO"

(collectively, the "Applicants")

APPLICATION UNDER THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C., c. C-36, AS AMENDED CERTIFIED COPY

TRANSITION ORDER

THIS MOTION, made by the Applicants for an order, among other things, approving the Transition Services Agreement among Arctic Glacier, LLC (formerly known as H.I.G. Zamboni, LLC) (the "Original Purchaser"), the Applicants and the Monitor, made as of July 12, 2012 and dealing with certain transition matters in respect of the Applicants, was heard this day at the Law Courts Building at 408 York Avenue, in The City of Winnipeg, in the Province of Manitoba.

ON READING the Affidavit of Keith McMahon sworn July 10, 2012 (the "July 10 Affidavit"), and the Fifth Report of Alvarez & Marsal Canada Inc. (the "Monitor") dated July 10, 2012 (the "Fifth Report"), and on hearing the submissions of counsel for the Monitor and counsel for the Applicants; counsel for the Purchaser, the Arctic Lenders, the US Direct Purchaser Antitrust Settlement Class and the Trustees of Arctic Glacier Income Fund also appearing, a representative of Talamod Master Fund L.P. also present by telephone, no one

appearing for any other person on the Service List, although properly served as appears from the affidavit of Corrine Smorhay sworn July 12, 2012, filed:

1. THIS COURT ORDERS that all capitalized terms used herein and not otherwise defined shall have the meaning ascribed thereto in the Transition Services Agreement (the "Transition Services Agreement") attached and marked as Exhibit* to the July 10 Affidavit.

SERVICE

2. THIS COURT ORDERS that the time for service of the Notice of Motion, the July 10 Affidavit and the Fifth Report is hereby abridged and validated so that this Motion is properly returnable today and hereby dispenses with further service thereof.

APPROVAL OF TRANSITION SERVICES AGREEMENT

3. THIS COURT ORDERS AND DECLARES that the Transition Services Agreement is hereby approved, and the execution of the Transition Services Agreement by the Applicants and the Monitor is hereby authorized and approved, with such minor amendments as the Applicants, the Purchaser and the Monitor may deem necessary.

TRANSITION POWERS OF THE MONITOR

- 4. THIS COURT ORDERS that on and after the closing of the transactions contemplated by the Purchase Agreement (the "Closing"), the Monitor is authorized, but not required, in the name of and on behalf of the Applicants, to prepare and file various returns, remittances, statements, records or other documentation on behalf of Applicants, including but not limited to, tax returns, employee-related remittances, T4 statements, W2 and W3 forms and records of employment for the Applicants' former employees based solely upon information provided by the Applicants and on the basis that the Monitor shall incur no liability or obligation to any person or entity with respect to such returns, remittances, statements, records or other documentation.
- 5. THIS COURT ORDERS that on and after the Closing, the Monitor shall be at liberty to engage such persons or entities as the Monitor deems necessary or advisable respecting the exercise of its powers and performances under this Order and any other Order of this Honourable Court and to assist in facilitating the administration of these proceedings.

- 6. THIS COURT ORDERS that in addition to its prescribed rights in the CCAA and the powers granted by Orders of this Honourable Court, the Monitor is empowered and authorized, on and after the Closing, to take such additional actions and execute such documents, in the name of and on behalf of the Applicants, as the Monitor considers necessary or desirable in order to perform its functions and fulfill its obligations under this Order, any other Order of this Honourable Court or in connection with the Transition Services Agreement, or to assist in facilitating the administration of these proceedings.
- 7. THIS COURT ORDERS that, without limiting the provisions of the Initial Order, on and after the Closing, the Applicants shall remain in possession and control of the Property (as defined in the Initial Order), if any, which remains following the Closing and the Monitor shall not be deemed to be in possession and/or control of any such remaining Property.
- 8. THIS COURT ORDERS AND DECLARES that nothing in this Order shall constitute or be deemed to constitute the Monitor as a trustee, receiver, assignee, liquidator, administrator, receiver-manager, agent of the creditors or legal representative of any of the Applicants within the meaning of any relevant legislation.
- 9. THIS COURT ORDERS AND DECLARES that in addition to the rights and protections afforded the Monitor under the CCAA, any plan of arrangement and any Order of this Honourable Court, the Monitor shall not be liable for any act or omission on the part of the Monitor, or any reliance thereon, including without limitation, with respect to any information disclosed, any act or omission pertaining to the discharge of duties or obligations under any Order of this Honourable Court, in connection with the Transition Services Agreement or as otherwise requested by the Applicants, save and except for any claim or liability arising out of any gross negligence or wilful misconduct on the part of the Monitor. Nothing in this Order shall derogate from the protections afforded to the Monitor by the CCAA, any other applicable legislation, the Initial Order or any other Order of this Honourable Court.

TRANSITION POWERS OF THE CHIEF PROCESS SUPERVISOR

10. THIS COURT ORDERS that notwithstanding anything to the contrary in the CPS Engagement Letter (as defined in the Initial Order), the Initial Order or any other Order of this Honourable Court, the CPS (as defined in the Initial Order) is hereby empowered and authorized,

but not required, on and after the Closing to take such additional actions as the Applicants or the Monitor, as applicable, considers necessary or desirable to assist (i) the Applicants in connection with the administration of these proceedings and (ii) the Monitor in performing the Monitor's functions and fulfilling its obligations under this Order, any other Order of this Honourable Court or in connection with the Transition Services Agreement.

11. THIS COURT ORDERS that the CPS shall continue to be paid its fees, expenses and any other amounts payable to the CPS under and pursuant to the CPS Engagement Letter after Closing until it is no longer necessary or desirable for the CPS to provide the assistance to the Applicants and Monitor as set out in this Order. Nothing in Order shall derogate from the protections afforded to the CPS by the Initial Order.

AMENDED AND RESTATED VESTING AND APPROVAL ORDER

12. THIS COURT ORDERS that the Canadian Vesting and Approval Order dated June 21, 2012 is hereby amended and restated in the form attached as **Schedule "1"** hereto.

ADDITIONAL PROVISIONS

- 13. THIS COURT ORDERS that the Fifth Report of the Monitor and the activities described therein are hereby approved.
- 14. THIS COURT ORDERS that the Applicants, the Purchaser or the Monitor may apply to this Honourable Court for advice and direction, or to seek relief in respect of, any matters arising from or under this Order.
- 15. THIS COURT HEREBY REQUESTS the aid and recognition of any court, tribunal, regulatory or administrative body having jurisdiction in Canada or in the United States, including the United States Bankruptcy Court for the District of Delaware, to give effect to this Order and to assist the Applicants, the Monitor and their respective agents in carrying out the terms of this Order. All courts, tribunals, regulatory and administrative bodies are hereby respectfully requested to make such orders and to provide such assistance to the Monitor, as an officer of this Court, as may be necessary or desirable to give effect to this Order, or to assist the Monitor and its agents in carrying out the terms of this Order.

16. THIS COURT ORDERS that this Order shall have full force and effect in all provinces and territories in Canada.

July 12, 2012

CENTIFIED A TRUE COPY

Spivak, J.

DEPUTY REGISTRAR

SCHEDULE "A" - Additional Applicants

Arctic Glacier California Inc.

Arctic Glacier Grayling Inc.

Arctic Glacier Lansing Inc.

Arctic Glacier Michigan Inc.

Arctic Glacier Minnesota Inc.

Arctic Glacier Nebraska Inc.

Arctic Glacier Newburgh Inc.

Arctic Glacier New York Inc.

Arctic Glacier Oregon Inc.

Arctic Glacier Party Time Inc.

Arctic Glacier Pennsylvania Inc.

Arctic Glacier Rochester Inc.

Arctic Glacier Services Inc.

Arctic Glacier Texas Inc.

Arctic Glacier Vernon Inc.

Arctic Glacier Wisconsin Inc.

Diamond Ice Cube Company Inc.

Diamond Newport Corporation

Glacier Ice Company, Inc.

Ice Perfection Systems Inc.

ICEsurance Inc.

Jack Frost Ice Service, Inc.

Knowlton Enterprises, Inc.

Mountain Water Ice Company

R&K Trucking, Inc.

Winkler Lucas Ice and Fuel Company

Wonderland Ice, Inc.

File No. CI 12-01-76323

THE QUEEN'S BENCH Winnipeg Centre

IN THE MATTER OF THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c. C-36, AS AMENDED

AND IN THE MATTER OF A PROPOSED PLAN OF COMPROMISE OR ARRANGEMENT WITH RESPECT TO ARCTIC GLACIER INCOME FUND, ARCTIC GLACIER INC. AND ARCTIC GLACIER INTERNATIONAL INC. and the ADDITIONAL APPLICANTS LISTED ON SCHEDULE "A" HERETO"

(collectively, the "Applicants")

APPLICATION UNDER THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c. C-36, AS AMENDED

AMENDED AND RESTATED CANADIAN VESTING AND APPROVAL ORDER

DATE OF HEARING: THURSDAY, JUNE 21, 2012 AT 10 A.M. BEFORE THE HONOURABLE MADAM JUSTICE SPIVAK

McCARTHY TÉTRAULT LLP

Barristers and Solicitors
Suite 5300, Box 48
Toronto Dominion Bank Tower
Toronto-Dominion Centre
Toronto, ON M5K 1E6

Kevin McElcheran

Tel: (416) 601-7730 Fax: (416) 868-0673 Law Society No. 22119H

Heather L. Meredith

Tel: (416) 601-8342 Fax: (416) 868-0673 Law Society No. 48354R

File No. 10671373

AIKINS, MacAULAY & THORVALDSON LLP

30th Floor – 360 Main Street Winnipeg, MB R3C 4G1

G. Bruce Taylor

Tel: (204) 957-4669 Fax: (204) 957-4218

J.J. Burnell

Tel: (204) 957-4663 Fax: (204) 957-4285

File No.: 1103500

THE QUEEN'S BENCH Winnipeg Centre

THE HONOURABLE MADAM)	THURSDAY, THE 21st
)	
JUSTICE SPIVAK)	DAY OF JUNE, 2012

IN THE MATTER OF THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c. C-36, AS AMENDED

AND IN THE MATTER OF A PROPOSED PLAN OF COMPROMISE OR ARRANGEMENT WITH RESPECT TO ARCTIC GLACIER INCOME FUND, ARCTIC GLACIER INC. AND ARCTIC GLACIER INTERNATIONAL INC. and the ADDITIONAL APPLICANTS LISTED ON SCHEDULE "A" HERETO"

(collectively, the "Applicants")

APPLICATION UNDER THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C., c. C-36, AS AMENDED

AMENDED AND RESTATED CANADIAN VESTING AND APPROVAL ORDER

THIS MOTION, made by the Applicants for an order, among other things, approving the sale transaction (the "Transaction") contemplated by the Asset Purchase Agreement (the "Asset Purchase Agreement") between the Applicants and Glacier Valley Ice Company, L.P. (California) (together, the "Vendors"), as vendors, and H.I.G. Zamboni, LLC (now known as Arctic Glacier LLC), as purchaser, made as of June 7, 2012; vesting the Vendors' right, title and interest in and to the assets described in the Asset Purchase Agreement (the "Assets"), to Arctic Glacier LLC, Arctic Glacier Canada Inc. and Arctic Glacier U.S.A., Inc. (collectively, the "Purchaser"); and, extending the Stay Period defined in paragraph 30 of the Initial Order of the Honourable Madam Justice Spivak dated February 22, 2012 (the "Stay Period"), was heard this day at the Law Courts Building at 408 York Avenue, in The City of Winnipeg, in the Province of Manitoba.

ON READING the Affidavit of Keith McMahon sworn June 13, 2012 (the "Affidavit"), and the Fourth Report of Alvarez & Marsal Canada Inc. (the "Monitor") dated June 15, 2012 (the "Fourth Report"), and on hearing the submissions of counsel for the Monitor, the Applicants, the Purchaser, the Arctic Lenders, TD Bank and the US Direct Purchaser Antitrust Settlement Class, counsel for the Trustees of Arctic Glacier Income Fund also appearing, counsel for Centerbridge Partners L.P. appearing on a watching brief, representatives of Talamod Master Fund, L.P. and TD Securities Inc. also present in person or by telephone, and no one appearing for any other person, including the U.S. Department of Justice Antitrust Division and parties to Assigned Contracts that are being assigned pursuant to this Order, although properly served as appears from the Affidavit of Corrine Smorhay and the Affidavit of Kelly Peters, both sworn June 20, 2012, both filed:

1. THIS COURT ORDERS that all capitalized terms used herein and not otherwise defined shall have the meaning ascribed thereto in the Asset Purchase Agreement.

SERVICE

2. THIS COURT ORDERS that the time for service of the Notice of Motion, the Affidavit, the Fourth Report and the supporting materials is hereby abridged and validated so that this Motion is properly returnable today and hereby dispenses with further service thereof.

SALE TRANSACTION

- 3. THIS COURT ORDERS AND DECLARES that the Transaction is hereby approved, and the execution of the Asset Purchase Agreement by the Vendors is hereby authorized and approved, with such minor amendments as the Vendors may deem necessary. The Vendors are hereby authorized and directed to take such additional steps and execute such additional documents as may be necessary or desirable for the completion of the Transaction and for the conveyance of the Assets to the Purchaser or such other person(s) as the Purchaser may direct and the Monitor may agree.
- 4. THIS COURT ORDERS AND DECLARES that upon the delivery of a Monitor's certificate to the Purchaser substantially in the form attached as Schedule "B" hereto (the "Monitor's Certificate"):

- all of the Vendors' right, title, benefit and interest in and to the Assets other than (a) the Canadian Assets (as herein defined) (the "U.S. Assets"), including, without limitation, the Vendors' rights, title and interest in and to any applicable Assigned Contracts, including all leases of real property, shall vest, without further instrument of transfer or assignment, absolutely in Arctic Glacier U.S.A., Inc. or such other person(s) as the Purchaser may direct and the Monitor may agree (provided that no agreement will be required if such transfer is to an Affiliate that agrees to be jointly and severally liable with the Purchaser), free and clear of and from any and all security interests (whether contractual, statutory, or otherwise), hypothecs, mortgages, pledges, options, warrants, trusts or deemed trusts (whether contractual, statutory, or otherwise), encumbrances, obligations, liabilities, demands, guarantees, restrictions, contractual commitments, rights, including without limitation, rights of first refusal and rights of set-off, liens, executions, levies, penalties, charges, or other financial or monetary claims, adverse claims, or rights of use, puts or forced sales provisions exercisable as a consequence of or arising from closing of the Transaction, whether arising prior to or subsequent to the commencement of these CCAA Proceedings, whether or not they have attached or been perfected, registered or filed and whether secured, unsecured, legal, equitable, possessory or otherwise, actual or threatened civil, criminal, administrative, regulatory, arbitral or investigative inquiry, action, complaint, suit, investigation, dispute, petition or proceeding by or before any Governmental Authority or person at law or in equity whether imposed by agreement, understanding, law, equity or otherwise, and any claim or demand resulting therefrom including but not limited to Antitrust proceedings commenced by the U.S. Department of Justice and various State's Attorney Generals (collectively, the "Claims") including, without limiting the generality of the foregoing: (i) any encumbrances or charges created by the Order of the Honourable Justice Spivak dated February 22, 2012 and any subsequent charges created by the Court (the "Court Charges"); (ii) all charges, security interests or claims evidenced by registrations pursuant to the Personal Property Security Act (Manitoba) or any other personal property registry system; (iii) Excluded Liabilities as defined in the Asset Purchase Agreement: and (iv) those Claims listed on Schedule "C" hereto (all of which are collectively referred to as the "Encumbrances", which term shall not include the permitted encumbrances, easements and restrictive covenants listed on Schedule "D") and, for greater certainty, this Court orders that all of the Claims and Encumbrances affecting or relating to the U.S. Assets are hereby released, extinguished, expunged and discharged as against the U.S. Assets; and
- (b) all of Arctic Glacier Inc.'s right, title, benefit and interest in and to the Assets (the "Canadian Assets"), including, without limitation, the Vendors' rights, title and interest in and to any applicable Assigned Contracts, including all leases of real property, shall vest, without further instrument of transfer or assignment, absolutely in Arctic Glacier Canada Inc. or such other person(s) as the Purchaser may direct and the Monitor may agree (provided that no agreement will be required if such transfer is to an Affiliate that agrees to be jointly and severally liable with the Purchaser), free and clear of and from any and all Claims including, without limiting the generality of the foregoing: (i) any Court Charges; (ii) all charges, security interests or claims evidenced by registrations pursuant to

the Personal Property Security Act (Manitoba) or any other personal property registry system; (iii) Excluded Liabilities as defined in the Asset Purchase Agreement; and (iv) those Claims listed on Schedule "C" hereto (all of which are collectively referred to as the "Encumbrances", which term shall not include the permitted encumbrances, easements and restrictive covenants listed on Schedule "D") and, for greater certainty, this Court orders that all of the Claims and Encumbrances affecting or relating to the Canadian Assets are hereby released, extinguished, expunged and discharged as against the Canadian Assets.

- 5. THIS COURT ORDERS that upon the registration in the applicable land registry office or land titles office of a Transfer/Deed of Land or equivalent document, or of an application for registration of this vesting order in the applicable prescribed form, the applicable land registrar or equivalent official is hereby directed to enter the applicable Purchaser (or such other person(s) as the Purchaser may direct and the Monitor may agree) (provided that no agreement will be required if such transfer is to an Affiliate that agrees to be jointly and severally liable with the Purchaser) as the owner of the subject real property in fee simple, and is hereby directed to delete and expunge from title to the real property any and all Claims and Encumbrances, including, without limitation, all of the Claims and Encumbrances listed in Schedule "C" hereto.
- 6. THIS COURT ORDERS that upon delivery of the Monitor's Certificate all of the rights and obligations of the Vendors under the Assigned Contracts (as defined in the Asset Purchase Agreement) shall be assigned to the applicable Purchaser or such other person(s) as the Purchaser may direct and the Monitor may agree (provided that no agreement will be required if such transfer is to an Affiliate that agrees to be jointly and severally liable with the Purchaser) (the "Assignee") pursuant to section 2.12 of the Asset Purchase Agreement and pursuant to section 11.3 of the CCAA and remain in full force and effect for the benefit of the Purchaser in accordance with their respective terms.
- 7. THIS COURT ORDERS that the assignment of the rights and obligations of the Vendors under the Assigned Contracts to the Assignee pursuant to section 2.12 of the Asset Purchase Agreement and pursuant to this order is valid and binding upon all of the counterparties to the Assigned Contracts, without further documentation, as if the Assignee was a party to the Assigned Contracts, notwithstanding any restriction, condition or prohibition contained in any such Assigned Contracts relating to the assignment thereof, including any provision requiring the consent of any party to the assignment.

- 8. THIS COURT ORDERS that each counterparty to the Assigned Contracts is prohibited from exercising any right or remedy under the Assigned Contracts by reason of any defaults thereunder arising from these CCAA proceedings or the insolvency of the Vendors, or any failure of the Vendors to perform a non-monetary obligation under the Assigned Contracts, or as a result of any actions taken pursuant to or as a result of the Asset Purchase Agreement. All notices of default and demands given in connection with any such defaults under, or non-compliance with the Assigned Contracts shall be deemed to have been rescinded and shall be of no further force or effect.
- 9. THIS COURT ORDERS that as a condition of the closing of the Transaction, all existing monetary defaults in relation to the Assigned Contracts, other than those arising by reason of the Vendors' insolvency, the commencement of these CCAA Proceedings, or the Vendors' failure to perform a non-monetary obligation, shall be paid in accordance with section 2.12 of the Asset Purchase Agreement.
- 10. THIS COURT ORDERS that notwithstanding anything contained in this order, nothing shall derogate from the obligations of the Purchaser (or such other person(s) as the Purchaser may direct and the Monitor may agree) (provided that no agreement will be required if such transfer is to an Affiliate that agrees to be jointly and severally liable with the Purchaser) to assume the Assumed Liabilities, including the Assumed Accounts Payable, and to perform its obligations under the Assigned Contracts, as set out the Asset Purchase Agreement and the Designated Purchaser Agreement.
- 11. THIS COURT ORDERS that for the purposes of determining the nature and priority of Claims, the net proceeds from the sale of the Assets shall stand in the place and stead of the Assets, and that from and after the delivery of the Monitor's Certificate all Claims and Encumbrances shall attach to the net proceeds from the sale of the Assets with the same priority as they had with respect to the Assets immediately prior to the sale, as if the Assets had not been sold and remained in the possession or control of the person having that possession or control immediately prior to the sale.
- 12. THIS COURT ORDERS that the Monitor shall, in accordance with the provisions of the SISP (as defined in the Affidavit), be authorized and directed to pay to the Arctic Lenders (as defined in the Asset Purchase Agreement) from the net proceeds of the sale of the Assets an

amount sufficient to pay the Lender Claims (as defined in the SISP and as calculated on the closing of the Transaction) in full and in cash, as specified in a pay-out letter to be provided by the Arctic Lenders on or before the closing of the Transaction. Such payment shall be made concurrently with, and as a condition precedent to, the closing of the Transaction. The balance of the net proceeds of the sale of the Assets shall be held by the Monitor in accordance with the terms hereof or any further order of the Court; provided that the Monitor may pay any amounts owing from time to time to persons who are entitled to the benefit of a Court Charge.

- 13. THIS COURT ORDERS AND DIRECTS the Monitor to file with the Court a copy of the Monitor's Certificate, forthwith after delivery thereof.
- 14. THIS COURT ORDERS that, pursuant to clause 7(3)(c) of the Canada Personal Information Protection and Electronic Documents Act and any equivalent legislation in any other jurisdiction applicable, the Vendors are authorized and permitted to disclose and transfer to the Purchaser all human resources and payroll information in the Vendors' records pertaining to the Vendors' past and current employees. The Purchaser shall maintain and protect the privacy of such information and shall be entitled to use the personal information provided to it in a manner which is in all material respects identical to the prior use of such information by the Vendors.
- 15. THIS COURT ORDERS that, notwithstanding:
 - (a) the pendency of these proceedings;
 - (b) any applications for a bankruptcy order now or hereafter issued pursuant to the Bankruptcy and Insolvency Act (Canada) in respect of and of the Vendors and any bankruptcy order issued pursuant to any such applications; and
 - (c) any assignment in bankruptcy made in respect of and of the Vendors;

the vesting of the Assets in the Purchaser (or such other person(s) as the Purchaser may direct and the Monitor may agree) pursuant to this Order shall be binding on any trustee in bankruptcy that may be appointed in respect of the Vendors and shall not be void or voidable by creditors of the Vendors, nor shall it constitute nor be deemed to be a settlement, fraudulent preference, assignment, fraudulent conveyance, transfer at undervalue, or other reviewable transaction under the *Bankruptcy and Insolvency Act* (Canada) or any other applicable federal or provincial legislation, nor shall it constitute oppressive or unfairly prejudicial conduct pursuant to any applicable federal or provincial legislation.

- 16. THIS COURT ORDERS AND DECLARES that the Transaction is exempt from the application of the *Bulk Sales Act* (Ontario) and any equivalent legislation in any other jurisdiction in which all or any part of the Assets are located.
- 17. THIS COURT ORDERS AND DECLARES that nothing in this Order or the Asset Purchase Agreement discharges, releases, or precludes any environmental liability under United States law to the United States or any department, agency, or instrumentality thereof (each, a "U.S. Governmental Unit") of any entity based on its ownership or operation after the Time of Closing (as defined in the Asset Purchase Agreement) of real property. Nor shall anything in this Order enjoin or otherwise bar a U.S. Governmental Unit from asserting or enforcing, outside this Court, any liability described in the preceding sentence.
- 18. THIS COURT ORDERS that any interested party served with notice of this motion after Friday, June 15, 2012, including those additional parties identified by the Purchaser as parties to receive service after the issuance of this Order, may apply to this Court by notice of motion served on or before July 3, 2012 for hearing on July 12, 2012 to vary or amend this Order other than paragraph 12 hereof. Service on such parties in such manner is hereby validated. If no such application is brought on or before July 3, 2012, this Order shall be deemed effective, nunc pro tune, and without such further right of comeback, as against such parties.

STAY EXTENSION

19. THIS COURT ORDERS that the Stay Period is hereby extended until and including September 14, 2012.

MONITOR'S REPORT AND ACTIVITIES

20. THIS COURT ORDERS that the Third Report of the Monitor dated May 14, 2012 and the Fourth Report and the activities described therein are hereby approved.

SEALING

- 21. THIS COURT ORDERS that the Confidential Appendix to the Fourth Report shall be sealed, kept confidential and not form part of the public record, but rather shall be placed, separate and apart from all other contents of the Court file, in a sealed envelope attached to a notice that sets out the title of these proceedings and a statement that the contents are subject to a sealing order and shall only be opened upon further Order of the Court.
- 22. THIS COURT HEREBY REQUESTS the aid and recognition of any court, tribunal, regulatory or administrative body having jurisdiction in Canada or in the United States, including the United States Bankruptcy Court for the District of Delaware, to give effect to this Order and to assist the Monitor and their respective agents in carrying out the terms of this Order. All courts, tribunals, regulatory and administrative bodies are hereby respectfully requested to make such orders and to provide such assistance to the Monitor, as an officer of this Court, as may be necessary or desirable to give effect to this Order or to assist the Monitor and its agents in carrying out the terms of this Order.
- 23. THIS COURT ORDERS that this Order shall have full force and effect in all provinces and territories in Canada.

J

June 21, 2012	

SCHEDULE "A" - Additional Applicants

Arctic Glacier California Inc.

Arctic Glacier Grayling Inc.

Arctic Glacier Lansing Inc.

Arctic Glacier Michigan Inc.

Arctic Glacier Minnesota Inc.

Arctic Glacier Nebraska Inc.

Arctic Glacier Newburgh Inc.

Arctic Glacier New York Inc.

Arctic Glacier Oregon Inc.

Arctic Glacier Party Time Inc.

Arctic Glacier Pennsylvania Inc.

Arctic Glacier Rochester Inc.

Arctic Glacier Services Inc.

Arctic Glacier Texas Inc.

Arctic Glacier Vernon Inc.

Arctic Glacier Wisconsin Inc.

Diamond Ice Cube Company Inc.

Diamond Newport Corporation

Glacier Ice Company, Inc.

Ice Perfection Systems Inc.

ICEsurance Inc.

Jack Frost Ice Service, Inc.

Knowlton Enterprises, Inc.

Mountain Water Ice Company

R&K Trucking, Inc.

Winkler Lucas Ice and Fuel Company

Wonderland Ice, Inc.

Schedule B - Form of Monitor's Certificate

THE QUEEN'S BENCH Winnipeg Centre

IN THE MATTER OF THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c. C-36, AS AMENDED

AND IN THE MATTER OF A PROPOSED PLAN OF COMPROMISE OR ARRANGEMENT WITH RESPECT TO ARCTIC GLACIER INCOME FUND, ARCTIC GLACIER INC. AND ARCTIC GLACIER INTERNATIONAL INC. and the ADDITIONAL APPLICANTS LISTED ON SCHEDULE "A" HERETO"

(collectively, the "Applicants")

MONITOR'S CERTIFICATE

RECITALS

- A. Pursuant to an Order of the Honourable Madam Justice Spivak of the Manitoba Court of Queen's Bench (the "Court") dated February 22, 2012, Alvarez & Marsal Canada Inc. was appointed as the monitor (the "Monitor") in the Applicants' proceedings under the Companies' Creditors Arrangement Act, R.S.C. 1985, c. C-36, as amended.
- B. Pursuant to an Order of the Court dated June 21, 2012 (the "Canadian Vesting and Approval Order"), the Court approved an asset purchase agreement made as of June 7, 2012 (the "Asset Purchase Agreement") between the Applicants and Glacier Valley Ice Company, L.P. (California) (together, the "Vendors"), as vendors, and H.I.G. Zamboni, LLC (now known as Arctic Glacier LLC), and provided for the vesting of all of the Vendors' right, title and interest in and to the Assets described in the Asset Purchase Agreement, including, without limitation, the Vendors' rights, title and interest in and to any Assigned Contracts (as defined therein), including all leases of real property in Arctic Glacier LLC, Arctic Glacier Canada Inc. and Arctic Glacier U.S.A., Inc. (collectively the "Purchaser"), which vesting is to be effective with respect to the Assets upon the delivery by the Monitor to the Purchaser of a certificate confirming (i) the

payment by the Purchaser of the Purchase Price for the Assets; (ii) that the conditions to Closing as set out in the Asset Purchase Agreement have been satisfied or waived by the Vendors and the Purchaser, respectively; and (iii) the Transaction has been completed to the satisfaction of the Monitor.

C. Unless otherwise indicated herein, terms with initial capitals have the meanings set out in the Asset Purchase Agreement.

THE MONITOR CERTIFIES the following:

- 1. The Purchaser has paid and the Vendors have received the Purchase Price for the Assets payable on the Closing Date pursuant to the Asset Purchase Agreement;
- 2. The conditions to Closing as set out in the Asset Purchase Agreement have been satisfied or waived by the Vendors and the Purchaser, respectively; and
- 3. The Transaction has been completed to the satisfaction of the Monitor.
- 4. This Certificate was delivered by the Monitor at ____ [TIME] on ____ [DATE].

Alvarez & Marsal Canada Inc., in its capacity as Monitor, and not in its personal or corporate capacity

Per:				
	Name:	 <u></u>	 ***************************************	
	Title:			

Schedule C - Claims to be deleted and expunged

REAL PROPERTY ENCUMBRANCES TO BE DISCHARGED

A. OWNED PROPERTY

1. 12132 & 12136 - 121 A Street, Edmonton, Alberta, T5L 0A4

(a) Title No.: 012 170 358

Legal Description: Plan RN64, Block 24, Lot 8 excepting thereout the Westerly 10 feet throughout of the said lot, excepting thereout all mines and minerals.

Municipality: City of Edmonton

Registered Owner: Arctic Glacier Inc.

Encumbrances:

Registration No.	Date	Instrument Type/ Description	From/By
992 255 839	27/08/1999	Mortgage	Montreal Trust Company

(b) Title No.: 012 170 700

Legal Description: Plan RN64, Block 24, Lots 9 and 10 excepting thereout the most Westerly 10 feet in uniform width throughout said lots, taken for lane, as shown on Road Plan 2199NY excepting thereout all mines and minerals.

Municipality: City of Edmonton

Registered Owner: Arctic Glacier Inc.

Encumbrances:

Registration No.	Date	Instrument Type/ Description	From/By
992 255 839	27/08/1999	Mortgage	Montreal Trust Company

2. 412 - 41 Avenue N.E. Calgary, Alberta, T2E 2N3

(a) Title No.: 981 406 325

Legal Description: Plan Calgary 7410938, Block 13, that portion of Lot "A", which lies to the west of the easterly Fifty Four and Thirty Hundredths (54.30) metres in perpendicular width throughout containing 0.203 hectare more or less, excepting thereout all mines and minerals

Municipality: City of Calgary

Registered Owner: Arctic Glacier Inc.

Encumbrances:

Registration No.	Date	Instrument Type/ Description	From/By
991 250 891	31/08/1999	Mortgage	Montreal Trust Company

3. 625 Henry Avenue, Winnipeg, Manitoba, R3A 0B1

(a) Winnipeg Land Titles Office Title No.: 2028565/1

Legal Description: Parcels A to E Plan 42917 WLTO

Said Parcel A being together with a right-of-way for all purposes and as appurtenant to the land above described over and upon Parcel 2 Plan 2547 WLTO in RL 35 Parish of St. John.

Registered Owner: Arctic Glacier Inc.

Encumbrances:

Registration No.	Date	Instrument Type/ Description	From/By
2410597/1	1999/08/25	Mortgage	Montreal Trust Company

(b) Winnipeg Land Titles Office Title No.: 2030254/1

Legal Description: Firstly: Lot 3 and all those portions of Lots 1 and 2 Block 41 Plan No. 331 WLTO (W Div) lying to the NW of those portions of said Lots 1 and 2 shewn as Parcel 2 Plan No. 2547 WLTO Lot 35 Parish of St. John

Secondly: All those portions of said Lots 1 and 2 shewn as Parcel 2 on said Plan No. 2547 WLTO subject to a right-of-way for all purpose and as appurtenant to that portion of said Lot 1, lying to the SE of said Parcel 2 and appurtenant to Block 7 Plan 94 WLTO (W Div) over and upon the whole of said Parcel 2.

Registered Owner: Arctic Glacier Inc.

Encumbrances:

Registration No.	Date	Instrument Type/ Description	From/By			
3075752/1	2004/12/10	Mortgage	Computershare Canada	Trust	Company	of

(c) Winnipeg Land Titles Office Title No.: 2030253/1

Legal Description: ELY 20 feet of Lot 4 Block 41 Plan 331 WLTO (W Div) in RL 35 Parish of St

John.

Registered Owner: Arctic Glacier Inc.

Encumbrances:

Registration No.	Date	Instrument Type/ Description	From/By			
3075752/1	2004/12/10	Mortgage	Computershare T Canada	rust	Company	of

4. 200 Statesman Drive, Mississauga, Ontario, L5S 1X7

(a) Land Registry Office #43, Parcel Register for Property Identifier: 14029-1139 (LT)

Legal Description: Parcel Block 33-1, Section 43M-957; Block 33, Plan 43M957, together with Part Lot 11, Concession 1, East of Hurontario Street, Part 4, Plan 43R16717 as in TT81032; subject to LT1098087 Mississauga

Registered Owner: Arctic Glacier Inc.

Encumbrances:

Registration No.	Date	Instrument Type	Description
LT1098091Z	1990/02/12	Application to Annex Restrictive Covenants	Lostrock Corp.
LT1979090	1999/08/23	Charge	From 1179554 Ontario Inc. to Montreal Trust Company
PR180019	2001/12/14	Transfer	From 1394332 Ontario Inc. to The Arctic Group Inc.
PR255417	2002/06/04	APL Ch Name Owner	From The Arctic Group Inc. to Arctic Glacier Inc.

5. 6 McKinstry Street, Hamilton, Ontario, L8L 6C1

(a) Land Registry Office #62, Parcel Register for Property Identifier: 17192-0005 (LT)

Legal Description: Part Reserve 3, Survey 32, as in AB319263; Part Reserve 3, Survey 32, Part 1, 62R9795; Part Reserve 3, Survey 32, Part 2, 62R7060, except Part 1, 62R7413; Reserving Minerals in CD306923; together with access over Part 1 on 62R7413, as in CD305159; Hamilton

Registered Owner: Arctic Glacier Inc.

Encumbrances:

Registration No.	Date	Instrument Type	Description
LT566928	1999/08/23	Deed Trust Mort	From 1334202 Ontario Inc. to Montreal Trust Company
WE70318	2001/12/13	Transfer	From 1394332 Ontario Inc. to The Arctic Group Inc.
WE98279	2002/06/04	APL Ch Name Owner	From The Arctic Group Inc. to Arctic Glacier Inc.

6. 745 Park Avenue W., Chatham, Ontario, N7M 1X3

(a) Land Registry Office #24, Parcel Register for Property Identifier: 00527-0044 (LT)

Legal Description: Part of Lot 20, Concession 1 Eastern Boundary Raleigh as in 590170, except Part 1, 24R6402; together with 590170; subject to 495938, 495939; Chatham-Kent

Registered Owner: Arctic Glacier Inc.

Encumbrances:

Registration No.	Date	Instrument Type	Description
593547	1999/08/23	Deed Trust Mort	From 1334202 Ontario Inc. to Montreal Trust Company
595536	1999/11/04	Deed Trust Mort	From 1334202 Ontario Inc. to Montreal Trust Company
612238	2001/12/12	Transfer	From 1394332 Ontario Inc. to The Arctic Group Inc.
CK43065	2010/02/18	APL Ch Name Owner	From The Arctic Group Inc. to Arctic Glacier Inc.
CK43433	2010/03/03	APL (General)	Arctic Glacier Inc.

7. 2655 – 2677 Reading Street, Montreal, Quebec, H3K 1P6

(a) **Description:** An immovable property fronting on Reading Street, in the City of Montreal, Province of Quebec, known and designated as lot number ONE MILLION THREE HUNDRED AND EIGHTY-TWO THOUSAND THREE HUNDRED AND FIFTY-FIVE (1 382 355) of the Cadastre of Quebec, Registration Division of Montreal.

With the building thereon erected bearing civic numbers 2655, 2675 and 2677 Reading Street, City of Montreal, Province of Quebec.

Registered Owner: Arctic Glacier Inc. Deed of Transfer registered under number 5 293 999 on October 12, 2001.

Hypothecs and Encumbrances:

- i. Deed of Hypothec and Issue of Bonds executed before Mtre. Jean Mousseau, Notary, on August 19, 1999 and registered on August 20, 1999 under number 5 118 118 by 3149030 Canada Limited in favour of Montreal Trust Company for an amount of \$500,000,000.00 bearing interest at the rate of 25% per annum and an additional hypothec in the amount of \$100,000,000.
- ii. Deed of Hypothec and Issue of Bonds executed before Mtre. Steve Collins, Notary, on March 22, 2002 and registered on the same day under number 5 331 878 by Arctic Glacier Inc. in favour of Computershare Trust Company of Canada for an amount of \$500,000,000.00 bearing interest at the rate of 25% per annum.
- iii. Supplemental Deed of Hypothec executed before Me Tamar Chamelian, Notary, on February 8, 2010 and registered on the same day under number 16 919 886 by Arctic Glacier Inc. in favour of Computershare Trust Company of Canada for an amount of \$500,000,000.00 bearing interest at the rate of 25% per annum.

8. 2760 Reading Street, Montreal, Quebec, H3K 1P6

(a) Description: An immovable property fronting on Reading Street, in the City of Montreal, Province of Québec, known and designated as lot number ONE MILLION THREE HUNDRED AND EIGHTY-TWO THOUSAND THREE HUNDRED AND THIRTEEN (1 382 313) of the Cadastre of Québec, Registration Division of Montreal.

With a building thereon erected bearing civic number 2760 Reading Street, City of Montreal, Province of Québec.

Registered Owner: Arctic Glacier Inc. Deed of Transfer registered under number 5 293 999 on October 12, 2001.

Hypothecs and Encumbrances:

- i. Deed of Hypothec and Issue of Bonds executed before Mtre. Jean Mousseau, Notary, on August 19, 1999 and registered on August 20, 1999 under number 5 118 118 by 3149030 Canada Limited in favour of Montreal Trust Company for an amount of \$500,000,000.00 bearing interest at the rate of 25% per annum and an additional hypothec in the amount of \$100,000,000.
- ii. Deed of Hypothec and Issue of Bonds executed before Mtre. Steve Collins, Notary, on March 22, 2002 and registered on the same day under number 5 331 878 by Arctic Glacier Inc. in favour of Computershare Trust Company of Canada for an amount of \$500,000,000.00 bearing interest at the rate of 25% per annum.
- iii. Supplemental Deed of Hypothec executed before Me Tamar Chamelian, Notary, on February 8, 2010 and registered on the same day under number 16 919 886 by Arctic Glacier Inc. in favour of Computershare Trust Company of Canada for an amount of \$ 500,000,000.00 bearing interest at the rate of 25% per annum.

B. LEASED PROPERTY

1. 9679 (also known as 9669) 186th Street, Surrey, British Columbia, V4N 3N8

(a) New Westminster Land Title Office Title No.: AA60615E

Legal Description: Parcel Identifier: 007-144-431. Lot A (AA60615) District Lot 99 Group 2 New Work of the Physics A762

Westminster District Plan 54762.

Registered Owner: Shogun Compu-Time Ltd.

Encumbrances:

Registration No.	Date	Instrument Type/ Description	From/By
BT97364 (of Lease BT97363)	2002-03-25	Mortgage, transferred to BT130238	Computershare Trust Company of Canada (Inc. No. A52313)

2. 1625 McAra Street, Regina, Saskatchewan, S4N 6H4

(a) Title No.: 139229321

Legal Description: Lot K Blk/Par 96 Plan No. 87R08061 Extension 0 as described on Certificate of

Title 87R08068.

Registered Owners: Cynthia Hughes, James Hughes, Darlene Panchuk and Clayton Panchuk

Encumbrances:

Interest No./ Int. Register No.	Date	Instrument Type/ Description	From/By
151304183 100851612	26 May 1998	Personal Property Security Interest	RoyNat
151304172 100851601	26 Aug 1999	Personal Property Security Interest	Montreal Trust Company
153713718 117035883	17 Dec 2010	Mortgage of Lease	Arctic Glacier Inc. to Computershare Trust Company of Canada
153736768 117041914	21 Dec 2010	Mortgage of Lease	Arctic Glacier Inc. to Computershare Trust Company of Canada

(b) Title No.: 139229376

Legal Description: Lot K Blk/Par 96 Plan No. 87R08061 Extension 0 as described on Certificate of Title 87R08068.

Registered Owners: Cynthia Hughes, James Hughes, Darlene Panchuk and Clayton Panchuk

Encumbrances:

Interest No./ Int. Register No.	Date	Instrument Type/ Description	From/By
151304251 100851612	26 May 1998	Personal Property Security Interest	RoyNat
151304240 100851601	26 Aug 1999	Personal Property Security Interest	Montreal Trust Company
153713729 117035883	17 Dec 2010	Mortgage of Lease	Arctic Glacier Inc. to Computershare Trust Company of Canada
153736779 117041914	21 Dec 2010	Mortgage of Lease	Arctic Glacier Inc. to Computershare Trust Company of Canada

Schedule D – Permitted Encumbrances, Easements and Restrictive Covenants related to the Real Property

(unaffected by the Vesting Order)

- 1. Reservations, limitations, provisos and conditions expressed in any original grant from any Governmental Authority.
- 2. Liens for Taxes, local improvements, assessments or governmental charges or levies not at the time due or delinquent.
- 3. Applicable municipal by-laws, development agreements, subdivision agreements, site plan agreements, other agreements, building and other restrictions, easements, servitudes, rights of way and licences affecting the use or value of the Lands which do not materially impair the use or value of the Lands affected thereby as presently used.
- 4. Defects or irregularities in title to the Lands affecting the use or value of the Lands which do not materially impair the use or value of the Lands affected thereby as presently used.
- 5. Any matters which might be revealed by (i) an up-to-date survey of any Lands; or (ii) an inspection and/or site investigation of any owned Lands together with any errors in the survey, which do not materially impair the use or value of the Lands affected thereby as presently used.
- 6. Any rights of expropriation, access or use, or any other similar rights conferred or reserved by or in any statute of Canada or any province or territory thereof or of the United States or any state, jurisdiction, territory or possession thereof.
- 7. Undetermined, inchoate or statutory Liens (including the Liens of public utilities, workers, suppliers of materials, builders, contractors, architects and unpaid vendors of moveable property) incidental to the current operation of the Lands which relate to obligations not yet due or delinquent and which have not been registered in accordance with Applicable Law.